# University of Massachusetts Amherst ScholarWorks@UMass Amherst

Doctoral Dissertations 1896 - February 2014

1-1-1990

# Designing, implementing, and evaluating a staff development program on learning styles and teaching styles in an urban junior high school.

Roy R. Taylor University of Massachusetts Amherst

Follow this and additional works at: https://scholarworks.umass.edu/dissertations 1

#### Recommended Citation

Taylor, Roy R., "Designing, implementing, and evaluating a staff development program on learning styles and teaching styles in an urban junior high school." (1990). *Doctoral Dissertations* 1896 - February 2014. 4701. https://scholarworks.umass.edu/dissertations 1/4701

This Open Access Dissertation is brought to you for free and open access by ScholarWorks@UMass Amherst. It has been accepted for inclusion in Doctoral Dissertations 1896 - February 2014 by an authorized administrator of ScholarWorks@UMass Amherst. For more information, please contact scholarworks@library.umass.edu.

# DESIGNING, IMPLEMENTING, AND EVALUATING A STAFF DEVELOPMENT PROGRAM ON LEARNING STYLES AND TEACHING STYLES IN AN URBAN JUNIOR HIGH SCHOOL

A Dissertation Presented

by

ROY R. TAYLOR

Submitted to the Graduate School of the University of Massachusetts in partial fulfillment of the requirements for the degree of

DOCTOR OF EDUCATION

MAY 1990

Education

© Copyright by Roy R. Taylor 1990
All Rights Reserved

# DESIGNING, IMPLEMENTING, AND EVALUATING A STAFF DEVELOPMENT PROGRAM ON LEARNING STYLES AND TEACHING STYLES IN AN URBAN JUNIOR HIGH SCHOOL

A Dissertation Presented

by

ROY R. TAYLOR

Approved as to style and content by:

Byrd L. Jones, Chair

Robert W. Maloy, Member

Thomas Ratliffe, Member

Marilyn Haring-Hidore, Dean

School of Education

# DEDICATION

This dissertation is dedicated to:

My loving wife, Juanita, who was the "wind beneath my wings,"

and

My adorable daughter, Melanie, whose understanding and patience encouraged me to prevail.

# **ACKNOWLEDGEMENTS**

My deepest gratitude and appreciation go to the following individuals and organizations for supporting and assisting me in the completion of this study:

Byrd Jones, whose knowledge, directions, persistence, and faith enabled me to get through the dark tunnel.

Robert Maloy, whose outline provided me with a blueprint to assemble the components of this study.

Thomas Ratliffe, whose precise comments and support helped me over many hurdles.

Joan Zabawa, whose many long distance phone calls kept me "in touch" with U/Mass.

Susan Savitt, who encouraged me to keep the flames burning.

Dianne Crosson, Mary Garrett, Dianne Robinson, and Jerrie Smith, for volunteering to participate in the study.

Faculty and staff at Roosevelt Junior Senior High School, for their cooperation and support during the study.

Roosevelt Board of Education, for supporting the Roosevelt U/Mass Staff Development Program.

My parents, for encouraging me "to get as much education as you can."

My sister Mary, who encouraged and supported me in my educational endeavors since early childhood.

#### ABSTRACT

PROGRAM ON LEARNING STYLES AND TEACHING STYLES

IN AN URBAN JUNIOR HIGH SCHOOL

MAY 1990

ROY R. TAYLOR, B.S., TUSKEGEE INSTITUTE
M.ED., AUBURN UNIVERSITY

ED.D., UNIVERSITY OF MASSACHUSETTS
Directed by: Dr. Byrd L. Jones

This study documented the process of designing, implementing, and evaluating a low-cost, school-based, researcher-conducted staff development program for four voluntary Black seventh and eighth grade science and social studies teachers at Roosevelt Junior Senior High School, New York, during 1987-88. The underlying purpose was to learn more about effective school improvement in low-income school districts. Staff development efforts aimed to expand teachers' teaching styles to accord more closely with urban Black students' diverse learning styles in four homogeneously grouped classes with the intent of (a) improving students' academic achievement, (b) increasing attendance to class, and (c) improving attitudes about school.

Five collaboratively planned workshops on teaching styles and learning styles provided opportunities for

teachers (a) to expand their repertoire of teaching styles, (b) to recognize students' learning styles, (c) to develop problem-solving techniques for their classes, (d) to work cooperatively with other teachers, and (e) to enhance their professional growth.

Based on observations, discussions, interviews, and self-reports, the researcher concluded that teachers (a) utilized different teaching strategies in their classes to address more students' learning styles, (b) provided more opportunities for students to work cooperatively in class, (c) shared ideas with other teachers, (d) planned their lessons with other teachers, (e) incorporated students' suggestions into lessons, (f) utilized a larger variety of teaching aids and materials, and (g) praised their students more often. Their students (a) displayed less inappropriate behavior in class, (b) worked cooperatively with their classmates, (c) offered teachers suggestions to make classes more interesting, and (d) complained less often of boredom.

Evidence of positive gains suggests that staff development is feasible in low-income school districts when staff are encouraged to seek support and ideas from students, parents, teachers, and administrators about issues considered important in their setting. Collaborative planning fosters a climate that encourages a variety of school improvement efforts to flourish over time.

# TABLE OF CONTENTS

ACKNOWLEDGEMENTS	v
ABSTRACT	v i
LIST OF TABLES	кi
Chapter	
1. INTRODUCTION	1
Problem	4
Purpose	9
Setting	11
Community	11 13
Scope and Limitations	13
Generalization Limitations  Design Problems  Experimenter Blas	13 14 14
Outline of Chapters	15
11. REVIEW OF LITERATURE	17
Introduction	17 18 24 27
Individualized Classroom	29 30 30
Staff DevelopmentSummary	40 49
111. METHODOLOGY	5
Involving Others in Planning	5

Workshop #1	66
Objectives  Procedures  Activity #1Outlining Roles  Activity #2Goal Consensus  Activity #3Need for Study  Activity #4Analyze and Prioritize Needs  Activity #5Teaching Styles/Learning  Styles  Feedback for Workshop #1	67 68 68 68 69 70 71 72
Workshop #2	73
Objectives  Procedures  Activity #1Reviewing Results of Teaching	74 74
Styles Survey	74
Styles	78 80 81
Workshop #3	82
Objectives	82 83 83
Activity #2Discussions on Participants' Learning Styles	84 86
Workshop #4	87
Objectives  Procedures  Activity #1Designing Projects/Activities  Activity #2Discussion on Projects  Feedback for Workshop #4	87 87 88 88 88
Workshop #5	89
Objectives	89 90 90
Activity #3Learning Styles Inventory by Dunn and Dunn	90 91
Activity #5Collecting Data  Feedback for Workshop #5	9: 9:

	Summary	93
IV.	RESULTS	95
	Demographic Characteristics.  Student Attitudinal Survey.  Students' Grades, Attendance, and Attitudes  Summary	96 101 117 132
٧.	OVERVIEW, EFFECTS, OUTGROWTHS, RECOMMENDATIONS FOR STAFF DEVELOPMENT, FUTURE IMPLICATIONS, AND CONCLUSIONS	135
	Overview  Effects on Teachers  Effects on Students  Effects on Researcher  Outgrowths	135 138 144 146 147
	Pathfinder Program	148 149 150 151
	Objectives Feedback from Workshop Participants	152 153
	Recommendations for Staff Development	154 155
	in their Classes	157 159 160
APPE	NDICES	162
A. B. C.	IN THE SEVENTH GRADE LEVEL AP SCIENCE CLASS, THE SEVENTH GRADE LEVEL A SOCIAL STUDIES CLASS. THE SEVENTH GRADE LEVEL	164 166
D. E. F.	TEACHING STYLES SURVEYGURVEY	172 180 182 186
G. H.	POST-STUDENT ATTITUDINAL SURVEY	19
REFE	RENCES	

# LIST OF TABLES

Table		Page
1.	Seventh Grade Retentions at Roosevelt Junior Senior High School from 1984 to 1989	5
2.	Village and Student Population	12
з.	Erikson's Identity vs. Role Confusion Stage (11/12-18) and Early Adolescence	23
4.	Staff Development Workshops on Teaching Styles/ Learning Styles February 1, 1988 - April 4, 1988 Roosevelt Junior Senior High School Roy R. Taylor, Researcher	66
5.	Demographic Characteristics of the Seventh Grade Level AP Science ClassTeacher #3	97
б.	Demographic Characteristics of the Seventh Grade Level A Social Studies ClassTeacher #1	98
7.	Demographic Characteristics of the Seventh Grade Level B Social Studies ClassTeacher #2	99
8.	Demographic Characteristics of the Eighth Grade Level C Social Studies ClassTeacher #4	100
9.	Question #1. Is the classwork usually interesting?	101
10.	Question #2. Do you expect to pass this class this quarter?	. 103
11.	Question #3. Are you usually given assignments that are of interest only to you?	. 105
12.	Question #4. Do you like coming to this class?	. 108
13.	Question #5. What could (did) your teacher do differently to help you improve your learning in this class?	. 113
14.	Comparison of Students' Grades, Attitudes, and Attendance for the First, Second, Third, and Fourth Quarters during the 1987-88 School Term in the Seventh Grade Level AP Science ClassTeacher #3	

15.	Comparison of Students' Grades, Attitudes, and Attendance for the First, Second, Third, and Fourth Quarters during the 1987-88 School Term in the Seventh Grade Level A Social Studies ClassTeacher #1	120
16.	Comparison of Students' Grades, Attitudes, and Attendance for the First, Second, Third, and Fourth Quarters during the 1987-88 School Term in the Seventh Grade Level B Social Studies ClassTeacher #2	122
17.	Comparison of Students' Grades, Attitudes, and Attendance for the First, Second, Third, and Fourth Quarters during the 1987-88 School Term in the Seventh Grade Level C Social Studies ClassTeacher #4	124
18.	Changes in the Means of Students' Grades, Attitudes, and Attendance in the Seventh Grade Level AP Science Class	126
19.	Changes in the Means of Students' Grades, Attitudes, and Attendance in the Seventh Grade Level A Social Studies Class	127
20.	Changes in the Means of Students' Grades, Attitudes, and Attendance in the Seventh Grade Level B Social Studies Class	128
21.	Changes In the Means of Students' Grades, Attitudes, and Attendance in the Eighth Grade level C Social Studies Class	129
22.	Summary of Changes in the Means of Students' Grades, Attitudes, and Attendance in the Four Classes during the 1987-88 School	131

#### CHAPTER 1

#### INTRODUCTION

Teachers have various styles of teaching and students have various styles of learning. Combs, Avila, and Purkey (1971) stated that interchanges between teachers and their students differ from moment to moment. Teachers react to each student in terms of the unique questions, ideas, problems, and concerns that he/she is expressing at a particular instant. Teachers should increase their awareness of the variety of learning styles in their classes and implement various teaching strategies to help meet students' needs and to develop a school environment where students want to be on a regular basis.

According to Barr and Dreeben (1977), as long as classrooms retain their traditional character, students will not attend school on a regular basis. Monk and Ibrahim (1984) stated that absenteeism may also be a consequence of the negative activities taking place in school. For instance, students may absent themselves from an unruly class or a teacher whose approach does not match a student's needs.

Sarason (1982) posed thought-provoking questions relative to learning, schooling and education as he referred to processes that take place in school. What do children in a classroom think about the learning climate? Do students

have their own theories about how and why one learns? Are there discrepancies between the theories children think about and the theories they are required to adopt? Would children like to talk about these matters? Are they able to talk about them? How do teachers explain and discuss their theories of learning and thinking with their students? According to Dunn (1982), if teachers sought answers to these questions from their students, and implemented the responses into their teaching, students' academic achievement as well as their attitudes toward school would improve. Dunn (1982) stressed the need for teachers to listen to students.

Prior to the 1986-87 school term, seventh and eighth grade students at Roosevelt Junior Senior High School were academically grouped homogeneously and heterogeneously for placement into their major subjects—English, social studies, mathematics, and science—and foreign languages. Homogeneously, students were placed in their English class based on their language and reading skills on the Iowa Test of Basic Skills. For example, students who scored within a year's range were placed in the same group or track.

Heterogeneously, students were placed into their science, mathematics, social studies, foreign languages—except advanced placement—and their ancillary classes, which included physical education, typing, home and careers, music, and technology. Students were "labeled" according to

the English class in which they were enrolled, and, thus, teachers tended to set their expectations according to the homogeneous designation of the English class. To that extent, teachers did not allow for individual differences among students, for example, their interests, abilities, or learning styles. Teachers often taught their heterogeneously grouped classes homogeneously because they felt more comfortable and using the traditional teaching methods. The philosophy of the scheduling process for junior high students at Roosevelt Junior Senior High School was not clearly defined to encourage teachers to address the students' individual learning needs.

During the 1986-87 school term, however, a plan was devised to address the individual learning needs of the seventh grade students. Students were scheduled into their major subjects and foreign languages based on their interests, abilities, past academic performance in their major subjects, teachers' recommendations, and past performance in language, reading, and mathematics on the lowa Test of Basic Skills.

A scheduling committee, composed of administrators and counselors, reviewed the academic data and teachers' recommendations for each student and scheduled them into their major subjects and foreign language based on the analysis of the data. A student, for example, could have been scheduled for AP English, B science,

C mathematics, AP Spanish, and B social studies, which reinforced the previously undefined tracking system at Roosevelt Junior Senior High School. Thus, classes had become homogeneous or tracked again.

Olexa and Schafer (1971) reported that students who were assigned to lower tracks exhibited lower self-esteem, displayed more school misconduct, dropped out of school more often, and displayed a higher rate of delinquency than students in middle or high tracks (cited in Goodlad, 1984, p. 152).

Oakes (1985) also concluded that low-track students participate less in extracurricular activities at school, exhibit more school and classroom misconduct, are involved more often in delinquent behavior outside of school, are more alienated from school, and have higher drop-out rates.

#### Problem

A two item questionnaire (see appendix A) was administered to approximately 50 seventh graders, who had patterns of poor school attendance, during the spring of 1986 to ascertain why they had not attended school or their classes regularly. Although most of the students surveyed indicated they would prefer being in school and being in their classes, there was, however, a significant percentage (17%) of the 233 seventh graders who were not attending school or their classes on a regular basis, which

possibly contributed to the 16.30% retention rate of seventh graders during the 1984-85 school term.

Table 1 reveals the number and percent of students who repeated the seventh grade from 1984-1989. During the 1983-84 school term, 16.19% of the seventh grade students were retained. Whereas, during the 1988-89 school term, 10.00% of the seventh grade students were retained. The largest retention rate of 18.80% occurred during the 1985-86 school term. The mean number of seventh graders who were retained from 1984 to 1989 was 36, with a mean percentage of 16.39.

TABLE 1

Seventh Grade Retentions at Roosevelt Junior Senior High

School from 1984 to 1989

Year	# Students	# Retained	% Retained
1983-84	247	40	16.19
1984-85	233	38	16.30
1985-86	218	41	18.80
1986-87	212	39	18.39
1987-88	209	39	18.66
1988-89	170	17	10.00
Mean	215	36	16.39

Why are seventh graders at Roosevelt Junior Senior High School cutting their classes and being retained at alarming rates? Are they turned off by school? Do they think school is not worthwhile? There seemed to be a "self-destruct" mechanism built into the grading procedures for those students who did not attend their classes regularly. If students have four unexcused absences from class during a quarter (forty school days), they fall the course with a grade of 40C. If these students fail another quarter during the same academic year because of excessive unexcused absences, then they are denied the opportunity to attend summer school, which might enable them to be promoted to the next grade. (Note: Roosevelt School District's Promotion Policy (1987-88) states that seventh grade students must pass 3 of the 4 majors, English, mathematics, science, and social studies, with a grade of 65% to be promoted to the eighth grade.)

The following is the intervention procedure for excessive absences:

- Progress reports are sent home to parent(s)/ guardian(s) and guidance counselors at the mid-quarter.
- 2. Counselors arrange conferences with students' parent(s)/guardian(s) to discuss attendance problems.

- 3. Teachers and parent(s)/guardian(s) usually give their reasons for poor attendance, i.e., "lazy", "doesn't like school", "comes to school only to eat", etc.
- 4. Teachers and counselors offer suggestions for improving attendance.
- 5. Students are expected to improve their attendance to classes without input into the possible problems or suggestions for improvement.

The cycle of poor attendance tended to decline only temporarily after "intervention" by counselors, teachers, and parents. Part of the problem of poor attendance had not been addressed because the students had not, in most cases, had an opportunity to explain their absences. Their responses were seldom treated as significant, if someone took time to ask. When students were absent from school, absent from classes, late for classes, or cut classes, they were sending "educators" strong signals that needed to be addressed in the form of program evaluation and staff development for teachers and students.

Junior high school students needed more opportunities to be taught according to their learning styles and Junior high teachers needed to acquire a repertoire of teaching styles. The researcher observed that many Junior high students often failed classes, cut classes, exhibited negative behavior inside and outside classes, and portrayed signs of apathy toward teachers' prescribed lessons. The

researcher had also observed varying levels of frustration among Junior high teachers when their "well-planned" lessons were not accepted with enthusiasm by all students. Through numerous observations, in most instances, teachers presented their lessons in large groups, employing the same teaching style for all students. Very little, if any, attention was given to students who did not grasp information as quickly as others. Usually, if students asked questions in class, teachers answered with previously used examples or explanations. If students did not understand the first time, usually they did not understand the second time when teachers repeated their examples or explanations without regard for students' learning styles. For example, using picturesque descriptions, rephrasing statements, or using graphics on the chalkboard might increase students' understanding.

When junior high teachers searched for ways to maintain students' interest in classwork, there seemed to be a need for teachers themselves to acquire a repertoire of teaching styles. Presumably, experienced teachers have acquired a variety of teaching techniques which would enhance learning in their classes. Their inability or unwillingness to vary their teaching styles affected students' motivation, behavior, and learning.

These and similar observations motivated the researcher to share concerns with junior high teachers, administrators,

and students about teaching styles/learning styles and their effects on students' experiences in school. To assist junior high teachers in acquiring a repertoire of teaching styles, a plan was needed to (1) assess teachers' teaching styles, (2) offer staff development workshops at convenient times for teachers, and (3) to assist students in understanding their learning styles.

The need for teaching styles/learning styles staff development further manifested itself through the researcher's interactions with students, parents, teachers, and administrators. Students often exhibited signs of frustration in classes, but many could not intelligently analyze problems which confronted them.

# Purpose

The purpose of this study was to design, implement, and evaluate the effects of teachers acquiring a repertoire of instructional approaches appropriate to the diverse learning styles of seventh and eighth grade students to improve students' achievement, students' attendance, and to assess the impact on students' attitudes.

# Significance of the Problem

Recently, great emphasis has been placed on individualization of learning and teaching students in the least restrictive environment. Many educators have interpreted individualization and least restrictive

environment to mean placing students in special education classes.

Little attention has been given to individualization of teaching for "normal" students in the "regular" classrooms. Many programs and techniques, however, have expanded individualization for students who have been classified as physically or mentally handicapped to accommodate their rates and styles of learning. An Individualized Educational Plan (IEP) ordinarily helps meet the varied educational needs of handicapped students.

Doyle and Ponder (1975) suggested that teachers accepted advice for improving their teaching when it met three criteria. The advice was operational and described actual teacher behaviors; it was consistent with a teacher's role definition; and it was cost effective in terms of time and energy.

Teachers should become familiar with various teaching styles in order to have a repertoire of styles from which to choose. A variety of instructional modes might accommodate the learning styles of their students. Teachers who increase their awareness of a variety of learning styles may increase their range of teaching approaches and their flexibility in using varied approaches.

Instructionally, this study provided teachers with a repertoire of teaching techniques, through workshops and cooperative planning. As a possible outgrowth of this

study, the school district has begun to provide, as a part of its orientation for new teachers, an ongoing staff development program to provide teachers with instructional approaches, which would possibly reduce the number of students being referred to special education, outside educational facilities, and increase school attendance.

Academically, this study offered students the opportunity to be taught in the manner that they receive and process information best. Therefore, if students' learning styles are incorporated into classroom activities, their interest in school and their academic retention will likely increase, thus resulting in higher academic success in school and improved attendance, which may transcend academic disciplines and grade levels.

# Setting

#### Community

Roosevelt is a one square mile village located in the southwestern section of Nassau County, Long Island, New York, which is an unincorporated community of approximately 14,000 inhabitants. It is a part of the Town of Hempstead and does not have its own local government, health agencies, sanitation services, or police department.

According to the 1980 Federal Census there are 630 families of two or more, whose head of household is a single parent. This comprises 31 percent of the families in the

community. There are 581 families with children under 18 whose head of household is female and no husband is present. There are 49 households where there is a male head of household and no wife is present.

Roosevelt is a racially isolated community. The ethnic composition of the community has changed from predominately White in 1960 to predominately Black in 1980 as indicated by Table 2.

TABLE 2

Village and Student Population				
Village Population				
	U.S.	Census		
Category	<u>1960</u>	<u>1970</u>	1980	
Black	2,241	10,135	12,516	
White	8,000	4,730	1,259	
Others	108	235	334	
Student Population				
	School Dis	strict Census		
Category	1966	<u>1975</u>	<u>1983</u>	
Black	1,595	4,180	3,134	
White	1,872	87	25	
Other Ethnic	0	118	67	
(Federal Project Proposal Roosevelt School District, 1983)				

# School

Roosevelt Junior Senior High School has a comprehensive curriculum with leveling (tracking) according to students' academic abilities (AP, advanced placement; A above average; B, average; and C, below average) in each major academic discipline, English, social studies, science, and mathematics. There are 125 teachers. The administrative staff includes a principal, two assistant principals, a dean of students, and five administrative supervisors.

The junior high school is comprised of the seventh and eighth grades, which is not physically separated from the senior high school. There are approximately 450 students in grades seven and eight. There are ten seventh grade homerooms and ten eighth grade homerooms.

#### Scope and Limitations

# Generalization Limitations

The results of this study were limited in scope, because they were not used to generalize about increased students' achievement, students' attendance or students' attitudes in any other setting. The participant population included seventy-five Black students, three Hispanic students, and one White student, ages 12-17 years, who were from low to middle socioeconomic backgrounds. The results were limited to specific instructional materials and procedures used in this study.

# Design Problems

The possible problems which may have influenced the results of the study were (a) time of classes; (b) scheduled school activities during the school day; (c) absentee rate of participating teachers; (d) students losing interest; (e) inadequate teaching materials; (f) extra attention given to the participating students; (g) teachers' unwillingness to implement different teaching strategies; (h) small group of teachers participating in study; and (i) time of year.

### Experimenter Bias

All necessary precautions were taken to reduce the possibility of experimenter bias employing the following procedures:

- 1. Clarifying the purposes and procedures
- 2. Making participation voluntary
- 3. Adjusting to the group's needs and goals to the extent possible

The researcher's position, administrative supervisor in the Roosevelt Junior Senior High School's Junior high department, had possible advantages and disadvantages which could have affected the outcomes of the study.

- 1. Some advantages were the following:
  - a. Researcher had easy access to teachers, students, and students' academic and attendance records.
  - b. Researcher could adjust teachers' schedules which would have provided them time to participate in workshops during the workday.
  - c. Researcher adjusted his schedule to coincide with teachers' schedules.
  - d. Students did not feel compelled to attend classes.
- 2. Some disadvantages were the following:
  - a. Teachers initially provided the researcher with information they thought he needed for a successful study.
  - if the study was not successful.

# Outline of Chapters

Chapter 1 includes a statement of the problem, the purpose of the study, a description of the community and the school, and the need for staff development for learning styles/teaching styles. Chapter 2 reviews current literature relating to learning styles, teaching styles, student achievement, student attendance, attitudes, and staff development. Chapter 3 reports the procedures for

conducting the study. Chapter 4 reports and analyzes the data resulting from the study. And, chapter 5 offers recommendations for staff development, implications for future studies, and conclusions.

### CHAPTER 11

# REVIEW OF LITERATURE

# Introduction

This study proposed to design, implement, and evaluate an appropriate staff development program to expand the repertoire of teaching styles of junior high teachers at Roosevelt Junior Senior High School with the intent of improving students' academic achievement, attitudes, and attendance to class. The study was based on four propositions. First, students in early adolescence are at a stage of high variability and changing identity so that seventh and eighth grade students--whether tracked or not--include a wide range of learning styles. Second, if students find appropriate instruction and concern for themselves as individuals, they will attend class and that attendance should eventually yield academic achievement. Third, a planned staff development project to expand the repertoire of teaching styles to accord more closely, though not necessarily on a one to one basis, with students currently enrolled should increase opportunities for success in school. And fourth, a staff development program is more likely to be ongoing if participants are provided opportunities to "buy into" the process through collaborative planning. These four propositions guided the development and evaluation of this study and structured the

review of literature, which focused on adolescence, learning styles, teaching styles, and staff development.

### Adolescence

There are many definitions of adolescence, each of which reflects a somewhat different facet of the interaction between two societies and the adolescent as described by Haviland and Scarbough (1981, p. 6):

Medically, adolescence begins with the growth hormonal changes associated with the sexual maturity and ends when there is no further growth. Legally, adolescence ends with the assumption of adult responsibility for voting, for the draft, and for debts and contracts, including marriage. Ages for the assumption of these duties range from 14 to 21 in various countries and There are also variations in the age at which a child becomes responsible for criminal acts; these range from 14 to 18 in most countries and states. Educationally, adolescence is the time spent in junior high and high school, and sometimes college. When an adolescent leaves school, he or she may be considered an adult. Psychologically, adolescence is a period of transition during which cognitive, physical, Adult roles are personality, and social changes occur. practiced and a personal identity is formed.

Coleman (1978) noted that adolescence is a very stressful developmental stage. Youth often experience an identity crisis as they define roles for themselves.

Safer (1986, p. 406) emphasized that the Junior High/Middle School is relatively hectic for adolescents, particularly when compared to elementary school:

Junior High/Middle School allows its students far more independence and mobility. In Junior High/Middle School, students move through the halls for over 30 min/day, which is at least three to four times more than which occurs in tightly supervised Elementary School corridors. The seven different classrooms/seven

between period hall transfer schedule creates both an open and a "fluid" educational setting. As such, it provides more possibilities for behavioral deviations, e.g., class cutting and substance abuse.

For example, at Roosevelt Junior Senior High School, seventh and eighth grade students attend eight different classes with eight different teachers daily, which requires their moving through the building unescorted nine times each day. Whereas at the elementary schools in the Roosevelt School District, students remain with the same teacher for their major subjects—English, reading, science, social studies, and mathematics—and they only changes classes with teacher supervision for remedial classes, enrichment classes, and physical education. Additionally, seventh and eighth grade students at Roosevelt Junior Senior High School lack opportunities to go outside to "run and play" with their peers after they eat lunch.

Nisbet and Entwistle (1969) stated that the transition from Elementary School to Junior High/Middle School is stressful, but transitions to new schools and to secondary schools usually take a short period of time for most students to make the adjustments. Elias et al. (1985) added that vulnerable students entering secondary school face problems that rarely diminish with time. For example, stress that is experienced by adolescents is usually compounded for the poor, racial or ethnic minorities, or recent immigrants ("American Adolescent," 1989). Elias et al. further added that adolescents' higher levels of

stress may be manifested in different ways; high absenteeism from school/class; poor academic achievement, and disruptive behavior, which may be attributed to adolescent alienation and poor self-concept.

According to Mackey and Alghren (1977), adolescent alienation is a major problem of contemporary society. Bronfenbrenner (1974; 1986) stressed that adolescent alienation begins in the family. Some contemporary families have become disorganized as a result of external forces such as economic and social pressures.

Calabrese (1987) stated that the sense of alienation is carried by the adolescent to the school, and the school's response often reinforces the decision to withdraw or rebel. Usually these adolescents are easily identified because they eat lunch together, smoke together, and congregate together. Calabrese further stated that since the school makes little effort to integrate these students, they view this interaction as justification for rejecting the world of adults (p. 930).

Hotinen and Korpinen (cited in Korpinen, 1987) reported that the school self-image of weak students remains extremely low, since the majority of these students believe that teachers make them feel inadequate and that they often are put down in school. Similiarly, Ford (1986) emphasized that students who have a negative perception of school are also likely to have a poor self-image, inattentive parents,

low achievement, and a high absentee rate. These students are often placed in the back of the classroom, ignored by the teacher, and are treated with malice rather than concern and patience.

Adler (1982) stated that schools seem more concerned with graduation requirements, discipline, course offerings, and the records of the athletic teams, which are reflected in the school's curriculum that neglects programs to encourage integration. Additionally, students are often segregated by intellectual ability through grouping (Gullotta, 1983).

Korpinen (1987) maintained that the self-concept is a significant educational idea and its development is an essential aim of education. The self-concept "is shaped by environmental influences, especially in social interaction with those closest and most important to the individual" (p. 50). Further, Korpinen maintained that individuals respect themselves, consider themselves valuable, and accept themselves. They do not consider themselves better than others; neither do they consider themselves worse. They are aware of their limitations and they trust their own developmental possibilities (p. 52). DeAnde (1975) stated that children who have healthy self-concepts get along well with other children, know their strengths and weaknesses, are curious, and try out new things.

According to Burns (1982), if the pupil's individual characteristics have been considered sufficiently in teaching: in aims, in content, in methods, and in evaluative reporting, then the evaluation will have a positive effect on the pupil's self-concept. Hale-Benson (1988) suggested that "children's self-confidence will be fostered through frequent compliments, praise, display of work, performance, open houses, and frequent success experiences" (p. 163).

Manning (1988) stated that middle school educators can design learning environments based on an understanding of Erik Erikson's psychosocial theories by designing lessons and activities that incorporate children's developmental stages. Erikson (1963, p. 95) divided human life into eight psychosocial stages. The eight psychosocial stages include:

(a) trust vs. mistrust 0-18 months; (b) autonomy vs. shame

1.5-3 years; (c) initiative vs. guilt 3-6 years; (d)

industry vs. inferiority 6/7-11/12 years; (e) identity vs.

role confusion 11/12-18 years; (f) intimacy vs. isolation young adulthood; (g) generativity vs. stagnation middle age; and (h) ego integrity vs. despair older adulthood.

Table 3 (cited in Manning, 1988, p. 97) describes the characteristics of Erikson's Identity vs. Role Confusion Stage (11/12-18) and early adolescence and their implications for educators.

### TABLE 3

Erikson's Identity vs. Role Confusion Stage (11/12-18) and Early Adolescence (1963, p. 95)

#### Characteristics Implications for Educators Childhood ends/puberty 1. Recognize early begins adolescent's mental. physical, and social change require appropriate learning experiences Primarily concerned 2. Recognize that others' with others perceptions of early perceptions of them adolescents affect self-perception and achievement. 3. Confused about roles 3. Accept that learners' and evolving identity evolving identity requires appropriate and planned educational activities 4. Can be cliquish/ 4. Accept cliques as part of early adolescent's life; clannish and cruel to however, remind learners peers that their behavior affects others 5. Has increased 5. Planned appropriate educational experiences independence and and responsibility of responsibility learners Support search for ideal 6. 6. Beginning search for yet encourage learners to ideal life and maintain sense of reality situations

According to Maier (1969), the Identity vs. Role
Confusion period is the beginning of the search for an
identity and a mastery of the problems of childhood and a
readiness to confront the challenges of the adult world.

During these years, adolescents begin to question many previously resolved issues (Erikson, 1963). Consequently, adolescents are at a stage of high variability and changing identity, which includes a wide range of learning styes.

## Learning Styles

What is learning style? Canfield and Lafferty (1970) discussed conditions, content modes and expectations; Dunn and Dunn (1978) itemized stimuli and elements; Gregorc (1979) emphasized distinctive behaviors and dualities—abstract and concrete levels; Hunt (1979) referred to conceptual level; Kolb (1981) specified heredity equipment, past experience, and the environment; and Schmeck et al. (1977) contrasted deep and shallow information processing (cited in Dunn et al., 1981, p. 372). Fischer and Fischer (1979) referred to learning style as a pervasive quality in the behavior of an individual (p. 245). Guild and Garger (1985) categorized learning styles into four basic groups: (a) cognition, (b) conceptualizations, (c) affect, and (d) behavior (p. 6).

Fisher and Fisher (1979, pp. 246-250) categorized learning styles into ten groups:

1. The Incremental Learner--These students proceed step-by-step fashion, systematically adding bits and pieces together to gain larger understanding. . . .

2. The Intuitive Learner--The learning style of these students does not follow traditional logic, chronology, or step-by-step sequence. . . .

- 3. The Sensory Specialist--These students rely primarily on one sense for meaningful formation of ideas. . . .
- 4. The Sensory Generalist--These students use all or many of the senses in gathering information and gaining insight. . . .
- 5. The Emotionally Involved--These students function best in a classroom in which the atmosphere carries a high emotional charge. . . .
- 6. The Emotionally Neutral--These students function best in a classroom where the emotional tone is low-keyed and relatively neutral. . . .
- 7. Explicitly Structured--These students learn best when the teacher makes explicit a clear unambiguous structure for learning. . . .
- 8. Open-Ended Structure--These students feel at home and learn best in a fairly open-ended environment. . . .
- 9. The Damaged Learner--These students are physically normal yet damaged in self-concept, social competency, aesthetic sensitivity, or intellect in such a way that they develop negative learning styles. . .
- 10. The Eclectic Learner--These students can shift learning styles and function profitably may find one or another styles more beneficial, but can adapt to and benefit from others. . . .

Among the most widely used learning styles model in United States' schools, according to Wedlund (1987), is the Dunn model. The Dunn model described learning styles in terms of how the individual's ability to learn new or difficult material is affected by the following variables:

(a) the immediate environment (noise level, temperature, amount of light, and furniture design); (b) emotionality (degree of motivation, persistence, responsibility, and need for structure); (c) sociological needs (learning alone or with peers, learning with adults present, learning in groups); (d) physical characteristics (auditory, visual, tactile, and kinesthetic strengths, best time of day for

learning, need for food and drink while learning, and mobility requirements); and (e) psychological inclinations (global and analytic strengths) (cited in Carbo & Hodges, 1988, p. 55).

Carlisle (1985, p. 78) introduced four basic strategies used to learn: "stating memorized information; performing a sequence of activities; distinguishing and categorizing a thing or idea; and identifying relationships and their importance." According to Carlisle, the learning that results from these strategies varies from stage to stage. For example, stating memorized information and performing a sequence of activities require an inordinate amount of practice and hard work. And memorizing can be boring for students and can be easily forgotten. However, performing is usually motivating for students and is not forgotten as easily, because relationships are formed.

Hunkins (1987) stated that students at any age can benefit greatly from being aware of the processes they use in their learning. Hunkins also stated that teachers can be instrumental in helping students process their learning strategies by sharing their teaching methods with the students and when students are having difficulty comprehending an idea, teachers can act as a "consultant" and use the "teachable moment" to help the students understand (pp. 65-66).

Keefe (1979) stated:

Learning style diagnosis opens the door to placing individualized instruction on a more rational basis. It gives the most powerful leverage yet available to educators to analyze, motivate, and assist students in school. As such, it is the foundation of a truly modern approach to education. (quoted in Dunn et al., 1981, p. 372)

For example, when teachers are aware of their students' learning styles, teachers can design more diverse learning environments to incorporate a broader range of students' learning styles to increase students' grades, attendance, and attitudes.

# Teaching Styles

Teachers' teaching styles are influenced by their learning styles (Gephart et al., 1980; MacNeil, 1980; McDaniel, 1982). Teachers, for example, tend to utilize their preferred teaching style more often when teaching their classes, as opposed to utilizing a variety of teaching styles to address students' diverse learning styles.

According to Heikkinen, Pettigrew, and Zakrajsek (1985), "A broad understanding of individual learning styles and preferred learning conditions ought to underscore the need for an eclectic approach to teaching, one that provides for a variety of learning styles" (p. 80).

Various researchers categorized and discussed teaching styles and their impact on teaching and learning. Fischer

and Fischer (1979, p. 251) categorized teaching styles into six groups:

- 1. The Task-Oriented--These teachers prescribe materials to be learned and demand specific performance on the part of the students.
- 2. The Cooperative Planner--These teachers plan the means and ends of instruction with student cooperation.
- 3. The Child Centered--These teachers provide a structure to pursue whatever the students want to do or whatever interests them.
- 4. The Subject Centered--These teachers focus on organized content to the near exclusion of the learner.
- 5. The Learning Centered--These teachers have equal concern for the student and for the curricular objectives, the materials to be learned.
- 6. The Emotionally Exciting and Its Counterpart--These teachers show their own intensive emotional involvement in teaching.

Smith (1975) stated that the goal of teaching should support students' learning and respond to students' interests. Carlisle (1985) reported that teachers and students can promote or impede learning in classrooms. In a study, student trainees were asked to list things which they believed lead to unsatisfactory learning and to list things which they believed lead to satisfactory learning. The students indicated several variables that they believed promoted or impeded learning. According to the students, the following variables lead to satisfactory learning: The teacher provides a positive atmosphere; allows group involvement; and treats the students with respect. The student applies what is learned; is motivated; sets goals; sets priorities concerning study time; relates concepts to understanding ideas; and respects the teacher.

Conversely, students reported, according to Carlisle, that the following variables lead to unsatisfactory learning: The teacher is self-centered; lacks the ability to communicate information to students; speaks in a monotone; does not help students learn; allows personal problems to interfere; teaches incorrect information; and tests on information not taught. The teachers stated that students are usually unprepared; are not alert; do not want to learn because the subject is boring; lack motivation to learn; cannot relate concepts to familiar things; and do not set learning goals.

Crenshaw (1986) stated that teachers should assist students in developing good traits and changing or eliminating bad ones. Crenshaw also stated that students can maintain their individuality but still have a feeling of belonging.

Dunn and Dunn (1975) described three classroom conditions which teachers may use to meet the individual learning needs of their students: individualized classroom; open classroom; and alternative programs:

## Individualized Classroom

The teacher is responsible for diagnosing, prescribing for, and guiding each student through the learning process. Recognizing the different elements of learning style, he permits students to work anywhere in the environment, in any sociological pattern that they choose. When a student demonstrates ability to follow objectives that have been assigned to him, he is permitted to continue working as he prefers and is gradually permitted more and more options in objectives, resources, activities, and evaluation.

When a student does not appear to be able to work independently, structure is added to his prescription so that he works, to varying degrees, under the direct supervision of the teacher. Multimedia, multisensory resources are available to students who may select from among them. Objectives are written on an individual basis and may be contributed to or developed by the students. When progress is not satisfactory, the teacher becomes increasingly directive. Grades are determined as a result of testing related to each youngster's specific objectives. (p. 41)

## Open Classroom

Children are permitted to select their own curriculum, resources, schedule, and pace of learning. Students may remain with a topic as long as it interests them and may study alone, with a friend or two, or in a small group. Since youngsters learn in very individual ways, the teacher is responsible for providing an environment rich in multimedia resources and for encouraging student involvement with the materials. Objectives, if used, are determined by the child and may vary from student to student and on a continuously changing basis. Grades are not given, but evaluations are made in terms of each child's demonstrated growth. A positive and "happy" attitude is considered very important for student progress. (p. 42)

## Alternative Programs

Students are given curriculum choices, freedom, and objectives and are expected to gather and retain information independently. Students are usually permitted a voice in their program development. Since alternative programs differ widely, the degree to which options are provided concerning objectives, resources, activities, and evaluation is dependent on the individual program, not the student. (p. 43)

An alternative program, which was known as The School Within a School, was instituted at Roosevelt Junior Senior High School during the 1985-86 and 1986-87 school terms.

The program was designed to address the needs of 60 students in grades 9-12 who had been identified as "at-risk" students. The students, in most instances, had repeated

several grades; they posed an inordinate amount of disciplinary problems; and some had even dropped out of school.

The School Within a School incorporated characteristics of the individualized classroom as well as the open classroom. Students progressed at their own rate on individually prescribed lessons. Reports indicated that many students progressed two or more years in their major subjects during one academic year. In addition to the required instructional materials, which the students learned at their individual rates, students could select related learning materials to enhance their class activities or for personal gratification.

During the two years the School Within a School was in operation, reports indicated that it had a positive impact on its students. Students' disruptive behavior declined significantly. Students' attendance to school/class increased significantly. And, the majority of the students received regular high school diplomas with their respective classes.

Similarly, an Alternative School was begun at Roosevelt Junior Senior High School in September, 1989. The newly developed Alternative School is based on the concepts of the School Within a School, except the 80 students attend classes from 3:30 P.M. to 7:30 P.M. at Roosevelt Junior Senior High School.

On the contrary, Goodlad (1984) described the typical, predominant approach to instruction as follows:

The teacher sat at a desk watching the class, lecturing, or reading. Most students were writing, a few were stretching, and the remainder were looking contemplatively or blankly into space. The classrooms we observed were more like than unlike those in the old images so many of us share. Usually we saw desks or tables arranged in rows, oriented toward the teacher at the front of the room. (pp. 93-94)

Four elements of classroom life in the schools of sample came through loud and clear from our data. First, the vehicle for teaching and learning is the total group. Second, the teacher is the strategic, pivotal figure in the group. Third, the norms governing the group derive primarily from what is required to maintain the teacher's strategic role. Fourth, the emotional tone is neither harsh and punitive nor warm and joyful; it might be described most accurately as flat. (p. 108)

The classes described by Goodlad differ markedly from the learning styles-based classes that Dunn and Griggs (cited in Dunn, 1989) observed during their visits to secondary schools throughout the United States in 1987.

During Dunn and Griggs' visit to several secondary schools throughout the United States in 1987, they reported (a) patterns within the classrooms consisted of pairs, small groups, or individuals working alone, or with a teacher or an aide, (b) all learning was compatible with the students' assessed learning style preferences, (c) students assumed a major responsibility for learning, (d) students were extensively involved in the learning process, and (e) students acquired the required knowledge, skills, and competencies through a wide variety of strategies (p. 3).

Dunn and Griggs reported that teachers at Corsicana High School, Corsicana, Texas, analyzed students' perceptual strengths and developed tactual/kinesthetic material for students, who previously achieved poorly (p. 3). Midwest High School, Midwest, Wyoming, reported an increase in teachers' positive attitudes, a significant decrease in students' absences over a five year period, and a significant number of students scoring on grade level on standardized testing (p. 3). Sacred Heart Academy, Hempstead, New York, reported a significant increase in the responsibility levels of the students, which translated into less direct supervision and a significant increase in students' academic achievement gains (p. 4). Roberson High School, Chicago, Illinois, reported that students have conducted themselves with dignity that reflects how they feel about themselves, their schooling and their teachers (p. 4). And Cedar Crest High School, Lebanon, Pennsylvania, reported that the students behaved with decorum and earned better grades than previously when they were permitted to use radios in class (p. 4).

The identification of children's modality strengths and appropriate planning of activities to incorporate children's modality strengths are thought to prevent learning difficulties in the classroom (Barbe and Milone, 1980; Barbe and Milone, 1981; Barbe and Swassing, 1979; Dunn and Carbo, 1981; Pizzo, 1982; Reiff, 1985). For example, "auditory

learners want to express themselves orally and to hear the assignment. Visual learners must see new materials, have pictures, study guides, and charts available. Kinesthetic learners need to touch and act out what they are learning" (Reiff, 1987, p. 107). For example, when teachers say to students: "Let me tell you," "Let me show you," "Let's do it," are comments one may hear in a classroom where different modalities are considered. When a student says, "I don't understand," the teacher then attempts to explain the material in a different way than it was originally presented (Guild & Garger, 1985, p. 66).

Glasser (cited in Brandt, 1988, p. 39) argued that teachers need to find ways in class to give students chances to associate with others in friendly ways and do this as a planned part of learning.

Further, Glasser stated:

Good researchers like David Johnson, Robert Slavin, and Spencer Kagan have proven the effectiveness of cooperative learning or as I call it, learning-teams. Here the teacher, instead of trying to force all the students down a single learning pathway, becomes a facilitator who goes from team to team encouraging helping, inquring, and prodding. In this method of instruction, the teacher gets nose to nose with learning, and students are encouraged to branch out creatively and stray from the "common" pathway. (p. 41)

Oakes (1985, p. 211) concluded that "cooperative learning strategies are an appropriate place to begin to develop the substance of an instructional mode designed to counter the limited, uninspired classroom instructional practices and homogeneous grouping." Further, Oakes

emphasized that cooperative learning has three advantages over competitive and individualistic methods: (1) a built-in incentive for students to interact with one another as learning resources; (2) a means of accommodating learning differences in the learning process; and (3) a way of greatly minimizing or eliminating the effects of initial differences in students' skill levels or learning rates in the assigning of rewards for learning (p. 210).

Slavin (1984) argued that for team learning to be effective, instructional methods must adequately provide for the four alterable elements of Carroll's (1963) model: appropriate levels of instruction (ability to understand instruction), incentives (perseverance), time (opportunity), and quality of instruction.

When cooperative learning methods provide group goals based on the learning of all members, the effects on student achievement are remarkably consistent. Of thirty-eight studies of at least four weeks duration comparing cooperative methods to traditional control methods, thirty-three found significantly greater achievement for the cooperatively taught classes, and five found no significant differences (Slavin, 1987).

In a synthesis of research on cooperative learning,
Slavin (1981) indicated that the positive effects of
cooperative learning methods on student achievement appear
equally frequently in elementay and secondary schools, in

urban, suburban, and rural schools, and in subjects as diverse as mathematics, language arts, social studies, and reading. However, studies indicated that there is a tendency for Blacks to gain outstandingly in achievement as a result of working cooperatively (Lucker et al., 1976; Slavin, 1977; Slavin & Oickle, 1980), although Whites also achieve more as an outcome of cooperative learning (p. 657). Hale-Benson (1989) maintained that Black children's learning is "very highly characterized by learning from people rather than objects. Therefore, a high adult-child ratio, small group learning and peer-group tutoring are useful" (p. 81). Cureton (1978) and Slaughter (1969) suggested that schools incorporate culturally induced teaching styles used with Afro-American children into their repertoire of instructional methods (cited in Shades, 1982, p. 236).

In addition to academic gains, research overwhelmingly support the usefulness of cooperative learning for improving the social outcomes of schooling, such as intergroup relations, attitudes towards mainstreamed students, and general positive relations between students of different ethnic backgrounds. Otterbourg (1986) reported that students in cooperative learning settings "are more likely...to name students of other races as best or close friends and to have more cross-race interactionns outside as well as inside the classroom" (cited in Ianni, 1989, p. 138). Cooperative learning also seems to make students feel

better about themselves (Slavin, 1981, p. 658). Another teaching strategy is peer tutoring. Hurtford and Hart (1979) concluded from their study that peer tutoring challenges many of our assumptions about who is best able to teach children. Teachers' negative views of peer interaction may reflect a lack of training in how to use this valuable resource according to Hiebert (1980). She also states that we can hardly expect harmonious and effective interaction to magically follow the command "interact" if children have not had prior peer learning experiences within the classroom and if teachers have provided little structure for these experiences (pp. 877-878). Teachers must tap this valuable resource because studies on peer tutoring and group learning indicate that peer learning can enhance instructional outcomes (Allan et al., 1976; Webb, 1977).

Mastery learning, (Bloom, 1976; Block and Burns, 1976), one of the most widely used alternatives to traditional teaching methods, focuses primarily on students' abilities to understand instruction by attempting to insure that all students have mastered the previous skill before attempting the next. Mastery learning can be individually-paced or group-paced (Block and Anderson, 1975).

Slavin and Karweit (1984) reported that a study was conducted which incorporated team learning components and mastery learning components. They reported that no

significant interactions between mastery and team were found, indicating that the addition of the mastery component neither enhanced nor detracted from the effectiveness of the team component (p. 732).

However, in order to promote educational growth within the classroom, teachers must be effective managers.

Evertson and Emmer (1982) proposed thirty-six strategies by which teachers' effectiveness can be enhanced for improved academic success for their students. They stated that teachers can (a) provide assignments for different students, (b) use a variety of materials, (c) make presentations adapted to different ability levels, (d) be receptive to students' input, and (e) encourage group cohesiveness.

Evertson and Emmer concluded that the results of their survey had several implications for teacher training programs and for research on teaching. They stressed that the beginning of the year is a critical time for establishing behavioral patterns, expectations, and procedures that can persist throughout the year for improved academic success. Evertson and Emmer further emphasized that the established behavioral and academic expectations at the beginning of the year should be planned collaboratively with the students so as to be more realistic and more meaningful.

Squires, Huitt, and Segars (1981, p. 174) reported that "student classroom behaviors that indicate involvement,

success, and coverage of appropriate content are most closely linked to student achievement, which is supported by teacher behaviors such as planning, management, and instruction."

Adelman and Taylor (1983) concluded that many students with learning problems are not highly motivated to improve their academic skill deficiencies. They also stated if a greater number of these students are to be helped, it will be necessary to understand better the role motivation plays in the maintenance and correction of behavioral problems.

Trickett and Moos (1973) linked student satisfaction and mood to the social environment of school classes. The study showed that students expressed greater satisfaction in classes where there was a high student involvement and innovative teaching methods. The study also showed that classrooms in which students reported a great deal of content learning reflected a high degree of teachers' concern for the students. Trickett and Moos further stated that classrooms must be intellectually challenging to encourage growth in achievement and understanding as well as cohesive and satisfying to encourage students' interest and motivation.

Fogelman (1978) concluded from a study that there is a straightforward relationship between school attendance, attainment, and behavior in that children with high attendance levels obtain an average higher scores on reading

comprehension and mathematics tests. Bloom's (1976) studies of the relationship between school attendance and achievement have produced mixed results.

Nielsen (1984) stated that rather than teaching adolescents methods for motivating themselves, teachers often use psychology only to make them comply with adult wishes. He further stated that because most youths are accustomed to following the orders of adults, their self-management skills are probably unrefined (p. 32). Nielson emphasized that school districts have the responsibility of ensuring that teachers are effective managers and they are kept abreast of changing techniques in an evolving society. One way to enhance the abilities of teachers is through collaboratively planned and ongoing staff development activities.

# Staff Development

Changing schools requires changing the culture of schools (Purkey and Smith, 1983). School culture, according to Sergiovanni, (1984) includes "values, symbols, beliefs, and shared meanings of parents, students, teachers, and others conceived as a group or community" (p. 9). Louis (1985) emphasized that members of school culture share common values and beliefs. Similarly, according to Lambert (1988), school culture is composed of beliefs, assumptions, and attitudes held by group members for a given period of

time. Sarason (1982) and Feiman-Nemser and Floden (1986) referred to school culture as the overall character of a school. Pfeffer (1982) argued that a school's culture can affect the motivation and committement of its members.

Elmore (1987) reported that the culture of schools affects teachers and students by defining and legitimating inequality which is often referred to as authority. Grant (1985) stated:

Authority in schools is formulated by (a) teacher's predisposition to influence students' learning, (b) students' predisposition toward learning and adult influence, (c) teachers' knowledge and skill in subject matter and pedagogy, (d) the structure of adult work in schools, (e) formal rules governing the structure and operation of schools, and (f) expert knowledge about what should be taught and how. (cited in Elmore, 1987, p. 67)

Lambert (1988) emphasized that the effectiveness of schools is contingent upon a healthy school culture.

Sergiovanni (1984) reported that excellent schools incorporate sacred values and beliefs supporting a vision of excellence. Studies (cited in Papalewis, 1988, pp. 158-159) have shown that effective schools have (a) a clear and vital mission (Boyer, 1983), (b) clearly articulated goals and identities (Lightfoot, 1983), and (c) positive values (Squires, Huitt, and Segars, 1983).

Sergiovanni (1984) further emphasized that effective schools reward and motivate their teachers which will promote the feeling of success. Lieberman and Miller (1984, p. 89) summarized several findings and reported that schools

which have been characterized as "effective schools" share a core of characteristics: (a) a sense of order in the school, (b) high expectations for student achievement, (c) strong leadership from the principal or other staff members, (d) schoolwide control of instructional and training decisions, and (e) clear goals collectively agreed upon (Rutter et al., 1979; Edmonds, 1978; Austin, 1979; Squires, 1980; Brookover and Lezotte, 1979). Similarly, Stefanich (1983) summarized the characteristics of school which are considered effective from his research findings: (a) a consensus exists on curricular goals, (b) the principal exhibits strong leadership; (c) the staff and administration reflect positive expectations, and (d) student assessment is consistent.

Conversely, Morgan (1986) found that less effective schools are composed of school leaders who are problem solvers and operate from a fixed stand point rather than remaining open and flexible to new ideas—status quo.

Parish et al. (1989) hold the belief that these schools present and maintain the economic and social order of the larger culture by passing norms values, practices and traditions to the next generation by sorting students along economic and class lines. Giroux and Purpel (1983) referred to the sorting of students by economic and class lines as the hidden curriculum of the organizational structure of schools. Oakes (1985, p. 175) stated:

Minority children and those from the lowest socioeconomic groups have been consistently found in disproportionate numbers in classes at the lowest track levels and children from the upper socioeconomic levels have been found to be consistently over represented in higher tracks.

Further, Bates (1987) stated that ruling-class schools make adjustments and refinements to acccommodate the educational needs of the ruling class. Where as on the other hand, work-class schools rely on the bureaucratic structures for changes to occur. The Roosevelt School District, New York, for example, is a predominately Black working-class school district with a minimal amount of local financial support. Consequently, the District relys heavily on financial support from state and national funding agencies. Therefore, the outside bureaucratic structures have an inordinate amount of influence on school change as opposed to school districts which rely less heavily on outside agenicies for financial support.

Lambert (1988) argued that movement in a culture is contingent upon (a) creating tension, (b) building from conflict, and (c) equipping teachers to deal with ambuiguity. Futhermore, Ornstein and Hunkins (1988) stated that school leaders must present ideas and values which are accceptable by the group members. Teachers, however, according to Corbert et al. (1987), are inherently and universally resistant to change.

Sarason (1982, p. 96) stated:

Any attempt to introduce change into the school setting requires, among other things, changing the existing regularities in some way. The intended outcomes involve changing the existing regularity, eliminating one or more of them, or prodding new ones.

Corbett et al. (1987, p. 39) reported that several researchers offered reasons for unsuccessful innovative efforts: poor administrative planning and a heavy logistical burden on teachers (Gross et al., 1971), insufficient time to learn new practices and inattention to later stage of the change cycle (Huberman and Miles, 1984), and the need for principals to be more dynamic leaders (Hall et al., 1984).

Garten and Valentine (1989) argued that effective instructional leadership is contingent upon the school leader's ability to involve teachers in focusing on the school's mission and an agreement on the means to achieving established goals. Squires (1980) proposed three leadership variables which usually build and maintain school norms: modeling, feedback and consensus building.

Principals, according to Rutter et al. (1979), are expected to be the tone setter for teachers by supporting inservice activities, monitoring classes, and supervising instruction. Brookover et al. (1979) stated that students' learning is linked to the beliefs of the principal, for example, if the principal does not believe that students

will learn, then the principal probably will not encourage the staff to devote sufficient time to instruction.

Principals can provide teachers with feedback by observing their classes on a regular basis and by conferring with teachers about instructional and inservice concerns (Wynne, 1980). Brookover et al. (1979) reported that effective schools are also characterized by feedback that acknowledges students' academic success and appropriate behavior.

Duckett et al. (1980) and Wynne (1980) contended that appropriate modeling and feedback produced consensus among teachers for desired school change. Smith (1987) emphasized that there is no one way to model. However, Smith further emphasized that a collaborative school is constructed on a set of beliefs and practices designed to achieve defined goals. Smith stated:

The belief that the quality of education is largely determined by what happens at the school site.... instruction is most effective in a school environment characterized by norms of collegiality and continuous improvement....teachers are responsible for the instructional process and accountable for its outcomes....a wide range of practices and structures that enable administrators and teachers to work together on school improvement, and the involvement of teachers in decisions about school goals and the means of implementing them. (p. 5)

Combs (1988) suggested that changing teachers' beliefs require creating an environment conducive for change as opposed to imposing reforms upon teachers. Smith (1987) revealed that "principals of collaborative schools have

often reported that power shared is power gained: the teachers' respect for them grows" (p. 6). Teachers, according to Smith, would assume new responsibilities through the collaborative process, such as setting goals and overseeing their own professional development. Smith, however, emphasized that collaborative schools require leadership styles that are different from those usually found in bureaucratic schools.

Combs stressed that "a major drawback of innovations is the lack of determining what is important" (p. 39). Garten and Valentine (1989) suggested that school leaders establish a committee, for example, an Instructional Improvement Council to assess instructional needs and make recommendations for ongoing staff development, which solicits teachers' imput. Garten and Valentine offered the following as responsibilities of an Instructional Improvement Council: (a) determine the inservice needs of faculty members, (b) assess curricular needs, (c) review current instructional objectives, (d) analyze student achievement, and (e) make recommendations for program improvement. Smith (1987) reported that when school leaders provide teachers opportunities to participate in school change, teachers cooperate more with each other and with the administrators. For instance, Palincscar et al. (1988/1989) reported that the Springfield, Illinois, schools began a collaborative reciprocal teaching program among teachers and administrators which, in addition to instructional chaining, produced peer support groups among teachers and greater understanding and appreciation of teaching and learning by the principals from the teachers' vantage point.

Additionally, Jones and Maloy (1988, pp. 55-56) emphasized that staff development involved four key processes:

- Creates synergy from professional and organizational developments.
   --mutual benefits for those involved.
   --both institutional direction and personal growth.
- 2. Fosters adoption and adaptation.

  --emphasis on communications, group processes,
  connecting diverse groups in new alliances.
- Values characteristics of successfully changing organizations.
   --leadership, goal clarity, high standards and expectations.
- 4. Commits priority resources to equity needs.
  --not to leave anyone out but to recognize a
  special place for traditionally disenfranchised
  groups.

Further, Joyce et al. (1987) suggested that school leaders select staff development programs which promise to substantially increase students' learning and to increase students' interest in learning. Recent research on staff development and curriculum implementation, according to Huberman and Miles (1984), emphasized establishing programs to enable teachers to acquire a repertoire of effective teaching strategies. Fenstermacher and Berliner (1983) further suggested that school leaders planning staff development activities to improve or expand teaching skills consider the teachers' learning styles. For example, Ogle

(1988/1989) reported that the East Meadow, Long Island, teachers at Meadowbrook School designed a staff development project to help their students become strategic learners.

Ogle further reported that the teachers designed activities "taking into account student interest and teacher preferences" (p. 47).

Recent innovations in staff development have come about through university and school partnerships. The Roosevelt School District, Roosevelt, New York, entered into a staff development partnership in August, 1982, with the University of Massachusetts at Amherst whereby teachers and administrators would design, implement, and evaluate staff development projects, while earning an advanced degree—master of education or doctor of education (Jones and Maloy, 1988).

Jones and Maloy reported that the University of
Massachusetts and the Roosevelt Union Free School District's
Staff development partnership:

will provide systematic and coordinated means of involving the professional employees in activities and study to identify inhibitors to wholesome learning, increase expectations, improve supervision/teaching/evaluations, define objectives, and locate resources both within and without the district. (p. 55)

Isher and Leslie (1987) reported that University-based personnel and school leaders have recognized the need to collaborate to strengthen educational programs in both university and school settings. Huling-Austin and

O'Bryan-Garland (1987) reported that Southwest Texas State University implemented a collaborative program with the local school district which has resulted in better communications between the school and university and improved the teaching/learning process in the local schools. Mackenzie and Urich (1984) reported that a collaborative program was established at Florida Atlantic University and a central Florida school district (unnamed) to improve the efficiency of the district's support personnel which also enhanced the district's and the University's public support through the attention which was focused on the programs. Maeroff (1983) concluded from his research that effective collaborative programs between universities and schools must (a) agree that they have common concerns, (b) overcome the traditional pecking order, (c) sharply focus projects, (d) recognize and reward participants, and (e) focus on action, not machinery (cited in Huling-Austin & O'Bryan-Garland, 1987, p. 42).

# Summary

This chapter presented four bodies of literature—adolescents' developmental characteristics, learning styles, teaching styles, and staff development—which supported the four propositions: (a) adolescents are at a stage of high variability and changing identity so that seventh and eighsth grade students—whether tracked or not—include a

wide range of learning styles, (b) if students find appropriate instruction and concerns for themselves as individuals, they will attend class and that attendance should eventually yield academic achievement, (c) a planned staff development project to expand the repertoire of teaching styles to accord more closely, though not necessarily on a one to one basis, with students currently enrolled should increase opportunities for success in school, and (d) a staff development program is more likely to be ongoing if participants are provided opportunities to "buy into" the process through collaborative planning.

Adolescence is characterized by rapid physical growth and hormonal development which is usually compounded by stress, alienation, and identity confusion, which may be manifested by high absenteeism from school/class, poor academic performance, and disruptive behavior. Adolescence is the stage of development which has many implications for educators to diversify their educational programs to accommodate their students varied social and educational needs. Adolescents have a variety of learning styles which are shaped by their environment, emotions, sociological needs, physical characteristics, and psychological inclinations.

Teachers usually teach according to their preferred learning style. Researchers have shown that when teachers acquired and utilized a repertoire of teaching styles in the

classes, student achievement usually increased. However, cooperative learning or team learning often produced more interaction among ethnic groups as well as increased academic achievement.

Implementing effective school change requires changing the culture of schools. Changing school culture necessitates school leaders involving teachers in the change process with clearly defined goals and objectives which have been collaboratively planned by school leaders and teachers. The goal of staff development should ultimately be to increase student achievement.

Recently, university-based partnerships have been implemented to strengthen the communications among universities and schools and to keep school districts abreast of current trends in education.

### CHAPTER 111

#### METHODOLOGY

The project was implemented during the 1987-88 school year at Roosevelt Junior Senior High School in two seventh grade social studies classes, one seventh grade science class, and one eighth grade social studies class, with four volunteer teachers. Other staff members were informed of the nature and purpose of the study on a regular basis. The researcher conducted a series of five staff development workshops during the spring of 1988 based on Fischer and Fischer's (1979) ten identified learning styles and six identified teaching styles and Dunn and Dunn's (1975) one hundred question Learning Styles Inventory (see appendix B).

The workshops addressed the following:

- 1. Designing goals and objectives to address the need for study based on a student questionnaire administered in 1986 and discussions with students, parents, teachers, and administrators
- 2. Introducing teaching styles by Fisher and Fisher (1979)
- 3. Introducing learning styles by Fisher and Fisher (1979)
- 4. Introducing varied teaching strategies

- 5. Reviewing Learning Styles Inventory by Dunn and Dunn (1975)
- 6. Designing various learning activities/projects based on students' learning styles

The researcher used Fischer and Fischer's ten identified learning styles and six identified teaching styles and Dunn and Dunn's one hundred question Learning Styles Inventory as models for this study after reviewing and discussing several other researchers' teaching and learning styles models. Fischer and Fischer's learning styles and teaching styles provided clearly defined characteristics, whereby teachers could categorize their students' learning styles and their teaching styles with more objectivity and less difficulty than with some of the other models which had been reviewed and discussed with teachers. Additionally, Dunn and Dunn's one hundred question Learning Styles Inventory provided teachers more objectivity and less difficulty in identifying their students' learning styles, individually or as a group, based on the tallied results (see appendix C).

The following sample of statements were collected from junior high students in an informal survey, conducted prior to the study:

- 1. All she/he does is sit at her desk.
- 2. She/He gives too many projects.
- 3. I get tired of reading from the book.

- 4. I get tired of watching movies.
- 5. We never go on field trips like the other students.
- 6. Her/His class is so boring.
- 7. All we do is read all the time.
- 8. I wish we would do something different. We just copy from the book.
- 9. She/He talks too much.
- 10. The students are too noisy.

Parents expressed their unsolicited concerns about teaching styles/learning styles as evidenced by the following statements which were made in the presence of the researcher prior to the study:

- I do not want my child in his/her class because too many students fail.
- 2. She/He doesn't care about the students. All she/he wants to do is finish the book.
- 3. My child never does homework in her/his class.
- 4. Do they ever go on field trips?
- 5. Do they ever do projects?
- 6. I don't like the way she/he teaches.
- 7. It's crazy in that class.

Teachers expressed the following concerns:

- 1. I can't teach slow students. I am not a special education teacher.
- 2. Some of these students can't learn.
- 3. They need to be in special education.

- 4. They keep me busy all the time.
- 5. My students become bored so easily.
- 6. I become so frustrated when they don't understand after I have explained something several times.
- 7. It's really difficult to plan for my classes. They don't seem interested.

Administrators were often heard by the researcher expressing the following concerns:

- 1. Too many students are cutting classes.
- 2. Too many students are failing.
- 3. Too many students are repeating the 7th and 8th grades.
- 4. Too many students are repeating for the third time.
- 5. Some of the teachers don't seem to be meeting the needs of the students.
- 6. Some of the students seem to be turned off by the teachers.
- 7. What are possible options that we can offer our students?
- 8. Maybe we need to restructure the junior high.

Students, parents, teachers, and administrators
expressed a need for teachers to vary their teaching methods
on the junior high level to increase student achievement,
increase attendance, and improve students' attitudes about
school. In order to design a staff development program
which would be relevant for students and teachers, the

researcher administrated a needs assessment survey to seventh grade students and a teaching styles survey to junior high and special education teachers. The results from both instruments were categorized and prioritized in order of frequency distribution. Workshops were designed based on items which occurred most frequently. An extensive review of literature on learning styles/teaching styles was used to provide participants with current and relevant information relative to expressed concerns.

# Involving Others in Planning

The idea of teaching styles/learning styles for improved academic achievement, improved attendance, and improved attitudes did not take place in a vacuum.

Students, parents, teachers, and administrators collaborated among themselves and with the researcher to help construct an effective staff development program on teaching styles/learning styles to address their concerns. Teachers, especially, had direct impact on the planning and implementation of workshops and class activities because of their expressed desire to effect change in students.

Workshops were designed to provide participants data from experts, to share experiences regarding teaching styles/learning styles from a professional and a personal

perspective, and to help teachers acquire a repertoire of teaching styles.

The researcher sought students' ideas by asking them to complete a five question survey form designed by the researcher titled, "How Can Your Teacher Make Learning More Interesting?" Students provided an array of suggestions ranging from "I don't know" to "She could give us more experiments and projects or maybe even go on a fleld trip to a science fair or a museum. I feel we should expand our horizon on science." Seventy-nine seventh and eighth graders were surveyed. To compare the effects of employing varied styles with students of varied learning abilities, the participating teachers chose the following classes to provide a range of academic abilities: one seventh grade advanced placement science class, one seventh grade above average social studies class, one average seventh grade social studies class, and one eighth grade below average social studies class.

In providing for shared experiences, the researcher sought ideas from others by asking members of the junior high and special education faculty to complete a teaching styles survey, which was composed of fifteen questions with six major categories, and to suggest areas about which they wanted more information. Teachers also shared their teaching strategies with the researcher and with the participants. Teachers who were not active participants

in the workshops collaborated frequently with the participants to acquire teaching strategies. The researcher believed a secondary outcome of this project was the reduction of the fear among teachers to consult fellow teachers for suggestions when their teaching strategies did not seem to work.

The researcher surveyed twelve junior high teachers and ten special education teachers to determine if they would 1) complete a teaching styles survey, 2) participate in staff development workshops, 3) share opinions with the researcher, and 4) complete evaluation forms. Seven junior high teachers and six special education teachers consented to participate in staff development workshops on teaching styles/learning styles. Due to the following constraints, only four junior high teachers participated in the workshops and in the implementation of the study:

- 1. According to the teachers' contract, teachers must be paid at an hourly rate of \$8.00 (eight) to attend afterschool workshops, which are in addition to the two regularly scheduled meetings per month.
- Several junior high teachers chose not to participate in staff development activities.
- Of the thirteen consenting teachers, the largest number of junior high teachers who had common planning periods was six.

- 4. Junior high special education teachers did not have planning periods when the workshops were offered.
- 5. One teacher refused to relinquish his planning period to attend workshops.
- 6. An administrative supervisor would not provide relief time for one teacher to participate in workshops.

The researcher involved four junior high teachers in five workshop training sessions as a need to acquire a repertoire of teaching styles based on an extensive review of current and relevant literature, students', parents', teachers', and administrators' concerns regarding students' success in school, and observations of teachers imparting knowledge to students in their classrooms.

Permission to conduct the study was obtained from the principal and the administrative supervisors of social studies and science. The teachers signed consent forms (see appendix D) giving the researcher permission to use their information in his dissertation.

A teaching styles survey (see appendix E) was administered to twelve seventh and eighth grade teachers and ten special education teachers at Roosevelt Junior Senior High School during the fall of 1987. Of the twelve junior high teachers who completed the teaching styles survey, four teachers, whose names were not used, participated in five

staff development workshops on learning styles and teaching styles from February 1, 1988 to April 4, 1988.

The participating teachers, four Black females, ages thirty-three to fifty-three years, taught seventh and eighth grade social studies and science. Participant #1 is a fifty year old Black female seventh grade social studies teacher. Her undergraduate major is social studies and her graduate major is reading. She has been teaching fourteen years. Participant #2 is a thirty-three year old Black female seventh grade social studies teacher, with a Bachelor of Science degree in history and a Master of Science degree in historical writing. She has been teaching eight years. Participant #3 is a fifty-three year old Black female eighth grade science teacher, with an undergraduate major in biology and a graduate major in education. She has twenty-two years of teaching experience. Participant #4 is a thirty-eight year old Black female eighth grade social studies teacher. She has an undergraduate and a graduate major in sociology. She has been teaching ten years.

The teachers selected four classes to participate in the study, one "AP" class (Advanced placement), one "A" class (above average), one "B" class (average), and one "C" class (below average), to include a wide range of learning styles, academic abilities, interests, and performance levels.

There were sixty-four seventh grade and fifteen eighth grade boys and girls, ages twelve to seventeen years. Ethnically, there were seventy-five Black students, three Hispanic students, and one White student who were from low to middle socioeconomic backgrounds in a suburban/urban community.

Teachers administered a five question researcherdesigned questionaire to the students in April, 1988 which posed the following questions:

- 1. Is the classwork usually interesting?
- 2. Do you expect to pass this class this quarter?
- 3. Are you usually given interesting assignments or activities by your teacher?
- 4. Do you like coming to this class?
- 5. What could your teacher do differently that would help you improve your learning in this class?

The students were administered a Learning Styles

Inventory by Dunn and Dunn (1975) by the teachers during the first week of the study. The teachers reviewed the surveys and designed varied lessons to accord more closely with students' learning styles.

The teachers provided the researcher with the following data:

1. Attendance of each student in the participating classes, for the first, second, third, and fourth quarters of 1987-88

- 2. A breakdown of academic grades for the first, second, third, and fourth quarters for 1987-88, for example, number of students earning grades between 90-100 or 80-89
- 3. Written or oral assessments from participating teachers regarding the study, for example, problems and/or rewards of implementing the study into classes, increased student enthusiasm, additional planning required, etc.
- 4. A copy of the grade reports, which included students' grades for attitude (see appendix F).

This study was implemented for five weeks during the fourth quarter of the 1987-88 school term. The classes met five times per week for thirty-eight minute sessions.

At the end of the five week implementation period, the teachers administered a post-student questionnaire to assess students' attitudes about learning. Teachers also reported students' attendance, attitudes, and grades to the researcher from their grade books. The researcher analyzed the attendance, attitudes, and grades of the first three quarters of 1987-88 and compared students' grades, attitudes, and attendance during the study.

The four participating teachers were administered a teaching styles survey by the researcher at the beginning of the study. The data from the teaching styles survey was

used to help formulate objectives and activities of the workshops.

The participating teachers provided the researcher with oral or written assessments of the study at the end of the five week implementation period. The teachers reported any problems, rewards, or suggestions for implementing varied teaching strategies into the classroom in the seventh and eighth grades.

The following procedures were used to gather data:

- Researcher selected teachers from a pool of volunteers.
- 2. Teachers administered learning styles survey to students.
- 3. Researcher conducted five workshops for teachers.
- 4. Teachers designed and implemented individual learning activities/projects to accord more closely with students' learning styles.
- 5. Teachers observed each other two or three times to modify, design, and implement teaching strategies in their classes.
- 6. Researcher met with students individually and in small groups to get oral and written feedback.
- 7. Researcher monitored students' attendance to classes.
- 8. Researcher compiled and analyzed data (students' attendance, students' grades, students' attitudes, and oral and written feedback from teachers).

The researcher chose to use action research, as opposed to other research models, because of its characteristics, which were easily adapted to this study. Cohen and Manion (1985, p. 208) defined action research as

situational—it is concerned with diagnosing a problem in a specific context and attempting to solve it in that context; it is usually collaborative—teams of researchers and practitioners work together on a project; it is participatory—team members themselves take part directly or indirectly in implementing the research; and it is self—evaluative—modifications are continuousyl evaluated within the ongoing situation, the ultimate objective being to improve practice in some way or other.

Formative and summative evaluative approaches were used to assess this study. Formatively, the procedures were modified to accommodate feedback from the participants when it was in accord with the goals and objectives of the study. And, summatively, students' grades, attitudes, and attendance to class were collected by the teachers and reported to the researcher for analysis during the 1987-88 school term.

All junior high teachers, including special education, were asked to complete a teaching styles survey and list areas about which they wanted more information, which helped direct the focus of the workshops.

After each workshop, the participating teachers completed assessment forms. They also gave oral feedback which was recorded by the researcher. Upon completion of the workshops, the teachers implemented various teaching strategies in their classes, which they had agreed to

assess by comparing students' grades, attendance to class and attitudes. They also administered a survey to assess students' attitudes about learning and their learning styles.

Five staff development workshops were conducted by the researcher from February 1, 1988, to April 4, 1988, in room #205 at Roosevelt Junior High School from 1:47 p.m. to 2:25 p.m. (see table 4). The workshops were designed to help junior high teachers acquire a repertoire of teaching styles and to assist teachers in designing lessons and activities to accord more closely with students' learning styles.

After the final workshop, the teachers summarized their experiences in narrative form. The teachers reflected on (a) students' attitudes about learning, (b) their attitudes about different teaching approaches, (c) fellow teachers' attitudes about different approaches to teaching, (d) amount of additional time needed for planning lessons, (e) increased enthusiasm among students about class assignments and lessons, (f) their reluctance to try new teaching strategies, and (g) suggestions for future workshops.

TABLE 4

Staff Development Workshops on Teaching Styles/Learning Styles February 1, 1988 - April 4, 1988 Roosevelt Junior Senior High School Roy R. Taylor, Researcher

Workshops	Dates	Activities
#1	2/1/88	<ol> <li>Outlining Participants' Roles</li> <li>Goal Consensus</li> <li>Need for Study</li> <li>Analyzing and Prioritizing Needs</li> <li>Discussion on Teaching Styles/ Learning Styles</li> </ol>
#2	2/9/88	<ol> <li>Reviewing Results of Teaching Styles Survey</li> <li>Acquisition of Teaching Styles</li> <li>Discussion on Teaching Styles</li> </ol>
#3	3/7/88	<ol> <li>Identification of Participants'         Learning Styles</li> <li>Discussion on Participants'         Learning Styles</li> </ol>
#4	3/23/88	<ol> <li>Designing Projects/Activities</li> <li>Discussion on Projects/Activities</li> </ol>
#5	4/4/88	<ol> <li>Role-Playing Teaching Styles</li> <li>Role-Playing Learning Styles</li> <li>Learning Styles Inventory</li> <li>Implementation Timeline</li> <li>Collecting Data</li> </ol>

# Workshop #1

# February 1, 1988

This workshop began with the researcher thanking the junior high teachers for agreeing to participate in staff

development workshops on teaching styles/learning styles. The researcher shared with participants how he became interested in the project. He stated that he observed during the fall of 1984, which was his first year in Roosevelt, a large number of junior high students failing classes or repeating grades and generally displaying low self-esteem and negative feelings about school.

Participants browsed through the teaching styles/learning styles literature, which had been displayed by the researcher. Participants also chatted about various situations they had encountered in their classrooms during the day. One eighth grade participant asked another, "How did you get Johnnny (not his real name) to do any work for you last year?" The other participant responded, "You have come to the right place to get that question answered, honey."

The workshop was designed to outline participants' roles in the staff development project, to discuss and analyze need for study, and to generate discussions on teaching styles/learning styles. The workshop was composed of five specific objectives and five corresponding activities.

#### Objectives

Specific objectives for workshop #1 were the following:

1. To outline participants' role in the project

- To begin building goal consensus and goal clarification
- 3. To discuss need for study
- 4. To analyze and prioritize needs
- 5. To generate dialogue on teaching styles/learning styles

### Procedures

The following activities corresponded to the objectives by number:

## Activity #1--Outlining Roles

Discussion among the researcher and the participants generated seven responsibilities.

- To serve as a planning group and as workshop participants
- To provide input in analyzing and prioritizing the results of students' assessment forms
- 3. To categorize and analyze students' learning styles
- 4. To acquire a repertoire of teaching styles
- 5. To investigate and evaluate curricular materials
- 6. To modify curricular materials to accord more closely with students' learning styles
- 7. To share information and ideas with other teachers

# Activity #2--Goal Consensus

The researcher and the participants agreed that the following were realistic goals:

- To promote positive attitudes and positive work habits among junior high students
- 2. To create a harmonious working relationship among junior high teachers
- 3. To bridge the gap from elementary to high school for junior high teachers
- 4. To increase communication among students, parents, teachers and administrators

## Activity #3--Need for Study

This activity generated much discussion among participants and the researcher. The participants discussed a range of needs based on teacher and student surveys and discussions with students, parents, teachers, and administrators. The following is a compilation of needs which were generated by the dialogue among the participants and the researcher:

- To provide teachers with a repertoire of teaching styles
- 2. To increase teachers' awareness of students' learning styles
- 3. To improve students' attitudes about learning
- 4. To improve teachers' attitudes about teaching
- 5. To increase students' awareness of their learning styles
- 6. To increase teachers' awareness of their learning styles

- 7. To improve parents' attitudes about schools
- 8. To improve administrators' attitudes about teachers
- 9. To improve students' academic achievement
- 10. To improve students' attendance
- 11. To reduce teacher burnout
- 12. To improve school climate
- 13. To reduce competition among students
- 14. To reduce competition among teachers
- 15. To increase collaboration among students
- 16. To increase collaboration among teachers

## Activity #4--Analyze and Prioritize Needs

The participants experienced much disagreement regarding the order of importance of the needs based on their ideas about the role of schools. After sharing opinions, the participants suggested the following order to address the needs:

- To design activities to increase teachers' awareness of students' learning styles
- 2. To design activities to provide teachers with a repertoire of teaching styles
- 3. To design activities to increase teachers' awareness of their learning styles
- 4. To design activities to increase students' awareness of their learning styles

The researcher and the participants believed that other needs, which were listed in Activity #1, would emerge if the aforementioned activities were implemented and monitored closely.

# Activity #5--Teaching Styles/Learning Styles

This activity was included to give participants the opportunity to express personal attitudes towards teaching styles/learning styles. Discussion centered around resistance which may be encountered by participants who did not want to change their teaching style(s). One participant also discussed her reluctance to incorporate "something new" into her classes because it may impede her ability to complete state curricular mandates.

Implementation of varied teaching styles to accord more closely to the learning styles of students generated concerns among the participants. The participants believed they would be expected to perform the following tasks: to design individual learning styles assignments for students, to design individual examinations for students, to design individual examinations for students, and to teach directly to the learning styles of students. The researcher re-emphasized that the participants would be expected to develop a repertoire of teaching styles which would provide them with more opportunities to address the varied learning styles of their students, not Individualized Educational Plans (IEP's).

# Feedback for Workshop #1

In the absence of a formal objective assessment form, the researcher asked the participants to assess workshop #1 in a narrative form. The researcher gave the participants the option of providing their names and the grades and/or subjects they taught, which would not be used in reporting the results of the assessments.

Because of time restraints, the participants did not complete their assessments at the end of the workshop. The researcher asked participants to return assessments to him the next day. Three assessments were turned in the next day and one was turned in two days later.

The following are the narrative assessments of workshop #1:

The workshop was interesting. I had not given much thought to teaching styles and learning styles before. I am not sure if time will allow me to individualize my lessons because of the State's mandates. I also don't think I can individualize in my classes because they are too large.

I just love the idea of learning styles because my students really enjoy doing projects, making speeches, etc. This will provide me with different ways to construct my lessons. I think we should always look for new ways to help our students improve.

I think the workshop was very informative. I am willing to try new techniques which will help my students learn. I have tried every technique that I know, and my students still seem bored. If this will help me meet the needs of my students, you have my support. I will let you know if it works in my classes.

I think this is a great idea. I have always used different approaches in my classes. This will be an added dimension. I think we should explore all the capabilities and interests of our students.

The participants expressed their enthusiasm about attending the workshops, their eagerness to learn different teaching strategies, and their desire to motivate their students to learn. The agenda seemed appropriate because participants did not discuss displeasure with any of the agenda items.

The assessments did not reveal logistical concerns.

Thus, the researcher did not make plans to change the length of the workshops (thirty-eight minutes), the place (room #205 in the junior high wing), nor the time (1:47 p.m. to 2:25 p.m.).

#### Workshop #2

## February 9, 1988

This workshop opened with the researcher reviewing objectives and activities from workshop #1 and summarizing feedback from participants. Participants shared various disruptive situations which had transpired in their classes during the past week and various strategies they had used to restore order for approximately five minutes before the researcher directed them toward their focus for today—to discuss their teaching styles. The workshop was designed to review participants' teaching styles, to discuss how participants acquired their teaching styles, and to present

Fisher and Fisher's (1979) identified teaching styles. The workshop was composed of three objectives and three corresponding activities.

## Objectives

Specific objectives of workshop #2 were the following:

- To review results of researcher designed Teaching
   Styles Survey with participants
- 2. To provide opportunities for participants to discuss how they acquired their teaching styles
- 3. To present Fisher and Fisher's (1979) six identified teaching styles

### Procedures

The following activities corresponded to the objectives by number:

## Activity #1--Reviewing Results of Teaching Styles Survey

This session began with the researcher sharing the results of the Teaching Styles Survey which was administered to twelve junior high and ten special education teachers during the fall of 1987. The emphasis of the survey was on teacher methodology. The following is a tally of the results of the survey:

1. I plan my lessons Always Frequently Sometimes Never with equal concern 11 10 1 0 for students and curricular objectives.

I provide an Always Frequently Sometimes Never atmosphere that . 7 allows my students to pursue activities that interest them. I plan my class Always Frequently Sometimes Never activities with 5 9 student input. 4. I demand that my Always Frequently Sometimes Never students perform 5 2 academically at a specific level. 5. I must finish Always Frequently Sometimes Never teaching the 2 2 4 14 textbook in order to meet curricular requirements. 6. I prescribe Always Frequently Sometimes Never materials for a 10 student to learn based on his/her learning style. 7. Always Frequently Sometimes Never I become 10 emotionally involved in my teaching. Always Frequently Sometimes Never I use different 8. 11 techniques to involve all students in class activities. Always Frequently Sometimes Never 9. My students ask 0 10 3 questions for clarification.

10. My students complain of 0 0 16 2 boredom.

11. My students complete their nomework assignments.

Always Frequently Sometimes Never 0 7 10 0

- 12. I rephrase questions Always Frequently Sometimes Never or statements when 14 3 2 0 my students do not understand.
- 13. I use the same Always Frequently Sometimes Never materials with all 2 5 10 3 students.
- 14. Individualization Always Frequently Sometimes Never enables me to 4 12 6 0 involve more students in class.
- 15. I would attend Yes No workshops that 13 9 would provide me with a variety of teaching techniques.

The following is a list of topics about which teachers wanted additional information if they attended workshops:

- 1. All styles of modern music dating from about 1930 to present, i.e., swing, fiftles, peace movement, flower power, 70s hard rock, punk, disco, new wave, rap, reggae, etc.
- 2. I would attend, but because of other academic and athletic commitments, I do not have the time at this point.
- 3. Teaching reading, speaking the English language, and writing improvements.
- 4. I am always open to new learning styles and techniques.
- 5. Almost any professional secondary topic provided there is a follow-through on all resolutions.
- 6. Motivating students.
- 7. Math and technology.
- 8. Information concerning the student who has not been categorized as special education, but who needs something special in terms of learning with the "normal" group setting. Different learning patterns. Giving test for results or giving information for good test results.

- 9. About subject area. About new state requirements for Social Studies test.
- 10. Ways students learn. Behavioral modification. Special Education.
- 11. I would like to find some ways to keep my students interested in the materials that I must cover to get them ready for the State Examinations.
- 12. Program instructions. How to deal with the pathology that is now rampant in public education for which teachers were not trained.
- 13. How to teach the child who is unwilling to learn. How to improve the percentage of students doing their homework assignments.
- 14. How does one motivate the non-academic student?
  Creative teaching for the potentially motivated.
  Classroom control. In an effort to get through the material, I feel I sacrifice both ends of the spectrum--those who would enjoy a new learning experience and those who need to be gently and slowly brought up to their potential.
- 15. Team approach to managing students. How to enable all of us to work together in a cohesive multifaceted plan to help as many students as possible to learn.
- 16. Developing school and individual pride and self-discipline.
- 17. Individualized instruction.
- 18. Students with language barriers because of a second language. Teaching students waiting to be placed in Special Education. Best utilization of time for the teacher, student, and the class.
- 19. I would like to discuss some of these topics with my fellow teachers at a public forum. That would really be helpful.

The results of the survey were not consistent with concerns which junior high and special education teachers had informally expressed with the researcher in the initial planning of staff development workshops on teaching

styles/learning styles. According to the survey results, most teachers seemed to employ ideal teaching styles/learning styles strategies in their classes, but their oral and written comments were not as idealistic.

Activity #2--Acquisition of Teaching Styles

Participant #1 stated that she patterned her teaching styles after a "handsome" junior high teacher. She stated that he was the only teacher who really motivated her to learn. She stated that she was not really aware of the teaching styles he used to motivate her to learn, but she believed it was because of his "good looks." She stated that she completed all assignments, attended all classes, and performed well on tests because she wanted to satisfy him. She also stated that when she became a teacher she wanted to teach the way he taught. She did state that there were not any clearly defined teaching styles that he exhibited. He did involve students in the planning of lessons and designing lessons and activities according to students' needs and desires. Participant #1 stated that she also involves students in the planning process.

Participant #2 stated that she acquired her teaching styles from a college professor, whom she admired for his ability to make lessons come "alive", regardless of the topic. She also stated that prior to having taken a class from him during the second semester of her freshman year, she completed assignments only because they were required,

not because they were interesting or relevant. She stated that the professor was a great influence also in her decision to become a teacher. She had previously viewed teachers as those who "feed" students information and students would "regurgitate" it on examinations.

Participant #2 adds drama into her lessons to make them come alive as her college professor did.

Participant #3 stated that a high school science teacher and a college science professor influenced her teaching styles. She stated that both individuals were task oriented and good disciplinarians. They had little tolerance for things that deviated from the norm. They believed if they had a task to complete, all other things became secondary until the task had been completed. They also did not tolerate students who did not conform to the "standards of behavior." Participant #3 believed that these two individuals were role models for their students and other teachers because they produced outstanding academic and behavioral changes among their students. Participant #3 stated that she exhibits many of the traits of her two mentor teachers in her day-to-day interaction with students. She also stated that she does not believe that her styles are necessarily the best for other teachers, but they have worked for her for many years. She stated that she would like to become more student oriented, but she is afraid that the students will view her as becoming "easy."

Participant #4 stated that she cannot focus on any particular individual who may have influenced her teaching styles. She stated that she patterned her teaching styles after many teachers, especially those who were emotionally exciting. She stated that she likes to get her students "emotionally worked up" for a lesson. She also stated that she likes to pattern her lectures after southern Black preachers so she can really get the students mentally involved. She stated that she likes to move around the room a lot. She feels like this is coming down from the pulpit and mingling with the congregation to let them know she is a part of them.

## Activity #3--Teaching Styles

What influences teaching styles? It has been suggested that one's learning preference influences one's teaching style (Gepheart, Strother and Duchett, 1980).

If teachers, like other people, have experienced ways of perceiving and processing information, then it is reasonable to believe that teachers will communicate their subject matter in ways that are most compatible with their learning styles. A broader understanding of individual's learning styles and preferred learning conditions ought to underscore the need for and approach to teaching, one that provides for a variety of learning styles (Heikkinen, Pettigrew and Zakrajsek, 1985).

The following list of six identified teaching styles by Fisher and Fisher (1979, p. 251) was provided to give participants the opportunity to generate discussion about their teaching styles:

- 1. The Task-Oriented--These teachers prescribe materials to be learned and demand specific performance on the part of the students.
- 2. The Cooperative Planner--These teachers plan the means and ends of instruction with student cooperation.
- 3. The Child Centered--These teachers provide a structure to pursue whatever they want to do or whatever interests them.
- 4. The Subject Centered-These teachers focus on organized content to the near exclusion of the learner.
- 5. The Learning Centered-These teachers have equal concern for the student and for the curricular objectives, the materials to be learned.
- 6. The Emotionally Exciting and Its Counterpart--These teachers show their own intensive emotional involvement in teaching.

# Feedback for Workshop #2

The four participants felt that the activities for this session were appropriate. They particularly enjoyed discussing the results of the researcher designed Teaching Styles Survey and the opportunity to discuss how they acquired their teaching styles.

Participant comments were the following:

- I really never thought about how I acquired my teaching style. This workshop helped me learn about myself.
- 2. I think I need to become more accommodating of my students' needs.

- 3. The interaction among us was great.
- 4. We need more sessions like this.

## Workshop #3

## March 7, 1988

This workshop began approximately ten minutes late because one participant had an unscheduled parent conference that her supervisor arranged without consulting her. During this time, the remaining three participants shared opinions regarding students' success in school and teachers' roles in helping students achieve. Participants stated that it was a difficult transition for them to employ different teaching techniques in their classes because they feel uncomfortable trying new ideas. They stated that they feel "in charge" when they are using their "traditional" strategies.

The researcher formally began the workshop when the fourth participant returned. The researcher distributed a list of learning styles for discussion. The workshop was composed of two objectives and two corresponding activities. Objectives

Specific objectives of workshop 3 were the following:

- 1. To identify participants' learning styles based on Fisher and Fisher (1979)
- To generate discussions regarding participants' learning styles

## Procedures

The following activities corresponded to the objectives by number:

# Activity #1--Participants' Learning Styles

This activity began with the participants reviewing the ten learning styles by Fisher and Fisher (1979, pp. 246-250).

- 1. The Incremental Learner--These students proceed step-by-step fashion, systematically adding bits and pieces together to gain larger understanding. . . .
- 2. The Intuitive Learner--The learning style of these students does not follow traditional logic, chronology, or step-by-step sequence. . . .
- 3. The Sensory Specialist--These students rely primarily on one sense for meaningful formation of ideas. . . .
- 4. The Sensory Generalist--These students use all or many of the senses in gathering information and gaining insight. . . .
- 5. The Emotionally Involved--These students function best in a classroom in which the atmosphere carries a high emotional charge. . . .
- 6. The Emotionally Neutral--These students function best in a classroom where the emotional tone is low-keyed and relatively neutral....
- 7. Explicitly Structured-These students learn best when the teacher makes explicit a clear, unambiguous structure for learning. . . .
- 8. Open-Ended Structure--These students feel at home and learn best in a fairly open-ended environment. . . .
- 7. The Damaged Learner-These students are physically normal yet damaged in self-concept, social competency, aesthetic sensitivity, or intellect in such a way that they develop negative learning styles. . . .
- 10. The Eclectic Learner--These students can shift learning styles and function profitably and may find one or another styles more beneficial, but can adapt to and benefit from others. . . .

After participants had reviewed and discussed the learning styles, the researcher asked them to choose the styles which most closely identified their learning

style(s). Participants chose at least two learning styles to identify their preferred ways of learning. They stated that at times they learned in other ways, but they believed the following learning styles best represented the ways they learn in most situations.

- 1. Participant #1--Seventh grade social studies teacher
  - A. Open-ended structure
  - B. Emotionally involved
- 2. Participant #2--Seventh grade social studies teacher
  - A. Intuitive learner
  - B. Open-ended structure
- 3. Participant #3--Seventh grade science teacher
  - A. Explicitly structure
  - B. The incremental learner
  - C. Emotionally neutral
- 4. Participant #4--Eighth grade social studies teacher
  - A. Emotionally involved
  - B. Eclectic learner

# Activity #2--Discussions on Participants' Learning Styles

This activity was designed to give the participants opportunities to generate discussions about their learning styles—how they acquired their learning styles, how their learning styles affected their success in school, and how their learning styles influenced their career choices.

The participants generally agreed that they did not choose a learning style. They believed they acquired their

learning styles according to circumstances. If a classroom were well-structured, they worked in a well-structured setting. If a classroom were emotionally charged, they worked in an emotionally charged atmosphere. The participants stated that the ways they learned as adolescents did not necessarily represent their preferred learning styles, but they were expected to adapt to teachers' teaching styles.

The participants revealed varied responses regarding their learning styles on their success in school. Three participants generally agreed that they excelled academically in school because their learning styles allowed them the flexibility to adjust to most school and classroom situations.

Participant #3, on the contrary, stated that she did not experience outstanding success in most of her classes because they were not explicitly structured. She stated that she usually did not participate in class discussions because she was a little shy. She usually did not receive extra credit for class participation. She did excel in her science class where all experiments followed a step-by-step sequence and the class was well-disciplined.

Three participants stated that they chose social studies as majors in college because the subject is open-ended. Students can read excerpts from history or sociological documents and express their varied opinions.

The three participants stated that other professions would have offered them open-ended opportunities, but they chose teaching because they enjoyed interacting with adolescents.

Participant #3 chose to become a science teacher because the step-by-step scientific method accorded more closely with her learning styles--explicitly structured, the incremental learner, and the emotionally neutral. She also stated that there were other careers which could have afforded her similar opportunities, but she enjoyed working with junior high students.

## Feedback for Workshop #3

The participants stated that the learning styles' identification activity and discussions that followed were very useful. Three of the four participants stated that if they fully understood their learning styles themselves, they could better determine the effects on their students' learning.

Participant comments were the following:

- Hands-on activities might help us get the feel of what we have been discussing in the previous sessions.
- 2. I am getting so much information that I don't know how to use it. I need time to "digest" it. Maybe we could reschedule next week's workshop.

## Workshop #4

March 23, 1988

The researcher began the workshop by reviewing the activities of workshop #3. He also discussed feedback provided by the participants. The participants expressed a need to "take a break" from the workshops temporarily. They stated that they had been provided more information than they could "digest" in a short period of time. The researcher changed the focus of workshop to incorporate hands-on activities, utilizing the concepts to which participants had been exposed in previous sessions.

The focus of the workshop was to design learning styles projects/activities and to generate discussion among participants relative to participant-designed projects/activities. The workshop was composed of two objectives and two corresponding activities.

## Objectives

Specific objectives for workshop #4 were the following:

- To design projects/activities for each of the ten identified learning styles according to participants' respective academic disciplines
- To generate discussions regarding the participant-designed projects/activities

#### Procedures

The following activities corresponded to the objectives by number:

# Activity #1--Designing Projects/Activities

The researcher asked the four participants to design projects/activities for each of the ten learning styles. This activity generated forty projects/activities, ranging from drawing a map of New York with red, white, and blue chalk on a 3'X 5' white poster-explicitly structured-to constructing a model of a futuristic city-open-ended. Activity #2--Discussion on Projects

# The participants expressed some frustration while they were completing this activity because they stated that designing projects for the ten learning styles was time consuming and required a thorough understanding of the learning styles. The researcher impressed upon the participants that they probably would not need to design activities for ten learning styles in their classes because,

based on previous surveys, usually four or five learning

# Feedback for Workshop #4

styles were more prevelant than others.

Participants felt that designing hands-on activities/projects were extremely useful because it provided them opportunities to view learning styles' activities in "real life." They stated that the discussions which were generated about the projects/activities were useful because participants shared ideas which other

participants could use with their students to create cross-disciplinary activities/projects.

### Workshop #5

## April 4, 1988

The researcher began the session with a review of the assessment forms for workshop #4 which helped structure this workshop. Participants had expressed an interest in role-playing teaching styles/learning styles to help them with their comfort level in the classroom. Participants asked the researcher for more clarification regarding collecting data and the implementation timeline.

The focus of the workshop was role-playing, introducing a Learning Styles Survey, designing an implementation timeline, and providing a format for collecting data. The workshop was composed of five objectives and five corresponding activities.

## Objectives

Specific objectives for workshop #5 were the following:

- 1. To role-play varied teaching styles
- 2. To role-play varied learning styles
- 3. To introduce Learning Styles Inventory by Dunn and Dunn (1975)
- 4. To plan implementation timeline for introducing teaching styles/learning styles into the classes

5. To provide participants with a format for collecting data: attendance, grades, and attitudes

## Procedures

The following activities corresponded to the objectives by number:

# Activity #1--Role-Playing Teaching Styles

This activity involved the four participants in presenting short lessons to other participants and the researcher. Each participant conducted a five minute lesson based on the participant's teaching preference(s). This session also involved each participant conducting a short lesson incorporating a repertoire of teaching styles.

## Activity #2--Role-Playing Learning Styles

Activity #1 and activity #2 were conducted concurrently. The two activities had been planned as separate activities, but as activity #1 began to unfold, participants began to assume certain learning styles—some of their own and others they acquired for the activity.

Activity #3—Learning Styles Inventory by Dunn and Dunn

The researcher introduced the one hundred item Learning Styles Inventory to the participants for discussion regarding administering the survey to their students.

Participants expressed concerns relative to the large number of items on the survey and the repetition of items.

The researcher impressed upon the participants that students should be monitored closely for boredom or fatigue

when taking the survey, and the time should be adjusted accordingly. The researcher stated that many items were repeated on the survey to check for consistency among responses.

# Activity #4--Implementation Timeline

The following timeline for implementing teaching styles/learning styles into class lessons and activities were collaboratively planned by the researcher and the participants:

Tuesday, April 12, 1988	Choose class to participate in study
Wednesday, April 13, 1988	Administer student questionnaire
Monday, April 18, 1988	Administer Learning Styles Survey to students
Monday, April 25, 1988 - Friday, May 27, 1988	Implement Learning Styles/Teaching strategies into class activities
Friday, May 27, 1988	Administer post-student questionnaire

# Activity #5--Collecting Data

The researcher asked the participants to collect or provide the following data to substantiate the study:

 Attendance of each student in the participating classes for the first, second, third, and fourth quarters of 1987-88 school term

- 2. A breakdown of the grades for the first, second, third, and fourth quarters, for example, number of students earning grades between 90-100, 80-89, and so on for the 1987-88 school term
- 3. Written or oral assessments from participating teachers regarding the study, for example, problems and/or rewards of implementing study into classes, increased student enthusiasm, or additional planning required.
- 4. A copy of the grade reports, which included the students' grades for attitude

## Feedback for Workshop #5

Participants stated that the objectives and activities this session were very helpful. They felt that the role-playing activities on teaching styles and learning styles provided them a more "realistic" feel of what they were expected to do in their classrooms with their students. They also stated that the implementation timeline and the format for collecting data clarified dates and provided them with a sense of direction. They stated that they knew when they were expected to perform certain tasks and what information they were expected to collect. Participant comments were the following:

- 1. I found these workshops to be very enlightening.
- 2. I can use everything I learned with my students.

- 3. I have a better understanding of how I learn as well as how my students learn.
- 4. Other teachers should be exposed to teaching styles/learning styles.

#### Summary

Five staff development workshops on teaching styles/learning styles were conducted at Roosevelt Junior Senior High School for four voluntary junior high teachers from February 1, 1988, to April 4, 1988. Opportunities were provided for participants to acquire a repertoire of teaching styles to accord more closely with students/ learning styles through reviewing literature, discussions, exchanging ideas, and role-playing. The researcher and participants collaboratively planned objectives and designed corresponding supporting activities for the workshops.

The enthusiasm of the participants indicated that they had bought into the program. They agreed to serve as ambassadors and to share their techniques and materials with other teachers, who wanted to try different teaching strategies with their students.

The results of the students' pre- and post-attitudinal surveys and a comparison of students' academic grades, attendance to class, and attitudinal ratings for the four

classes for four quarters during the 1987-88 school term are presented in Chapter 4.

#### CHAPTER 1V

#### RESULTS

The results of this study will be presented in this chapter. The purpose of this study is to document the process of designing, implementing and evaluating a staff development program to address seventh and eighth grade students' diverse learning styles and junior high teachers' varied teaching styles with the intent of improving students' academic achievement, attitudes, and attendance to class in four selected classes at Roosevelt Junior Senior High School during the 1987-88 school term.

Four propositions were developed to address adolescents' diverse learning styles and junior high teachers' varied teaching styles. First, students in early adolescence are at a stage of high variability and changing identity so that seventh and eighth grade classes—whether tracked or not—include a wide range of individual needs and learning styles. Second, if students find appropriate instruction and concern for themselves as individuals, they will attend class and that attendance should eventually yield increased academic achievement. Third, a planned staff development project to expand the repertoire of teaching styles to accord more closely, though not necessarily on a one to one basis, with students currently enrolled should increase opportunities for success in

school. And fourth, a staff development program is more likely to be ongoing if participants are provided opportunities to "buy into" the process through collaborative planning.

This chapter begins with the demographic characteristics of the students in the four selected classes. Next, the results of a five-question pre- and post-Student Attitudinal Survey (see appendices G and H) will be presented, followed by the comparison of students' grades, attitudes, and attendance to class for four quarters (1987-88 school term) in each of the four classes.

## <u>Demographic Characteristics</u>

Demographic characteristics of each class are profiled in Tables 5-8. Table 5 profiles the demographic characteristics of the students in the seventh grade level AP science class. The mean age of the males was 13.2500 years and the mean age of the females was 13.0000 years. There were 23 students enrolled in the class--7 Black males, 15 Black females, and 1 Hispanic female.

Academically, the females maintained a higher yearly average than males--females, 80.4% and males, 74.5%. This class was composed of students who scored 3-5 years above grade level in reading and mathematics on the Iowa Test of Basic Skills in May, 1986. No students failed this class for the year.

TABLE 5

Demographic Characteristics of the Seventh Grade Level AP

Science Class--Teacher #3

	Males	Females	Total
Age (mean)	13.2500	13.0000	13.0760
Sex (n)	7	16	23
Race (n)			
Black	7	15	22
White	0	0	0
Others	0	1	1
Academic Average (1987-88)	74.5%	80.4%	78.6%
Yearly Failures (n) (64-0)	0	0	0

The demographic characteristics of the students in the seventh grade level A social studies class are profiled in Table 6. The mean average of the males was 12.5833 years and the mean age of the females was 13.5000 years. There were 27 students enrolled in the class--5 Black males, 22 Black females, 2 Hispanic females, and 1 White female.

Academically, the females maintained a higher yearly average than the males-females, 88.5% and males, 82.8%.

The students in this class scored 1-2 years above grade level in reading and mathematics on the Iowa Test of Basic

Skills in May, 1986. There were no failures in this class for the year.

TABLE 6

Demographic Characteristics of the Seventh Grade Level A

Social Studies Class--Teacher #1

	Males	Females	Total
Age (mean)	12.5833	13.5000	13.3302
Sex (n)	5	22	27
Race (n)			
Black	5	19	24
White	0	1	1
Others	0	2	2
Academic Average (1987-88)	82.8%	88.5%	87.4%
Yearly Failures (n) (64-0)	0	0	0

Table 7 profiles the demographic characteristics of the students in the seventh grade level B social studies class. The mean age of the males was 13.1666 years and the mean age for the females was 13.3333 years. There were 14 students enrolled in the class--5 Black males and 9 Black females. No other ethnic groups were represented in this class.

The males maintained a higher yearly academic average than the females--males, 71.0% and females, 64.3%. This

class was composed of students whose scores in reading and mathematics ranged from 1 year below grade level to on grade level on the Iowa Test of Basic Skills in May, 1986. Two females failed this class for the year.

TABLE 7

Demographic Characteristics of the Seventh Grade Level B
Social Studies Class--Teacher #2

	Males	Females	Total
Age (mean)	13.1666	13.3333	13.2738
Sex (n)	5	9	14
Race (n)			
Black	5	9	14
White	0	0	0
Others	0	0	0
Academic Average (1987-88)	71.0%	64.3%	66.6%
Yearly Failures (n) (64-0)	0	2	2

The demographic characteristics of the students in the eighth grade level C social studies class are profiled in Table 8. The mean age of the males was 14.5000 years and the mean age of the females was 15.2500 years. There were 15 students enrolled in the class--6 Black males and 9 Black

females. No other ethnic groups were represented in this class.

Academically, the males maintained a higher yearly average than the females—males, 70.0% and females, 58.0%. This class was composed of students who repeated the 8th grade or 8th grade social studies one or more times and those who scored 2 or more years below grade level on the Iowa Test of Basic Skills in May, 1986. Three females failed this class for the year.

TABLE 8

Demographic	Charac	teristic	s of	the	Eighth	Grade	Level	С
:	Social	Studies	Class	5T	eacher :	#4		

	Males	Females	Total
Age (mean)	14.5000	15.2500	14.9500
Sex (n)	6	9	15
Race (n)			
Black	6	9	15
White	0	0	0
Others	0	0	0
Academic Average (1987-88) .	70.0%	58.0%	62.8%
Yearly Failures (n) (64-0)	0	3	3

## Student Attitudinal Survey

A five-question pre- and post-Student Attitudinal Survey was administered to the students in the four classes in April and June, 1988, to assess students' attitudes about their classwork and about their learning. The results of the survey are presented in Tables 9-13.

Table 9 presents the results of students' responses to question #1, "Is the classwork usually interesting?" The April and June, 1988, results indicated that the majority of the students in three classes stated that their classwork was usually interesting. However, ten (43%) students in the seventh grade level AP science class stated "yes" and ten (43%) stated "no" in the June, 1988, survey.

TABLE 9

Question #1.	Is the classwork usually	interesting?
Responses	Respondents	Percentages
	7th Grade Science AP Teacher #3	
	April 13, 1988	
Yes No Sometimes	5 10 8	22% 43% 35%

	T	
	Table 9, cont.	
	June 9, 1988	
Yes No Sometimes	10 10 3	43% 43% 13%
	7th Grade Social Studies A Teacher #1	
	Aprll 13, 1988	
Yes No Sometimes	11 5 3	58% 26% 16%
	June 10, 1988	
Yes No Sometimes	12 1 4	71% 5% 24%
	7th Grade Social Studies B Teacher #2	
	April 18, 1988	
Yes No Sometimes A Little	14 5 1 1	66% 24% 5% 5%
	June 10, 1988	
Yes No Sometimes	15 1 6	68% 5% 27%
	8th Grade Social Studies C Teacher #4	
	Aprll 14, 1988	
Yes	9	75%
No Sometimes	0 3	0% 25%
200000		Table 9, cont.

	Table 9, cont.	
	June 15, 1988	
Yes	8	73%
No	1	9%
Sometimes	2	18%

Table 10 presents the results of students' responses to question #2, "Do you expect to pass the class this quarter?" The April and June, 1988, results indicated that the overwhelming majority of the students in the four classes expected to pass the class during the fourth quarter.

TABLE 10

Question #2.	Do you expect to pass this class	this quarter?
Responses	Respondents	Percentages
	7th Grade Science AP Teacher #3	
	April 13, 1988	
Yes No	20 2	91% 9%
	June 9, 1988	
Yes No Maybe	22 0 1	96% 0% 4%

Table 10, cont.

	7th Grade Social Studies A Teacher #1	
	April 13, 1988	
Yes No Maybe	14 2 2	78% 11% 11%
	June 10, 1988	
Yes No Don't know No response	1 4 1 1	82% 6% 6% 6%
	7th Grade Social Studies B Teacher #2	
	April 18, 1988	
Yes Maybe	20 1	95% 5%
	June 10, 1988	
Yes Maybe Not really	17 4 1	77% 18% 5%
	8th Grade Social Studies C Teacher #4	
	April 14, 1988	
Yes No Maybe	1 1 1 1	84% 8% 8%
	June 15, 1988	
Yes No Maybe Hopefully	8 1 1 1	73% 9% 9% 9%

Table 11 presents the results of students' responses to question #3, "Are you usually given assignments that are of interest only to you"? The results of the April and June, 1988, survey indicated that the majority of the students (64% and 76%, respectively) in the seventh grade level AP science class stated "no." Seventy-four percent (74%) of the students in the seventh grade level A social studies class stated "yes' in April, 1988, and 41% of the students stated "yes" in June, 1988. The largest percentage (82%) of the students in the eighth grade level C social studies class stated "yes" in April, 1988, and 55% of the students stated "sometimes" in June, 1988.

TABLE 11

Question #3 Are you usually given assignments that are of

interest only to you?			
Responses	Respondents	Percentages	
	7th Grade Science AP Teacher #3		
	April 13, 1988		
Yes No Sometimes	4 16 5	16% 64% 20%	

	Table 11, cont.	
	June 9, 1988	
Yes No Sometimes	2 19 4	8% 76% 16%
	7th Grade Social Studies A Teacher #1	
	April 13, 1988	
Yes No Sometimes	14 3 2	74% 16% 10%
	June 10, 1988	
Yes No Sometimes Don't know No response	7 5 3 1 1	41% 29% 18% 6% 6%
	7th Grade Social Studies B Teacher #2	
	April 18, 1988	
Yes No Sometimes	10 9 2	48% 43% 9%
	June 10, 1988	
Yes No Sometimes	8 11 4	35% 48% 17%
	8th Grade Social Studies C Teacher #4	
	April 14, 1988	
Yes No Sometimes	9 1 1	82% 9% 9%

Table 11, cont.

Table 11, cont.

June 15, 1988

Yes	2	18%
No	3	27%
Sometimes	6	55%

Table 12 presents the results of students' responses to question #4, "Do you like coming to this class? If no, why not?" Forty-four percent (44%) of the students in the seventh grade level AP science class stated "sometimes" in the April, 1988, survey. Students' comments ranged from "Because I'm not very good in science and I'm always failing science" to "At times this class can be fun and at other times it can be boring." The June, 1988, survey results indicated that 57% of the students responded "sometimes." Students' comments ranged from "Because it's hard to understand and remember all the work" to "We never see movies—only on the subject."

Seventy-four percent (74%) of the students in the seventh grade level A social studies class stated "yes" in the April, 1988, survey. Students' comments ranged from "Because we always get a lot of homework" to "Because I don't like going to any classes." The June, 1988, results indicated that 88% of the students stated "yes." Students did not make comments.

Eighty-six percent (86%) of the students in the seventh grade level B social studies class stated "yes" in the April, 1988, survey, with no comments. The June, 1988,

survey results indicated that 87% of the students stated "yes," with two comments--"Because it's too hard" and "It can be a little boring."

Eighty-four percent (84%) of the students in the eighth grade level C social studies class stated "yes" in the April, 1988, survey, with one comment--"Because the teacher gets on my nerves." In June, 1988, eighty-two percent (82%) of the students stated "yes," with one comment--"I don't like school."

TABLE 12

Question #4. Do you like coming to this class?			
Responses Respondents Percent			
7th Grade Science AP Teacher #3			
	April 13, 1988		
Yes No Sometimes Not really	8 4 10 1	35% 17% 44% 4%	

# If no, why not?

- 1. Because I'm not very good in science and I'm always failing science.
- 2. Because it has no feeling or meaning so I would call it boring.
- 3. Sometimes it's fun and sometimes it's boring.

Continued, next page.

#### Table 12, cont.

- 4. Because It's so tiring that time of the day when It's almost time to go home.
- 5. It is sometimes a little boring.
- Because it's a boring class, and I don't really care for science.
- 7. Because it's very boring, and I don't like her.
- 8. Sometimes I don't like coming because I know we are getting a lot of work. I don't like science, but sometimes it's interesting.
- 9. At times this class is boring, but at other times it's interesting.
- 10. I don't have a great interest in science.
- 11. Sometimes it's interesting and sometimes it's not.
- 12. At times the class can be fun and at other times it can be boring.

#### June 9, 1988

Yes	13	57%
No	4	17%
Sometimes	б	26%

#### If no, why not?

- 1. Because it's hard to understand and remember all the work.
- 2. Because I don't understand it.
- 3. Sometimes I do and sometimes I don't.
- 4. Because all we do is read.
- 5. I don't like science.
- 6. It's boring. All you do is read, write and take notes.
- 7. Sometimes I am tired.
- 8. We never see movies -- only on the subject.

Table 12, cont.

#### 7th Grade Social Studies A Teacher #1

## April 13, 1988

Yes	14	74%
No	3	16%
Sometimes	2	10%
If no. why not?		

- 1. Because we always get a lot of homework.
- 2. Because some things are not interesting to me.
- 3. Because I don't like going to any class.

## June 10, 1988

Yes	15	88%
Sometimes	1	6%
No response	1	6%

If no, why not?

No responses.

#### 7th Grade Social Studies B Teacher #2

## April 18, 1988

Yes	19	86%
No	1	5%
Sometimes	2	9%

If no, why not?

No responses.

#### June 10, 1988

Yes	20	87%
	1	4%
No Secret Land	2	9%
Sometimes	2	

Table 12, cont.

## If no, why not?

- 1. Because it's too hard.
- 2. It can be a little boring.

#### 8th Grade Social Studies Teacher #4

April 14, 1988

Yes	10	84%
No	1	8%
Sometimes	1	8%

If no, why not?

1. Because the teacher gets on my nerves.

June 15, 1988

Yes	9	82%
No	· <b>1</b>	9%
Sometimes	1	9%

If no, why not?

1. I don't like school.

Table 13 presents the results of students' responses to the April and June, 1988, survey question #5, "What could your teacher do differently to help you improve your learning in this class?" and "What did your teacher do differently to help you improve your learning in this class?" Students' responses are reported in categories based on the similarity of the contents of the statements.

The April, 1988, survey produced eight response categories in the seventh grade level AP science class. Forty-two percent (42%) of the students stated, "She could

give us more projects and experiments." Seven response categories were produced in the June, 1988, survey.

Thirty-six percent (36%) of the students stated, "She did everything basically the same."

Nine response categories were produced by the students in the seventh grade level A social studies class in the April, 1988, survey. Thirty-five percent (35%) of the students stated, "She should let us discuss our work more openly among ourselves." The June, 1988, survey produced eight response categories. Eighteen percent (18%) of the students stated, "She gave us the work more than one time."

Students' responses in the seventh grade level B social studies class produced nine response categories in the April, 1988, survey. Thirty-two (32%) of the students stated, "Nothing." The June, 1988, survey produced seven response categories. Twenty-nine percent (29%) of the students stated, "She lets us do our work alone."

Students in the eighth grade level C social studies class produced six response categories in the April, 1988, survey. Fifty percent (50%) of the students stated, "Nothing." The June, 1988, results produced seven response categories. Forty-two percent (42%) of the students stated, "She explained the work more."

TABLE 13

Question #5. What could (did) your teacher do differently to help you improve your learning in this class?

Res	ponses	Respondents	Percentages
	7th Grade Sci Teacher		
	April 13,	1988	
1.	She could give us more projects and experiments.	12	42%
2.	She could explain the work more vividly.	5	18%
3.	Everything is okay.	4	1 4%
4.	She could take us on a field trip to a science fair or to a museum.		7%
5.	I don't know.	2	7%
6.	She needs a sense of humor.	1	4%
7.	Let us sit next to whom we want to.	1	4%
8.	She can't do anything. I don't study.	1	4%
	June 8,	1988	
1.	She did everything basicall the same.	y 8	36%
2.	She explained everything thoroughly.	5	23%
з.	She gave us more homework.	2	9%

# Table 13, cont.

	14516 15, 66	one.	
4.	I don't really know.	2	9%
5.	She gave us more experiments and projects.	2	9%
6.	She gave us more duplicated materials to review for our exams.	2	9%
7.	She showed more movies.	1	5%
	7th Grade Social Teacher #		
	April 13, 1	988	
1.	She should let us discuss our work more openly among ourselves.	7	35%
2.	She could give us more interesting work.	3	15%
з.	There's nothing she can reall do to improve the class.	) y З	15%
4.	She could have a sense of humor.	1	5%
5.	She can ask us what we want to learn.	1	5%
6.	There should be an extra class for slow people.	1	5%
7.	We shouldn't have to come to school everyday.	1	5%
8.	She could be meaner.	1	5%
9.	I don't know.	1	5%
	June 10,	1988	
1.	She gave us the work more than one time.	3	18%
			Table 13, cont.

Table 13, cont.

# Table 13, cont.

		33	
2.	She explained the lessons to me so I could understand better.	3	18%
з.	No response.	3	18%
4.	She was nicer and concerned about our education.	2	12%
5.	She gave us more work.	2	12%
6.	Yes.	2	12%
7.	Everything.	1	6%
8.	Nothing.	1	6%
	7th Grade Social Teacher April 18,	#2	
1.	Nothing.	7	32%
2.	Give more work.	3	1 4%
з.	No response.	3	14%
4.	Give us hard assignments and teach us more.	3	14%
5.	Make the assignments more interesting by showing films on the subjects.	2	9%
6.	Have more discussions.	1	5%
7.	Less movies.	1	5%
8.	More fun projects.	1	5%
9.	She could give us more time to study for tests.	1	5%
	June 10	, 1988	
1.	She let us do our work alone.	7	29%

# Table 13, cont.

2.	Nothing.	6	25%
з.	She taught us more.	6	25%
4.	She talked more.	2	8%
5.	She let us do more projects and other materials that were interesting.	1	4%
6.	She didn't let us go to the bathroom as often as before.	1	4%
	8th Grade Social Stu Teacher #4	dies C	
	April 14, 1988	3	
1.	Nothing	6	50%
2.	Explain the work more.	2	17%
3.	My teacher does everything interesting in her class.	1	8%
4.	I don't know.	1	8%
5.	She could give us more current events.	1	8%
6.	She helps me all the time.	1	8%
	June 15, 198	8	
1.	She explained the work more.	5	42%
2.	No response.	2	17%
з.	She was nice to us.	1	8%
4.	She made the discussions interesting.	1	8%
5.	She laughed with us.	1	8%
6.	I don't know.	1	8%
7.	No response.	1	8%

# Students' Grades. Attendance, and Attitudes

Tables 14-17 present a comparison of students' grades, attendance, and attitudes for four quarters during the 1987-88 school term in the seventh grade level AP science class, the seventh grade level A social studies class, the seventh grade level B social studies class, and the eighth grade level C social studies class.

As indicated by Table 14, the mean grade for the students in the seventh grade level AP science class for the first quarter was 80.27%. The second quarter showed a 3.46% decrease to 76.81%. The third quarter, however, indicated a gain of 3.9% to 80.77%. The fourth quarter reflected a significant decrease of 4.47% to 76.30%.

Attitudinally, the students displayed a mean score of 85.68% for the first quarter, whereas there was an attitudinal mean of 83.57% during the second quarter, reflecting a decrease of 2.11%. The third quarter represented an increase of 2.34% to 85.91%. The fourth quarter showed a slight decrease of 0.04% to 85.87%.

The mean attendance for the class was 93.32% during the first quarter. The attendance rate showed a significant reduction of 6.89% during the second quarter to 86.43%.

There was a significant increase of 8.48% during the third quarter to 94.91%. A slight decrease of 0.65% occurred during the fourth quarter, reducing the mean attendance rate to 94.26%

TABLE 14

Comparison of Students' Grades, Attitudes, and Attendance for the First, Second, Third, and Fourth Quarters during the 1987-88 School Term in the Seventh Grade Level AP Science Class--Teacher #3

	HTTH	<u>5</u> 88	885538	& & 2 2 5 &	55522585 55 5552585 55	8 8 8 <del>1</del> 5 1
QTR. 4	ATT	<b>585</b>	25 25 25 25 25 25 25 25 25 25 25 25 25 2	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	8 2 8 8 5 8 5 8 5 8 5 8 5 8 5 8 5 8 5 8
	8	889	28882 2888	882 88	88586888 K	88 88
	ATTN	00 88 88	2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	100 100 100 100	88 8 5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	25.5 2.59 2.59 2.50
QTR. 3	нт	5 8 K	£ £ 8 8 £	89.55 95.55 95.55	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	85 22 22 23 27 7.93
	g	25 56 57 57 57 57		88 S S S	85 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	8 22 27 28 8 8 8 4 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8
	HTTH	000	85589	98 84 100 100	525858578	28.06 28.06
QTR. 2	ATT	75 95	25 28 88 88 88 88 88 88 88 88 88 88 88 88	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	888888888648 8888888888	21 83.57 16.05
	g	K &	8882K	8888	<b>KK88288</b> 5K	55 21 3.56 13.56
	ATTN	88 <u>0</u>	2 8 5 8 S	8 8 2 8 8	888855585	23.32 13.64
QTR. 1	ATT	85	<mark>ጜ</mark> ቘ፝፞፞፞፞፞ጜ	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	888 888 888 888 888 888 888 888 888 88	22 22 85.68 12.46
	GR	80	8888	8 8868	888888888888888888888888888888888888888	22 80.27 10.45
	STUDENT	85	S	유 <u>교</u> 영유역	566688868686868686868686868686868686868	GY G

Table 15 indicates that students in the seventh grade level A social studies class displayed an academic mean of 84.96% during the first quarter. There was a slight increase of 1.26% during the second quarter, raising the mean to 86.22%. Likewise, there was an increase of 2.39% during the third quarter, increasing the mean to 88.61%. The fourth quarter showed a slight increase of 0.13%.

The class earned a mean attitudinal score of 91.67% during the first quarter. The second quarter reflected a slight increase of 0.29%, raising the mean to 91.96%. The third also represented a slight increase in the students' attitude of 0.43%, increasing the mean to 92.39%. There was a slight decrease of 0.35% during the fourth quarter, reducing the mean to 92.04%

The mean attendance rate for the students during the first quarter was 95.42%, whereas the second quarter represented a mean attendance rate of 93.87%, reflecting a decrease of 1.55%. The third quarter indicated a mean attendance rate of 93.26%, representing a decrease of 0.61%. An increase of 2.59% occurred during the fourth quarter, raising the mean attendance rate to 95.85%

TABLE 15

Comparison of Students' Grades, Attitudes, and Attendance for the First, Second, Third, and Fourth Quarters during the 1987-88 School Term in the Seventh Grade Level A Social Studies Class--Teacher #1

	ATTA	85558865588865888858888888888888888888	27 95.85 6.33
QTR. 4	ATT	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	27 92.04 6.56
	GR GR	96 97 97 97 98 98 98 98 98 98 98 98 98 98 98 98 98	27 88.74 8.37
	ATTN	98 98 98 98 98 98 98 98 98 98 98 98 98 9	23 93.26 7.77
отв. з	HTT	95 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	23 92.39 6.74
	GB	8-6-25	23 88.61 8.37
	ATTN	00 00 00 00 00 00 00 00 00 00 00 00 00	23 93.87 8.18
QTB. 2	RTT	0 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	23 91.96 6.87
	GR	\$60.00 \$7.00	23 86.22 8.19
	HTTH	882885558822558888555888	24 95.42 3.75
QTR. 1	ATT	88.88.88       88.88	24 91.67 6.24
	GR	85 92 92 93 93 93 93 93 93 93 93 93 93 93 93 93	24 84.96 8.02
	STUDENT	88 88 88 88 88 88 88 88 88 88 88 88 88	SHB N MEAN STD

Table 16 reveals that students in the seventh grade level B social studies class maintained a mean academic grade of 72.14% for the first quarter. The second quarter revealed a decrease of 3.21%, lowering the mean to 68.93%. However, during the third quarter, a significant increase of 5.50% occurred, raising the mean to 74.43%. On the contrary, the fourth quarter reported a decrease of 4.07%, lowering the mean grade to 70.36%.

The students' attitudes indicated a mean score of 81.07% during the first quarter. There was an overwhelming decrease in students' attitudes during the second quasrter, 8.57%. Contrarily, there was a significant increase of 7.41% increase in students' attitudes during the third quarter, raising the mean attitudinal rate to 79.64%. The fourth quarter represented a decrease of 3.93% in students' attitudes, lowering the mean to 75.71%.

The students displayed a mean attendance rate of 87.57% during the first quarter. The attendance rate during the second quarter dropped to 84.29%, representing a decrease of 3.28%. However, during the third quarter, there was a slight increase of 1.21%. There was a further increase during the fourth quarter, representing an increase of 0.86%.

TABLE 16

Comparison of Students' Grades, Attitudes, and Attendance for the First, Second, Third, and Fourth Quarters during the 1987-88 School Term in the Seventh Grade Level B Social Studies Class--Teacher #2

	ATTN	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	11.65
QTR. 4	BTT.	ក្ខខ្មស្តិស្ត្ត	75.75 5.75
	es S	88 0 0 8 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	21.17
	ATTN	88 68 68 68 68 68 68 68 68 68 68 68 68 6	85.50 15.99
отв. з	ATT	28888877 2888888 27 28 28 28 28 28 28 28 28 28 28 28 28 28	79.64
	GB	88 85 85 85 85 85 85 85 85 85 85 85 85 8	74.43 11.98
	HTTH	60 60 60 60 60 60 60 60 60 60 60 60 60 6	22.28 4.29 4.04
QTR. 2	ATT	2 8 8 8 4 7 8 8 8 7 7 8 8 8 7 7 8 8 9 9 9 9 9 9 9	14 72.50 17.50
	GR	08 7 4 8 8 8 8 7 5 9 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	14 68.93 12.84
	ATTN	68 0 5 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	14 87.57 24.58
QTR. 1	ATT	\$5.50 \$5.50	14 81.07 13.39
	GR	65 65 65 65 65 65 65 65 65 65 65 65 65 6	72. 14 12. 49
	STILDENT	<b>운용</b> 등 등 등 등 등 등 등 등 등 등 등 등 등	N MEAN STD

Table 17 indicates that the students in the eighth grade level C social studies class earned an academic average of 66.00% during the first quarter. The second quarter revealed a decrease of 4.46%, lowering the mean to 61.54%. However, during the third quarter, the mean increased by 3.46%, increasing the mean to 65.00%. Likewise, the fourth quarter revealed a slight increase of 0.11%, raising the mean to 65.11%

The mean attitudinal rating for the class was 77.67% during the first quarter. The second quarter represented an overwhelming decrease of 7.67%, lowering the attitudinal mean to 77.00%. The third quarter also reflected an increase students' attitudes, 4.00%, raising the mean to 74.00%. A decrease of 1.78% was reported in students' attitudes during the fourth quarter.

The students' attendance rate during the first quarter was 71.60%. There was a tremendous decrease of 8.45% during the second quarter, lowering the mean attendance rate to 63.15%. Contrarily, during the third quarter, there was a significant increase of 12.55% in the attendance rate. The fourth quarter reflected a slight decrease of 0.03%, lowering the mean to 75.67%

TABLE 17

Comparison of Students' Grades, Attitudes, and Attendance for the First, Second, Third, and Fourth Quarters during the 1987-88 School Term in the Seventh Grade Level C Social Studies Class--Teacher #4

	ATTA	8	87	32	8,4	82	880	15.83 2.9.8
QTR. 4	ATT	80	65	9	8 K	8 65 55	15 P.	22.22 15.11
	GB GB	08	61	40	22	802	55	65.11
	HTTN	8	84	24	) 6 1 0 0	8 7	8 4 8 4	6 8 8 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5
атв. з	НТ	25	65	10	8 8 K	9.5 65 65	K 8	10 74.00 11.36
	GR	80	65	40	8 8 8 8	884	75	10 65.00 14.66
	ATTN	- 85 -	91	67 58	6 6 6	98 4	93 93	13 63. 15 28. 59
QTR. 2	HTT	<b>\$</b> K K	55	65 65	95 8	6 K B	15 R	13 70.00 77.11
	8	<b>5 % 5</b>	8	88	82	8 8 K	65	61.54
	ATTN	8528	82	92,7	65 89	0 8 C	0 4 8 1 0 4 0	15 71.60 23.72
QTR. 1	ATT	9.97 7.85 7.55	<b>8</b> K	4 6	9 65 55 55	9 8 8 8	28 28 50 50 50 50 50 50 50 50 50 50 50 50 50	15 77.67 18.34
	æ	0.00	K A	) (C) (C)	2 SS C	08	0 8 4 6 0 0 0 N	15 66.00 13.81
	STUDENT	588	388	3 P. (	882	38:	4888	MEAN

Tables 18-21 indicate the changes in the means of students' grades, attitudes, and attendance from the first through the second quarter, from the second through the third quarter, and from the third through the fourth quarter for the four classes during the 1987-88 school term.

Table 18 reveals that no positive changes occurred in the students' grades, attitudes, or attendance in the seventh grade level AP science class from the first through the second quarter. However, the third quarter revealed all positive changes, with the most significant change in attendance, 8.48%. There were no positive changes reported during the fourth quarter. A decrease of 4.4% occurred in the students' grades during the fourth quarter.

TABLE 18

Changes in the Means of Students' Grades, Attitudes, and Attendance in the Seventh Grade Level AP Science Class

	1st - 2nd Quarter	
Grades Attitudes Attendance		-3.46% -2.11% -6.89%
	2nd - 3rd Quarter	
Grades Attitudes Attendance		+3.96% +2.34% +8.48%
	3rd - 4th Quarter	
Grades Attitudes Attendance		-4.47% -0.04% -0.65%

Table 19 reflects a positive change in the grades of the seventh grade level A social studies class from the first through the second quarter; however, negative changes were reported in students' attitudes and attendance.

The third quarter reported gains in two areas, grades and attitudes, with more gain occurring in students' grades.

Attendance showed a slight decrease of 0.61%. During the fourth quarter, there were gains in two areas, grades and attendance, with the larger gain in students' attendance.

Students' attitudes reflected a slight decrease of 0.35%.

TABLE 19

Changes in the Means of Students' Grades, Attitudes, and Attendance in the Seventh Grade Level A Social Studies Class

	1st - 2nd Quarter	
Grades Attitudes Attendance		+1.26% -0.29% -1.55%
	2nd - 3rd Quarter	
Grades Attitudes Attendance		+2.39% +0.43% -0.61%
	3rd - 4th Quarter	
Grades Attitudes Attendance		+0.13% -0.35% +2.59%

As revealed in Table 20, no positive changes were reported in students' grades, attitudes, or attendance in the seventh grade level B social studies class from the first through the second quarter. However, during the third quarter, increases were reported in students' attitudes, and attendance, with the largest increase of 7.14% in students' attitudes. The fourth quarter revealed negative changes in students' grades and attitudes. However, a slight increase of 0.86% was reported in students' attendance.

TABLE 20

Changes in the Means of Students' Grades, Attitudes, and Attendance in the Seventh Grade Level B Social Studies Class

	1st - 2nd Quarter	
Grades Attitudes Attendance		-3.21% -8.57% -3.28%
	2nd - 3rd Quarter	
Grades Attitudes Attendance		+5.50% +7.14% +1.21%
	3rd - 4th Quarter	
Grades Attitudes Attendance		-4.07% -3.93% +0.86%

As indicated in Table 21, there were no positive changes reported in students' grades, attitudes, or attendance in the eighth grade level C social studies class from the first through the second quarter. During the third quarter, positive changes were reported in students' grades, attitudes, and attendance, with attitudes representing the most significant increase of 7.14%. The fourth quarter revealed a slight positive change in one area, students' grades.

TABLE 21

Changes in the Means of Students' Grades, Attitudes, and Attendance in the Eighth Grade Level C Social Studies Class

	1st - 2nd Quarter
Grades	-4.46%
Attitudes	-7.67%
Attendance	-8.45%
	2nd - 3rd Quarter
Grades	+3.46%
Attitudes	+4.00%
Attendance	+12.55%
	3rd - 4th Quarter
Grades	+0.11%
Attitudes	-1.78%
Attendance	-0.03%

Table 22 summarizes the changes in the means for students' grades, attitudes, and attendance in the seventh grade level AP science class, the seventh grade leve A social studies class, the seventh grade level B social studies class, and the eighth grade level C social studies class at Roosevelt Junior Senior High School during the 1987-88 school term.

As revealed by Table 22, there was only one positive gain of 1.26% in students' grades from the first through the second quarter in the seventh grade level A social studies class. All other classes showed decreases in grades, attitudes, and attendance, with the largest decrease of

8.45% in attendance, recorded in the eighth grade level C social studies class. There were no overall positive changes in the means from the first through the second quarter in students' grades, attitudes, or attendance, with attendance reflecting the largest decrease of 5.07%.

From the second through the third quarter, the seventh grade level AP science class, the seventh grade level B social studies class, and the eighth grade level C social studies class showed positive changes in the means for grades, attitudes, and attendance. The seventh grade level A social studies showed positive changes in the means for grades and attitudes; however, there was a 0.61% decrease in the mean for attendance. There were overall positive changes in the means for grades, attitudes, and attendance from the second through the third quarter. The greatest change in the means of 5.71% was reported in attendance.

From the third through the fourth quarter, the seventh grade level AP science class reported negative change in the means for grades, attitudes, and attendance, while the other three classes recorded mixed changes in the means.

The seventh grade level A social studies class showed positive changes in grades and attendance, with a negative change of the mean in attitudes. The seventh grade level B social studies class reported negative changes in grades and attitudes, while reporting a positive change in the mean for attendance. The eighth grade level C social studies class

reflected a positive change in the mean for students' grades, while reflecting negative changes in the means for students' attitudes and attendance. Overall, there were negative changes in the means for students' grades, attitudes, and attendance from the third through the fourth quarter.

TABLE 22

Summary of Changes in the Means of Students' Grades,
Attitudes, and Attendance in the Four Classes during the
1987-88 School Term

	7th Grade AP Science	7th Grade A Social Studies	7th Grade B Social Studies	8th Grade C Social Studies	Mean Changes
		1st - 2nd (	Quarter		
Grades Attitudes Attendance	-3.46% -2.11% -6.89%	+1.26% -0.29% -1.55%	-3.21% -8.57% -3.28%	-4.46% -7.67% -8.45%	-2.71% -4.64% -5.07%
		2nd - 3rd	Quarter		
Grades Attitudes Attendance	+3.96% +2.34% +8.48%	+2.39% +0.43% -0.61%	+5.50% +7.14% +1.21%	+3.46% +4.00% +12.55%	+3.83% +3.48% +5.71%
		3rd - 4th	Quarter		
Grades Attitudes Attendance	-4.47% -0.04% -0.65%	+0.13% -0.35% +2.59%	-4.07% -3.93% +0.86%	+0.11% -1.78% -0.03%	-2.08% -1.53% -0.69%

#### Summary

Chapter 4 presented the results of implementing and assessing a staff development program to address seventh and eighth grade students' diverse learning styles and junior high teachers' varied teaching styles with the intent of improving academic achievement, attitudes, and attendance to class in four selected classes at Roosevelt Junior Senior High School. Based on a review of relevant literature, this study relied on four propositions related to adolescents' learning styles and junior high teachers' teaching styles. First, students in early adolescence include a wide range of needs and learning styles. Second, if students find appropriate instruction and concern for themselves, they will attend class. Third, a planned staff development project to expand the repertoire of teaching styles to accord more closely with students currently enrolled should increase opportunities for success in school. And fourth, a staff development program is more likely to be ongoing if participants are provided opportunities to "buy into" the process through collaborative planning.

This study involved four classes, three seventh grade classes (science AP, social studies A, and social studies B) and one eighth grade class (social studies C). Seventy-nine students participated in the study--56 females and 23 males. Ethnically, there were 52 Black females, 23 Black males, 3 Hispanic females, and 1 White female. The mean age of the

students was 13.6 years--females, 13.8 years and males, 13.4 years.

Academically, the females maintained a higher yearly average in the respective classes than the males. No males failed either of the four classes for the year, but five (12%) females failed for the year.

A five-question pre- and post-Student Attitudinal Survey was administered in April and June, 1988. The results indicated that the majority of the students stated that their classwork was usually interesting; they expected to pass their class; they were usually given assignments of interest to them; they liked coming to their class; and they expected to pass their class for the quarter. Students submitted a wide array of comments and suggestions intended to help teachers provide for students' individual learning needs.

An analysis of students' grades, attitudinal ratings, and attendance to class for the four classes revealed increases in the means during the third quarter of the 1987-88 school term in all three areas, with the largest increase of 12.55 percent reported in attendance by the students in the eighth grade level C social studies class. Mixed results were recorded for the four classes during the fourth quarter in three of the twelve areas for the four classes. Eight decreases in the means and four increases

in the means were reported for the four classes during the fourth quarter.

Chapter 5 will present effects, outgrowths, implications for future studies, recommendations for staff development, and conclusions.

## CHAPTER V

OVERVIEW, EFFECTS, OUTGROWTHS, RECOMMENDATIONS FOR STAFF
DEVELOPMENT, FUTURE IMPLICATIONS, AND CONCLUSIONS

## Overview

This dissertation documented the process of designing, implementing, and evaluating a staff development program which provided Junior high teachers with a repertoire of teaching styles to address the diverse learning styles of their seventh and eighth grade students at Roosevelt Junior Senior High School during the 1987-88 school term with the intent of increasing students' achievement, attitudes, and attendance to class. The study was conceived during the fall of 1984 by the researcher who observed that the failure and retention rates among Junior high students at Roosevelt Junior Senior High School was alarmingly high. The researcher was motivated to investigate academic failures and retentions at the Junior high level and to develop strategies upon discussing the concerns with students, parents, teachers, and administrators.

The researcher collaboratively designed workshops to help junior high teachers acquire a repertoire of teaching styles and to assist teachers in designing lessons and activities to motivate junior high students to increase their academic achievement, to improve their attitude about school, and to increase their attendance to class. Teachers

were involved in a variety of discussions and activities relating to their teaching and learning styles, ranging from discovering their learning and teaching styles to role-playing students and teachers, utilizing information they had acquired from the five teaching styles and learning styles workshops.

After completing the five workshops, the four voluntary teachers implemented varied teaching strategies into their selected classes, a seventh grade science level AP class, a seventh grade social studies level A class, a seventh grade social studies level B class, and an eighth grade social studies level C class, during the fourth quarter, April-June, 1988. The teachers collected data, students' grades, attitudinal ratings, and attendance to class, for their respective class for four quarters during the 1987-88 school term and reported it to the researcher for comparison and analysis.

The comparison and analysis of students' grades, attitudinal ratings, and attendance to class for four quarters revealed increases during the third quarter, with mixed results in the fourth quarter. Additionally, through shared experiences with the participating teachers, there was an increase in enthusiasm among other teachers and administrators who were interested in implementing programs designed to address the diverse learning styles of students at Roosevelt Junior Senior High School.

This study was designed to equip junior high teachers with a repertoire of teaching styles to address their students' diverse learning styles with the intent of increasing students' grades, attendance to class, and their attitudes about school. Although, no conclusive effects on students' grades, attendance, and attitudes occurred, a longer study employing the same methodology, may produce more conclusive results. More importantly, however, the process of staff development in an urban junior high school might be applied elsewhere using different themes or goals but many of the processes.

The following variables could have possibly affected the outcomes of this study: (a) small number of teachers participating in the study—four Black females; (b) homogeneous grouping of students in each of the four classes; (c) five short workshops to help teachers become more aware of teaching/learning styles; (d) additional attention given to the students by the teachers and the researcher; (e) teachers' higher expectations of the students participating in the study; (f) teachers possibly providing the researcher with expected results; (g) students' awareness of their participating in the study; (h) teachers utilizing varied teaching strategies while the workshops were conducted; (i) teachers utilizing varied teaching strategies after the workshops had concluded; and (j) time of the year the study was implemented (third and

fourth quarters--teachers were preparing students for end-of-year standardized tests and local final examinations immediately after the study ended).

Although changes in students' grades, attendance, and attitudes were not conclusive, there were some interesting effects on teachers, students, and the researcher. Through interviews, observations, discussions, and analysis of data, the researcher reports the effects of the study on the teachers, the students, and the researcher.

## Effects on Teachers

In April and June, 1988, Student Attitudinal Surveys were conducted to provide students more opportunities to express concerns regarding their teachers providing more opportunities for their educational needs to be addressed by various means. Based on the researcher's observations and discussions with the four teachers, there were concerns that the teachers and students shared and there were concerns that were not shared. The following effects are reported regarding teachers responding to the students' concerns.

Forty-two percent (42%) of the students in teacher #3's seventh grade science level AP class expressed a desire to have more projects and experiments in the April, 1988, survey. Eighteen percent (18%) also stated, "She could explain the work more vividly."

The post-Student Attitudinal Survey, June, 1988, revealed that only nine percent (9%) of the students stated that she gave them more experiments. However, twenty-three percent (23%) stated that she explained everything thoroughly. Teacher #3 clarified her instructions, but she did not provide more experiments, as a large percentage of her students had indicated in the April, 1988, survey.

As indicated by the survey results, students expressed a need for more hands-on activities, such as projects, experiments and field trips. She provided some hands-on activities, such as dissecting frogs, using the microscope, and planting seeds, but she felt compelled to complete the required "textbook materials" to prepare students for the state tests, as reflected by her explaining the lessons thoroughly.

Teacher #3 introduced some hands-on activities into her classes as the students had requested, but she taught basically the same, as indicated by thirty-six percent (36%) of her students in the June, 1988, survey.

Thirty-five percent (35%) of the students in the seventh grade social studies level A class (teacher #1) expressed a need to discuss their classwork openly among themselves. Teacher #1 responded to the students' concerns by dividing the class into small groups, thus, providing more opportunities for her students to interact and to share ideas among themselves. Also, in conjunction with teacher

#2, teacher #1's students constructed projects which were displayed at a Social Studies Fair at Roosevelt Junior Senior High School. She is a student-centered teacher who believes that students' suggestions should be incorporated into the learning process.

Based on teacher #1's beliefs that all students can learn if "nurtured" properly and teachers get more from students If they are "kind and gentle," the following concerns were in conflict with her beliefs: "There should be an extra class for slow people," and "She could be meaner." Conversely, teacher #1 praised her students often.

Fourteen percent (14%) of the students in the seventh grade social studies class expressed the following concerns which were in accord with teacher #2's educational goals:

"Make the assignments more interesting by showing more films on the subject" and "More fun projects." Teacher #2's students constructed projects which were displayed at a Social Studies Fair. As indicated by the June, 1988, survey results, teacher #2 incorporated all the students' concerns into her teaching methodology.

Seventeen percent (17%) of the students in the eighth grade social studies level C class expressed a need for teacher #4 to.explain the work more. Teacher #4 stated that she does not usually explain the lessons from the textbook in the "traditional way." She uses demonstrations, projects, news clippings, field trips, movies, records, and

videos to convey the essence of the written text to her students. As a result of the students' request for more explanations, she began to use the textbook more in class and explained the text more.

Teacher #4 also stated that she viewed herself as the students' friend rather than as their teacher. Students expressed their awareness of her humanistic approaches by the following comments: "She helps me all the time," She was nice to us," She laughed with us," and "She made the discussions interesting."

There were concerns that the teachers and students shared and there were concerns that were not shared, as indicated by the results of the April and June, 1988, Student Attitudinal Surveys. The results of the post-Student Attitudinal Survey indicated that most concerns expressed by the students were addressed by the teachers, as indicated by their teaching techniques, their lessons, their activities, and their attitudes.

To further determine the effects of the study on the four participating teachers, in September 1989, the researcher asked the teachers to respond to the following open-ended question: "What effects did the workshops on learning styles/teaching styles and your participating in the study have on you?"

Participant #1, who is still teaching seventh grade social studies at Roosevelt Junior Senior High School, stated:

I think it was just wonderful. Since I attended the workshops, I have tried so many different things in my classes. My students do individual and group projects. We go on field trips that relate to the lessons. We also watch historical movies that relate to the lessons. I give them choices, rather than making all the selections myself.

One great thing I see is that more teachers are

sharing ideas with each other.

Participant #2, who is currently teaching minth and tenth grade social studies at Roosevelt Junior Senior High School, stated:

Actually, I did not make too many changes in the way I was teaching, because my students have always made projects, read newspapers, worked in groups, watched films related to the lessons, and a variety of other things. But, what I found very helpful was knowing each student's learning style so I could design my lessons and activities broad enough to include most learning styles. From time to time I have to use specific examples to make sure all students are included in class discussions. However, when planning for my classes, my lesson plans must be more detailed and provide more options for my students.

I also listen to my students' suggestions, and I provide them more opportunities to participate in open discussions. I also provide a larger portion of time for them to participate in small groups. This seems to work well because the students helped each other do their work. I observed that students who had not done so well in the past have improved their grades as well

as their oral participation in the groups.

Participant #3, who still teaches seventh grade science at Roosevelt Junior Senior High School found the workshops helpful," Her fears about following the mandated curriculum made her "hesitant" at first:

I think I changed somewhat in the ways I viewed students' opinions. Now, I ask them what kind of activities or lessons that would be interesting to them. They usually have good suggestions. I have also noticed that when they suggest activities or projects, most of them complete them, where as before they would usually give me excuses for not doing the work.

I feel that at this point I am more aware of different teaching styles, but I often go back to the lecture method that I used before. I know this is not good for all my students because of their reactions. For example, they talk, they sleep or they move around in class. I know I must change my teaching methods more often to incorporate the students' learning styles because these students are not the same as when I was in school. They demand more attention. We believed that the teacher knew what was best for the students. But, times have changed.

Teacher #4, who is currently teaching social studies in the newly developed Alternative School at Roosevelt Junior Senior High School, stated:

I was so excited about trying different things with my students in a constructive way. I have always tried innovative approaches with my students, but I did not have clearly defined goals for all of my students. Basically, I tried different techniques that generated enthusiasm among my students. Maybe, that's why I was always given the students other teachers had problems with. With these students, a teacher can't sit at the desk and read to them. They must be actively involved in the classwork.

I suppose the greatest effect that the study had on me was that I could explain to my co-workers that what I was doing in my classes was not crazy. Often when a teacher is not teaching in the traditional way and many different things are going on in the class--some may be noisy and seem unorganized--other teachers and administrators think you are crazy. I felt good that I could explain to my peers that most of my students learned best in an emotionally charged setting.

Observations, individual and group discussions, and interviews with the four teachers revealed that the teachers' attitudes and teaching styles have changed as a

result of attending workshops and participating in the study on learning styles and teaching styles. Attitudinally, teachers changed in the following ways: 1) acknowledged that students know how they learn best; 2) valued students' opinions and ideas; 3) shared teaching ideas with other teachers; 4) interacted more with other teachers in planning for their classes; and 5) seeked assistance more often from other teachers and resource personnel when confronted by difficult situations in their classes.

Teachers changed their teaching styles in the following ways: 1) provided more opportunities for students to participate in open-class discussions; 2) provided students with opportunities to participate in cooperative learning groups; 3) incorporated students' suggestions into the lesson plans; 4) utilized a larger variety of teaching materials; 5) utilized more hands on activities; 6) utilized more electronic teaching aids; and 7) varied teaching techniques, homework assignments, tests, and evaluation procedures.

# Effects on Students

Through observations, interviews, discussions, and evaluations of the students' grades, attendance to class, and attitudinal ratings, the students, the teachers, and the researcher reported the following effects of the study on the participating students. Many students reported that

they felt better about attending class because (a) teachers seemed to care about them more; (b) they made better grades; (c) their teachers praised them more; (d) their teachers provided more options for assignments; (e) students misbehaved less in class; (f) students had more opportunities to work with others in class, and (g) students had more opportunities to express their opinions in class.

The teachers reported that students (a) complained less about being bored in class; (b) worked cooperatively on class assignments; (c) turned in their homework assignments more often; (d) arrived to class on time more often; (e) improved their grades on quizzes and tests; (f) displayed less inappropriate behavior in class; and (g) students offered ideas to make classes more interesting and relevant.

To further determine the effects of the study on the students, the researcher interviewed four students, one from each of the four participating classes, in October, 1989.

A female student from the seventh grade level AP science class-presently a ninth grader--stated:

When I don't understand something in my classes, I am now able to say to my teachers, "I understand best when you use illustrations on the board." I am basically a visual person, so I understand best when I can see what's being discussed in my classes.

A male student from the seventh grade level A social studies class-currently a ninth grade student-revealed:

Since we were in Mrs. X's (name withheld) social studies class in the seventh grade, we have been able to offer suggestions to teachers about different kinds

of projects that would make the classes more fun. The teachers usually listen to what we have to say.

A female student from the seventh grade level B social studies class--presently a ninth grader--stated:

I liked working in the groups when we were in the seventh grade. We had fun doing our work. The class was not so boring because we could talk to our friends when we were doing our work. My grades got better, too.

A male student from the eighth grade level C social studies class--presently a ninth grade repeater--stated:

I liked all the different things we did in Mrs. Y's (name withheld) social studies class in the eighth grade. School did not seem like school, but we still learned a lot. The teacher was nice, but we still did our work, because it was fun. She let us choose things we wanted to do in class.

#### Effects on Researcher

Through reading related literature on learning styles and teaching styles, designing and conducting workshops for Junior high teachers and new teachers in the District on learning styles and teaching styles, implementing a study on varied teaching strategies to address the diverse learning styles of adolescents, interviewing teachers and students, observing teachers' classes and evaluating the study, the researcher reports the following effects: 1) The trusting relationship between the researcher and the teachers has been reinforced because a greater understanding and appreciation of their roles in the school's culture; 2) The researcher became an "on-site" resource person for the

District to assist teachers in designing lessons to provide more diverse learning opportunities for students; 3) The researcher increased his confidence in teachers' abilities to provide learning environments to address students' social, educational, and emotional needs; 4) The researcher became more tolerant of the high level of "structured noise" in classes where teachers utilized emotionally charged activities to address students' diverse needs; 5) The researcher became more aware of research studies designed to address adolescents' diverse social, educational, and emotional needs; 6) The relationship among junior high teachers and the researcher became more collegial rather than the traditional hierarchical administrator-teacher pattern; 7) The researcher became a more cooperative planner by soliciting input from parents, students, teachers, and administrators; and 8) The researcher expanded his repertoire of teaching styles through reading, conducting workshops, implementing the study, and observing teachers' classes.

## Outgrowths

As outgrowths of the staff development program on teaching styles and learning styles on the junior high level at Roosevelt Junior Senior High School during the 1987-88 school term, three new programs, as well as a workshop for first year teachers, were implemented to provide students

more opportunities for their diverse learning styles to be addressed by their teachers. The programs were the following: Pathfinder Program; Modern Languages Tutoring Program; and English as a Second Language (ESL) Peer Assisted Teaching Program.

#### Pathfinder Program

The Pathfinder Program, an example of individualized instruction, was conceived by Dr. Sampson, principal of Roosevelt Junior Senior High School, in October, 1988. She had observed that there was a need for a positive school climate at Roosevelt Junior Senior High School if learning were to take place as indicated by her interview in September, 1988. Dr. Sampson expressed the following:

There were too many students in the halls. They had several factors in common: 1) repeaters, 2) a history of negative encounters with school administration, and 3) poor attendance in class.

We needed to make a positive impact to get these students to return to class. Classes seldom offered any motivation to get these kids back into class. Many times these students were 2-3 years chronologically older than their classmates.

Dr. Sampson indicated that she is hoping that by June, 1989, students who are currently in the 7th grade can go directly to the 9th grade based on their performance in the Pathfinder Program. Some 8th graders may earn credit in math, science, and English to be promoted to the 10th grade. However, the students would still be required to pass state exams.

According to the Pathfinder Program's proposal, the program was designed to provide "at risk" students in grades 7-12, who have repeated one or more grades, opportunities to do computer assisted academic work which could be utilized to help them get in their proper grade. Students would utilize the computer room during non-academic time, study hall, lunch, before school, and after school, to complete their academic work. Students must attend their regularly scheduled classes and complete the assigned work in addition to participating in the Pathfinder Program.

# English as a Second Language (ESL) Peer Assisted Teaching Program

For the past few years, there has been an increasing number of Hispanic students who have enrolled in Roosevelt Junior Senior High School. Many of the students speak very little or no English. Thus, they are placed in an English as a Second Language (ESL) class where they spend the majority of their time interacting only with other Hispanic students, which limits their opportunities to learn English and interact with English speaking students.

Mrs. Cassar, a seventh grade Spanish teacher at
Roosevelt Junior Senior High School, and Dr. DeFina, an
ESL teacher at Roosevelt Junior Senior High School,
conceived the idea for the ESL Peer Assisted Teaching
Program, an example of cooperative learning, by reviewing

the need of the students and planning educational strategies for them. Mrs. Cassar provided the researcher with the following description of the ESL Peer Assisted Teaching Program, which was implemented into her seventh grade Spanish class in December, 1988:

The goal of the group is to create a communication's network between Spanish speaking students and English speaking students. The students serve as a resource to each other. The vocabulary is repeated over and over by each student—first in the known language, then in the language that is being learned. Students will test each other, both oral and written.

A variety of teaching strategies was utilized in the ESL Peer Assisted Tutoring Program. Mrs. Cassar utilized one on one, small groups, student- directed groups, and teacher-directed groups, as well as arranged the classroom furniture to accommodate various learning styles, which she attributed to increased interest, improved behavior, and improved self-esteem among her students.

# Modern Languages Tutoring Program

The Modern Languages Tutoring Program, an example of cooperative learning, designed by Dr. Doll, Project Director, State University at Old Westbury, New York, was implemented in October, 1988, by Mrs. Cassar, Junior high Spanish teacher at Roosevelt Junior Senior High School, and Mrs. Fader, program coordinator and senior high French teacher at Roosevelt Junior Senior high School, in one

seventh grade Spanish class and in one seventh grade French class, respectively.

Dr. Doll stated:

The use of tutors in a collaborative in-classroom effort with instructors is a radical departure from the current practice of having tutors meet privately with students outside of classroom time or of having tutors function as teaching assistants conducting their own classes.

Mrs. Fader's and Mrs. Cassar's seventh grade French and Spanish students participated in the Modern Languages
Tutoring Program from October, 1988, to June, 1989. The teachers indicated that their students had improved their attendance to class and their interest in the classwork.

The Pathfinder Program, the English as a Second

Language (ESL) Peer Assisted Teaching Program, and the

Modern Language Tutoring Program have shown positive

results, according to the teachers and administrators

involved, such as increased students' attendance to class,

positive interactions among students and teachers, increased

interest in classwork, improved behavior, and increased

self-esteem.

# Workshop for First Year Teachers

This workshop was not originally planned as a part of the five staff development workshops for junior high teachers at Roosevelt Junior Senior High School, but Dr. Smith, Director of Supporting Services, Roosevelt Public Schools, Roosevelt, New York, saw a need for first year

teachers to receive training in teaching styles/learning styles. The researcher was asked to conduct a workshop on teaching styles/learning styles for first year teachers in the District as a part of their ongoing staff development activities. The workshop was conducted in the Upper Library at Roosevelt Junior Senior High School on December 1, 1988, from 2:45 p.m. to 3:45 p.m.

Objectives. Specific objectives for this workshop were the following:

- To introduce teaching styles/learning styles by
   Fischer and Fischer
- 2. To introduce a Learning Styles Inventory by Dunn and
- 3. To generate discussions about teaching styles/learning styles
- 4. To introduce a researcher designed Teaching Styles
  Survey
- 5. To introduce teachers to research relative to teaching styles/learning styles
- 6. To introduce teachers to teaching styles/learning styles theorists
- 7. To introduce teachers to various learning styles assessment instruments

The researcher did not design activities to correspond to the objectives because of time constraints. Participants generated discussions on each objective, but objective #4

and #7 generated a disproportionate amount of discussion because the teachers expressed a desire to know their students' learning styles. They believed they could plan their lessons more meaningfully if they were aware of their students' learning styles. The teachers also expressed a desire for more workshops on teaching styles and learning styles because the researcher did not devote enough time on each objective due to the limited amount of time.

Feedback from Workshop Participants. The researcher did not design an assessment form for his presentation of teaching styles/learning styles for first year teachers in the District as a part of their ongoing staff development program. The assessment results were taken from Dr. Smith's evaluative instrument, which was used to assess the effectiveness of first year teachers' workshops.

Of fifteen first year teachers who had attended seven workshops, which were composed of fourteen presentations, four teachers indicated that the teaching styles/learning styles presentation was the most beneficial for them. One teacher stated that teaching styles/learning styles would be beneficial for all professional staff members.

## Recommendations for Staff Development

Staff members who are interested in designing, implementing, and evaluating a low-cost staff development program should build a broad base of support by administering a formal needs assessment to collect information from relevant stakeholders and encouraging feedback about developing plans. Feedback from the needs assessment should assist the staff developer in (a) determining the contents of staff development activities, (b) selecting interested staff members, (c) selecting appropriate topics, (d) determining the format for presentations, (e) determining the time and place, (f) determining the number of staff development activities, and (g) determining the cost.

In order to provide participants with relevant information in staff development activities, the staff developer should (a) utilize current research, (b) utilize district personnel who are experts in their fields, and (c) utilize outside resource personnel who are experts in their fields.

If the staff developer plans to use the results of the staff development activities for publication, he/she should secure permission from participants to use their comments. The staff developer should also secure permission from students' parents or guardians for their participation in the study.

Additionally, the staff developer should share the results of staff development activities with appropriate community and school personnel and solicit feedback to help determine future staff development needs.

Because of the changing demographics of the Roosevelt population, the Roosevelt School District might consider providing as a part of its orientation for new teachers, as well as experienced teachers, an ongoing staff development program to provide teachers with effective teaching strategies and effective managerial strategies which would accord more closely with students' diverse learning styles for improved success in school for teachers and students.

#### Ideas for Conducting Workshops

During the spring of the 1987-88 school term, the researcher conducted five staff development workshops for seventh and eighth grade teachers on teaching styles and learning styles at Roosevelt Junior Senior High School, and in December, 1988, the researcher conducted another workshop on learning styles and teaching styles for first year teachers in the Roosevelt School District. Each workshop was different as to contents, methods of presentation, levels of participation, etc.; yet, the workshops shared many commonalities, defined goals, objectives, and expected outcomes.

The methods of presentation were based on the goals, objectives, and activities of the workshops and the participants' interest and motivation. If the participants did not actively engage in verbal exchanges, as the activity may have required, it became necessary for the researcher to choose optional methods of presentation to accomplish the established goals.

Prospective workshop presenters should be aware that there is not an "only way" to conduct workshops on staff development. For those staff members who are considering conducting staff development workshops, these suggestions are offered:

- 1. Try to include participants from all subject disciplines in the workshops.
- 2. Schedule workshops at times which will cause the least amount of inconvenience for participants.
- 3. Use rooms where there will be few disturbances, for example, not near a gymnasium, a bathroom, or a cafeteria where students enter and exit frequently.
- 4. Use rooms where the Public Address System can be turned off.
- 5. Begin each workshop with a warm-up activity.
- 6. Provide opportunities for participants to become acquainted during the first session.
- 7. Schedule sufficient time for each workshop to complete goals and objectives.

- 8. Adhere to the established time for each workshop.
- 9. Provide as many opportunities as possible for participants to engage in relevant discussions.
- 10. Value participants' opinions.
- 11. Make topics as relevant as possible for participants.
- 12. Use current research to support goals, workshop's objectives, and activities.
- 13. Encourage participants to share class experiences with other participants.
- 14. Solicit a recorder for each session.
- 15. Solicit a "gate keeper" for each session to keep the group on task.
- 16. Seek immediate feedback from participants regarding workshop's presentations and contents.
- 17. Share feedback with participants and other staff members.
- 18. Incorporate participants' suggestions into future workshops.
- 19. End each session with a brief summary and introduce key points of discussion for the next session.

# Verification that Teachers Made Changes in their Classes

When staff development workshops are conducted and teachers are expected to make changes in their classes using acquired teaching methodologies, researchers should verify changes through a variety of relatively unobtrusive

measures. The researcher should not "stand guard" over teachers to verify that they made changes in their classes, because the teachers may feel that their integrity is being questioned. However, the researcher may assist teachers in the implementation process so he/she will have "first hand" knowledge of the procedures the teachers utilized in their classes. The researcher used the following means to assure that the teachers made changes in their classes using the methodologies they acquired in the staff development workshops:

- 1. Reviewed teachers' lesson plans on a regular basis
- 2. Observed participating classes on a regular basis
- 3. Interviewed participating teachers
- 4. Interviewed participating students
- 5. Conducted a Teaching Styles Survey for participating teachers
- 6. Conducted a pre- and post-Student Attitudinal Survey for participating students
- 7. Acquired copies of lessons used in participating classes
- 8. Reviewed copies of quizzes and tests used in participating classes
- 9. Documented activities teachers employed in their classes

The degrees of implementation of the acquired teaching methodologies from the staff development workshops varied

from teacher to teacher based on her teaching preference(s) and her ability to address students' concerns as expressed in the Student Attitudinal Surveys, April and June, 1988.

## Implications for Future Studies

Through observations and discussions with students, teachers, parents, and administrators, there is a need to sensitize teachers to the social, emotional, and educational needs of Black students who live in urban settings. To assist with the process of sensitizing teachers to the needs of poor Black urban students, the following studies could be designed and implemented in Roosevelt Junior Senior High School to specifically address the following concerns:

- Motivating Black, low socioeconomic, urban, junior high students to learn
- 2. Increasing attendance to class of Black, low socioeconomic, urban, junior high school students
- 3. Providing teachers with effective teaching skills to address the diverse learning styles of Black, low socioeconomic, urban, junior high students

There is also an emerging need to design staff development activities to provide teachers with varied instructional approaches to accord more closely with the diverse learning styles of the increasing number of immigrants from El Salvador, Jamaica, and Haiti, who are enrolling in Roosevelt's public schools.

### Conclusions

This study was designed to provide Junior high teachers at Roosevelt Junior High School with a repertoire of teaching styles to address the diverse learning styles of seventh and eighth grade students with the intent of improving academic achievement, attitudes about school, and attendance to class. Five staff development workshops were conducted by the researcher from February 1, 1988, to April 4, 1988, for four voluntary Junior high teachers to acquire varied teaching techniques to foster positive changes in their classes, with the intent of increasing students' grades, attendance, and attitudes.

The researcher concluded from this study that the four classes that participated in this staff development project displayed a range of educational needs and learning styles as indicated by the Student Attitudinal Surveys, April and June, 1988, and the Dunn and Dunn Learning Styles Inventory. Therefore, students in early adolescence are at a stage of high variability and changing identity so that seventh and eighth grade students— whether tracked or not—included a wide range of learning styles.

Instructionally, teachers used different teaching techniques in their classes that they acquired in the staff development workshops and from their fellow teachers, which increased their students' interest in the classwork; increased their sdtudents' self-esteem; and increased their

students' social interactions in class as supported by observations of classes, discussions with students and teachers, and by the Student Attitudinal Surveys, April and June, 1988.

The four voluntary teachers enhanced their repertoire of teaching styles from the five staff development workshops and implemented a variety of teaching strategies to accord more closely with the learning styles of their students as evidenced by teachers introducing varied lessons, activities, and projects into their classes, as well as teachers' increased sensitivity to students' needs, which provided increased opportunities for success in school.

Additionally, the researcher became a more cooperative planner by soliciting input from parents, students, teachers, and administrators. The researcher expanded his repertoire of teaching styles through reading, conducting workshops, implementing the study, and observing teachers' classes. Most specifically, this project demonstrated the potential of school-based improvement projects that respond to identified needs. These efforts will be supported by staff time and commitment if viewed as important by administrators and others. The key to successful staff development is not a series of workshops, but fostering a climate that encourages a variety of school improvement efforts to flourish over time.

**APPENDICES** 

APPENDIX A
STUDENT QUESTIONNAIRE

# ROOSEVELT SCHOOL DISTRICT Roosevelt, New York

# STUDENT QUESTIONNAIRE

	Please respond to the following questions honestly to
help	your teacher plan appropriate lessons for the class:
1.	What can your teacher do to make this class more
	interesting?
2.	What can your teacher do to help you improve your grade
	and attendance in this class?

APPENDIX B
LEARNING STYLES INVENTORY

# ROOSEVELT JR. - SR. HIGH SCHOOL Roosevelt, New York

# LEARNING STYLES INVENTORY

Date:	Grade:		_
1.	I study when it is quiet.	N	Y
2.	I can block out noise when I work.	N	Y
3.	I like studying with lots of light.	N	Y
4.	I study best when the lights are dim.	N	Y
5.	I concentrate best when I feel warm.	N	Y
6.	I concentrate best when I feel cool.	N	Y
7.	When I study, I like to sit on a soft chair or couch.	N.	Y
8.	I feel sleepy if I do not sit on a hard chair when I study.	N	Y
9.	I feel good when I do well in school.	N	Y
10.	When I do well in school, grown-ups in my family are proud of me.	N	Y
11.	Things outside of school are more important to me than my school work.	N	Y
12.	I try to finish all of my work.	N	Y
13.	I hardly ever finish all of my work.	N	Y
14.	I have to be reminded often to do something.	N	Y
15.	I remember to do what I am told.	N	Y
16.	I like to be told exactly what to do.	N	Y
17.	When I really have a lot of studying to do, I like to work alone.	N	Y
18.	When I really have a lot of studying to do, I like to work with a group of friends.	N	Y

19.	I like to study with one or two friends.	N	Y
20.	Sometimes I like to study alone and sometimes with friends.	N	Y
21.	The things I remember best are the things I read.	N	Y
22.	I study better when I eat while I study.	N	Y
23.	I like to eat, drink, or chewbut only after I finish studying.	N	Y
24.	I often nibble something as I study.	N	Y
25.	I think best just before lunch.	N	Y
26.	I remember things best when I study them early in the morning.	N	Y
27.	I remember things best when I study them in the afternoon.	N	Y
28.	I hate to go to sleep at night.	N	Y
29.	I usually start my homework after dinner.	N	Y
30.	When I study, I often get up to do something (like take a drink, get a cookie, etc.) and then go back to work.	N	Y
31.	When I study, I stay with it until all of my work is finished.	N	Y
32.	Noise usually keeps me from concentrating.	N	Y
33.	I can ignore most sound when I study.	N	Y
34.	At home I usually study under a shaded lamp while the rest of the room is dim.	N	Y
35.	When I study, I put on many lights.	N	Y
36.	I usually feel more comfortable in cool weather than I do in warm weather.	N	Y
37.	I usually feel more comfortable in warm weather than I do in cool weather.	N	Y
38.	When it's warm outside, I like to go out.	N	
39	I study best at a table or desk.	N	Y

40.	I like to study on carpeting or rugs.	N	Y
41.	I think my teacher feels good when I do well in school.	N	Y
42.	I like making my teacher proud of me.	N	Y
43.	Nobody really cares if I do well in school.	N	Y
44.	My mother wants me to get good grades.	N	Y
45.	I want to get good grades for me.	N	Y
46.	I love to learn new things.	N	Y
47.	I usually finish my homework.	N	Y
48.	I often forget to do or finish my homework.	N	Y
49.	I have to be reminded often to do something.	N	Y
50.	I keep forgetting to do the things I've been told to do.	N	Y
51.	I always do what I promise to do.	N	Y
52.	I like to be told exactly what to do.	N	Y
53.	I like to be able to do things my own way.	N	Y
54.	I do better if I know my work is going to be checked.	N	Y
55.	When I really have a lot of studying to do, I like to work alone.	N	Y
56.	When I really have a lot of studying to do, I like to work with friends.	N	Y
57.	I like adults nearby when I work alone or with a friend.	N	Y
58.	The things I like doing best in school I do with a group of friends.	N	Y
59.	The thing I like doing best in school I do with grown-ups.	N	Y
60.	If I have to learn something new, I like to learn about it by playing games.	N	Y

61.	The things I remember best are the things I hear.	N	Y
62.	I really like people to talk to me.	N	Y
63.	I really like to watch television.	N	Y
64.	I really like to mold things with my hands.	N	Y
65.	I really like to do experiments.	N	Y
66.	I like to eat or drink, or chew while I study.	N	Y
67.	I do not like to eat or drink, or chew while I study.	N	Y
68.	I study best near lunchtime.	N	Y
69.	I remember things best when I study them early in the morning.	N	Y
70.	I remember things best when I study them before dinner.	N	Y
71.	If I could go to school anytime during the day, I would choose to go in the early morning.	N	Y
72.	When I can, I do my homework in the afternoon.	N	Y
73.	I stay awake for a long time after I get into bed.	N	Y
74.	When I study I stay with it until all my work is finished.	N	Y
75.	It's hard for me to sit in one place for a long time.	N	Y
76.	I can sit in one place for a long time.	N	Y
77.	Noise bothers me when I am studying.	N	Y
78.	My father wants me to get good grades.	N	Y
79.	There are many things I like doing better than going to school.	N	Y
80.	I hardly ever finish all my work.	N	Y
81.	I oftern get tired of doing things and want to start something new.	N	Y

82.	I keep my promises most of the time.	N	Y
83.	I like to be given choices of how I can do things.	N	Y
84.	When I really have a lot of studying to do I like to work with two friends.	N	Y
85.	I like to study by myself.	N	Y
86.	Things I like doing best in school, I do with one friend.	N	Y
87.	The things I like doing best in school, I do with a grown-up.	N	Y
88.	If I have to learn something new, I like to learn about it by having it told to me.	N	Y
89.	If I have to learn something new, I like to learn about it be seeing a filmstrip or a film.	N	Y
90.	The things I remember best are the things I write about.	N	Y
91.	I really like to draw, color, or trace things.	N	Y
92.	I really like to listen to people talk.	N	Y
93.	I often nibble on something as I study.	N	Y
94.	I feel wide awake after 10:00 in the morning.	N	Y
95.	It's hard for me to sit in one place for a long time.	N	Y
96.	I can sit in one place for a long time.	N	Y
97.	I think my teacher wants me to get good grades.	N	Y
98.	I like to do things in my own way.	N	Y
99.	I really like to build things.	N	Y
100.	I can work best for short amounts of time with rest periods.	N	Y

#### APPENDIX C

COMPILATION OF STUDENTS' LEARNING STYLES IN THE SEVENTH GRADE LEVEL AP SCIENCE CLASS, THE SEVENTH GRADE LEVEL A SOCIAL STUDIES CLASS, THE SEVENTH GRADE LEVEL B SOCIAL STUDIES CLASS, AND THE EIGHTH GRADE LEVEL C SOCIAL STUDIES CLASS

Compilation of Students' Learning Styles in the Seventh Grade Level AP Science Class, the Seventh Grade Level A Social Studies Class, the Seventh Grade Level B Social Studies Class, and the Eighth Grade Level C Social Studies Class

	7th Grade AP Science	<del>с.</del> п	7th Grade A Social Studies	Œ " Ŋ	7th Grade B Social Studies	ca vi	8th Grade C Social Studies	O – 8
	N = 22		N = 23	<b>m</b>	N N N N N N N N N N N N N N N N N N N		H Z	10
	QZ ₩	% YES	₽ %	% YES	% NO	&YES	ON €€	¥ ₩
LEARNING CONDITIONS	. 4 . N	55	22	28	53	12	20	80
1 study when it is quiet.	64	36	36	61	20	20	40	60
I can block out noise when I stady.	49	36	5.5	43	43	52	30	20
like studying with lots of light.	50	50	52	<b>4</b> 00	င္မ	20	20	30
study best when the lights with the study best when the lights with the study w	4	59	36	61	23	7.1	20	80
concentrate best when I feel cool.	55	<del>4</del> 5	52	<b>4</b> 00	98	4	90	40
When I study, I like to sit on a soft chair or couch.	36	64	71	88	4	86	0	100
I feel sleepy if I do not sit on a hard chair when I study.	73	72	52	48	71	29	06	5 8
feel good when I do well in school.	O	16	4	96	0	9	2	5
When I do well in school, grown-ups in my family are proud of me.	ø	91	<u>6</u>	87	0	100	20	80
Things outside of school are more important to me than my school work.	64	36	87	<u>e</u>	98	<del>7</del>	06	0

S

SNOTTIONS CALVAGE	ON ₩	₩ YES	ON ₩	% YES	ON R	₩ YES	Q <b>₩</b>	% YES
And the Anti-	23	77	σ	16	~	63	0	06
try to finish dil of my work.	50	4	70	30	37	43	90	40
hardly ever finish all of my work.	<u>.</u> 6	32	61	36	43	52	90	40
have to be reminded often to do sometime.	32	99	36	61	21	5	40	90
I remember to do what I am told. I like to be told exactly what to do.	300	64	30	20	20	20	0	06
When I really have a lot of studying to do, I like to work alone.	32	68	22	78	21	62	30	20
When I really have a lot of studying to do,	ស	4 ت	10	36	64	36	60	40
like to work with a group of Trienus.   like to study with one or two friends.	5 5 5	<del>4</del> 5	30	20	29	71	30	20
Sometimes I like to study alone and sometimes with friends.	27	<b>4</b> 3	26	46	4	98	20	08
The things I remember best are the things I	1-4	59	30	20	4	86	90	40
read. I study better when I eat while I study.	20	20	<b>6</b> 6	61	29	7.1	90	40
1 like to eat, drink, or chewbut only	64	36	55	£43	ر و	21	20	30
after I finish studying.	32	89	22	28	29	71	30	20
often nibble something ds   study.   think best just before lunch.	22	23	20	30	62	21	80	20
I remember things best when I study them early in the morning.	55	45	20	30	<u>م</u>	21	06	10
I remember things best when I study them	32	68	43	25	36	64	50	50
in the afternoon.	55	45	20	30	7.1	29	20	30
I hate to go to sleep at higher.	5	4-	83	17	20	20	20	20
I usually start ing nomination								

LEARNING CONDITIONS	₽ ₩	% YES	₽ ₩	S≡ YES	0N ₩	₩ YES	Q ₩	₩ YES
When I study, I often get up to do something (like take a drink, get a cookie, etc.) and then go back to work.	8	82	7	83 83	<del>4</del>	98	20	08
When I study, I stay with it until all of my more is finished.	82	18	36	61	12	29	20	30
Main the bears me from concentrating.	32	68	32	65	23	71	9	00
Noise usually respendence of study.	64	36	26	74	20	20	30	20
At home I usually study under a shaded lamp	40	36	52	84	7.	29	80	20
while the rest of dig 100m is a second while the rest of many lights.	64	36	52	<b>4</b> 80	64	36	20	20
l usually feel more comfortable in cool weather than I do in warm weather.	4 ال	را تا	65	35	49	36	04	90
usually feet more comfortable in warm	27	73	36	6.1	21	79	30	20
weather than I do in cool weather.	81	82	13	87	21	49	20	80
When it's warm outside, I like to your.	20	20	22	78	20	20	20	30
i study bast at a table or uesh. I like to study on carpeting or rugs.	4	50 0	10	<b>6</b> 8	36	64	30	20
I think my teacher feels good when I do	23	77	56	74	~	893	20	80
well in school.	36	64	30	20	23	7.1	30	20
like making my teacher product mass	20	4 5	48	52	2	21	20	80
Nobody really cares if I do well in school.	σ	0	26	74	0	001	0	100
My mother wants me to get good grades.	ម	92	4	96	0	100	0	100
I want to get good grades for me.	23	77	<b>O</b>	16	4	86	ō	06
I love to learn new things. I usually finish my homework.	23	22	22	78	4	86	30	20

OZO F	0Z	% YES	QN ₩	% YES	ON ₩	% YES	oN ₩	% YES	
LEHRAING CONDITIONS		10 10	55	35	30	20	90	40	
often torget to do of titles my commercial		C	2	48	57	43	80	20	
have to be reminded often to do something.	2	0	3	)					
keep forgetting to do the things I've	Q8	32	20	30	<b>6</b> 4	52	20	30	
almand do mhat I promise to do.	36	64	<u>5</u>	87	20	20	30	20	
diago to be told exactly what to do.	4	59	36	61	£43		30	20	
like to be able to do things my own way.	32	68	26	74	29	7.1	20	80	
l do better if I know my work is going to be checked.	₹ 10	55	17	83	4	86	0	100	
When I really have a lot of studying to do, I like to work alone.	4-	26	26	74	53	71	30	20	
When I really have a lot of studying to do, I like to work with friends.	<del>4</del> ان	55	5.7	<b>6</b> 4 α	52	<b>6</b>	40	09	
l like adults nearby when I work alone or with a friend.	<del>1</del> 0	22	8	52	4	36	90	40	
The things I like doing best in school I do with a group of friends.	32	89	26	45-	4	36	20	20	
The thing I like doing best in school I do with grown-ups.	92	ر د	28	22	52	21	Ō	06	
If I have to learn something new, I like to learn about it by playing games.	45	55	<b>4</b> 8	52	<b>6</b>	57	70	30	
The things I remember best are the things I	36	64	56	44	20	50	20	30	
hear.	27	73	13	87	29	71	50	50	
TREATLY LIKE PROPIE to talk to mit.		22	13	87	21	50	20	80	
really like to match television.		50	43	57	25	43	20	50	
I really like to moid things with my imprison									

	0Z	S⊒\ ₩	Q ₩	% YES	QZ ₩	₩ YES	ON SK	% YES	
LEARNING CONDITIONS 1 really like to do experiments.		16	7	83	4	86	20	80	
i like to eat or drink, or chew while I study.	32	8	30	02	29	7.1	04	09	
I do not like to eat or drink, or chew	26	4-	•	33	20	20	20	50	
while istudy. I study best near lunchtime.	86	4	65	35	7.	29	80	20	
I remember things best when I study them early in the morning.	59	4	52	<b>6</b> 4	7	29	09	04	
I remember things bast when I study them before dinner.	55	4 ت	43	52	20	20	90	40	
If I could go to school anytime during the day, I would choose to go in the early morning.	δ 0	20	65	35	52	φ 6	20	20	
When I can, I do my homework in the afternoon.	27	73	22	28	21	62	30	20	
I stay awake for a long time after I get into bed.	20	20	52	<b>4</b> 00	71	29	20	20	
When I study, I stay with it until all my work is finshed.	64	36	52	<b>4</b> 80	<b>6</b>	57	20	30	
1t's hard for me to sit in one place for	23	7.2	£4	52	98	64	40	<b>9</b> 6	
a larg time.	99	32	61	36	98	4	80	20	
Can be Commented to a studying.	36	64	43	52	21	29	50	20	
Noise bothers me when I am John John Brades.	23	27	35	65	~	66	20	20	
There are many things I like doing better	27	73	39	61	29	7.1	20	50	
than going to school.	4 N	55	57	<b>4</b> ω	52	£ 4	20	30	

I hardly ever finish all my work

LEARNING CONDITIONS	ON ₩	₩ YES	Q ₩	₩ YES	<b>9</b>	% YES	Q ₩	₩ YES	
i often get tired of doing things and mant to start something new.	32	89	1-	83	~	93	30	20	
I keep my promises most of the time.	27	73	<b>6</b> 4	52	5	<b>4</b> ω	30	5	
l like to be given choices of how I can do things.	23	22	2 <b>6</b>	74	<del>7</del>	86	04	09	
When I really have a lot of studying to do. I like to work with two friends.	<del>1</del> 4	55	19	6°	40	36	60	9	
like to study by myself.	27	23	26	46	<del>7</del>	98	6	<b>6</b> 0	
Things I like doing best In school, I do with one friend.	65	4	30	20	49	36	20	30	
The things I like doing best in school, I do with a grown-up.	<b>98</b>	4	83	17	52	<b>4</b> 0	20	30	
If I have to learn something new, I like to learn about it by having it told to me.	4-	53	30	20	<b>4</b> დ	52	90	40	
If I have to learn something new, I like to learn about it by seeing a filmstrip or a film.	<u>æ</u>	82	35	65	21	6-	80	20	
The things I remember best are the things I write about.	80	4-	26	74	<b>6</b> 4	52	40	09	
I really like to draw, color, or trace	32	89	26	74	21	29	40	90	
things.	50	50	39	61	36	40	40	90	
I really like to listen to respect to ften nibble on something as I study.	36	49	13	87	21	62	04	90	
I feel wide awake after 10:00 in the morning.	32	89	4 ω	52	21	62	04	90	
it's hard for me to sit in one place for a long time.	4	26	52	8	8	64	04	90	

LEARNING CONDITIONS	Q ₩	₩ YES	Q ₩	% YES	2 K	S √ES	0 X	% √ES
I can sit in one place for a long time.	49	36	35	<b>6</b> 5	52	21	80	20
i think my teacher wants me to get about arades.	<b>O</b>	16	13	87	0	100	20	80
1 like to do things in my own way.	<del>-</del> 4	86	26	74	59	71	20	20
really like to build things.	4	89	26	4	21	79	40	90
I can work best for short amounts of time with rest periods.	23	27	22	78	43	57	80	20

APPENDIX D
CONSENT FORM

### ROOSEVELT PUBLIC SCHOOLS Roosevelt, New York

#### CONSENT FORM

Dear Colleague:

I am a doctoral student at the University of Massachusetts. Your professional judgment is needed to help formulate a staff development program which addresses teaching styles and learning styles at the junior high level.

Participation in this project will involve:

1) completing a teaching styles survey, 2) participating in workshops, 3) sharing opinions, and 4) completing evaluation forms. Individual evaluation and survey forms will be reviewed and results will be shared with participants. The summarized survey data will be included in my dissertation. Your name will not be used in my dissertation. Statements made by workshop participants may be quoted in the dissertation. Written permission to quote an individual workshop participant will be obtained if necessary.

Participation in this project is voluntary and you may withdraw at any time. Any questions regarding staff development will be welcome.

Thanking you in advance for your support.

Sincerely,

Roy R. Taylor

Please sign below if you intend to be a voluntary participant in this project.

Signature	Date

APPENDIX E
TEACHING STYLES SURVEY

#### TEACHING STYLES SURVEY

SCHOOL:	
SUBJECT TAUGHT:	
DATE:	
Dear Colleague:	

Your professional judgment is needed to survey teaching styles and learning styles at the junior high level. For purposes of this survey, teaching styles and learning styles will be explored.

Participation in this survey is voluntary. Individual surveys will be reviewed, results will be summarized, and shared with survey participants. The summarized data will be used for a study at the University of Massachusetts; therefore, names should not be included to protect confidentiality.

Thank you for your participation.

Roy R. Taylor

Please circle your response.

- 1. I plan my lessons equal concern for students and curricular objectives.
- Always Frequently Sometimes Never
- 2. I provide an atmosphere that allows my students to pursue activities that interest me.
- Always Frequently Sometimes Never

- 3. I plan my class activities with student input.
- Always Frequently Sometimes Never
- 4. I demand that my students perform academically at a specific level.
- Always Frequently Sometimes Never

5. I must finish teaching the textbook in order to meet curricular requirements.

Always Frequently Sometimes Never

6. I prescribe materials for a student to learn based on his/her learning style.

Always Frequently Sometimes Never

7. I become emotionally involved in my teaching.

Always Frequently Sometimes Never

8. I use different techniques to involve all activities.

Always Frequently Sometimes Never

9. My students ask questions for clarification.

Always Frequently Sometimes Never

10. My students complain of boredom.

Always Frequently Sometimes Never

11. My students complete their homework assignments.

Always Frequently Sometimes Never

12. I rephrase questions or statements when my students do not understand.

Always Frequently Sometimes Never

13. I use the same materials with all students.

Always Frequently Sometimes Never

14. Individualization enables me to involve more students in class.

Always Frequently Sometimes Never

	workshops that would provide me with a variety of teaching techniques.
16.	If yes, please list topics about which you would like additional information.
	Comments:
	•

Yes

No

15. I would attend

APPENDIX F
GRADE REPORT

#### ROOSEVELT SCHOOL DISTRICT Roosevelt, New York

#### GRADE REPORT

#### Explanation of Marks

#### SUBJECTS

90 and above -- Excellent

80 - 89 -- Good

70 - 79 -- Fair

65 - 69 -- Poor

64 and below -- Failing

#### ATTITUDE

A -- Excellent

B -- Very Good

C -- Satisfactory

D -- Needs Improvement

F -- Unsatisfactory

# APPENDIX G PRE-STUDENT ATTITUDINAL SURVEY

### ROOSEVELT/UMASS STAFF DEVELOPMENT Roosevelt, New York

### PRE-STUDENT ATTITUDINAL SURVEY April, 1988

Date	Grade:
Dear	Students,
lear	Please answer the following questions so we may make rning more pleasant for you.
1.	Is the classwork usually interesting?
2.	Do you expect to pass this class this quarter?
з.	Are you usually given assignments that are of interest
	only to you?
4.	Do you like coming to this class? If no, why
	not?
5.	What could the teacher do differently that would help
	you improve your learning in this class?

APPENDIX H
POST-STUDENT ATTITUDINAL SURVEY

## ROOSEVELT/UMASS STAFF DEVELOPMENT Roosevelt, New York

### POST-STUDENT ATTITUDINAL SURVEY JUNE, 1988

Date	Grade:
Dear	Students,
lear	Please answer the following questions so we may make ning more pleasant for you.
1.	Is the classwork usually interesting?
2.	Do you expect to pass this class this quarter?
3.	Are you usually given assignments that are of interest
	only to you?
4.	Do you like coming to this class? If no, why
	not?
5.	What did your teacher do differently that helped you
	improve your learning in this class this quarter?

#### REFERENCES

- Adelman, H. S., & Taylor, L. (1983). Motivation for overcoming learning and behavioral problems.

  Journal of Learning Disabilities, 16, 384-392.
- Adler, M. (1982). The Paidela proposal: An educational manifesto. New York: MacMillan.
- Allan, V. L., Feldman, R. S., & Devin-Sheeham, L. (1976). Research on children tutoring children: a critical review. Review of Educational Research, 46, 355-386.
- American adolescent: Facing a "Vortex of new risks." (1989, June 21). Education Week. pp. 22-23.
- Austin, G. R. (1979). Exemplary schools and the search for effectiveness. Educational Leadership, 37, 10-14.
- Barbe, W. B., & Milone, M. N. (1980). Modality. Instructor, 89, 44-47.
- Barbe, W. B., & Milone, M. N. (1981). What we know about modality strengths. Educational Leadership, 38, 378-380.
- Barbe, W. B., & Swassing, R. (1979). <u>Teaching through</u> modality strengths: Concepts and practices. Columbus, OH: Zaner-Bloser.
- Barr, R., & Dreeben, K. (1977). Instruction in classrooms. Review of Research in Education, 5, 89-162.
- Bates, R. J. (1987). Corporate culture, schooling, and educational administration. Educational Administration Quarterly, 23, 79-115.
- Block, J. H., & Anderson, L. W. (1975). Mastery learning in classroom instruction. New York: Macmillan.
- Block, J. H., & Burns, R. B. (1976). Mastery learning. In L. S. Shulman (Ed.), Review of research in education (Vol. 4, pp. 3-49). Itasca, IL: F. E. Peacock.
- Bloom, B. S. (1976). Human characteristics and school learning. New York: McGraw-Hill.
- Boyer, E. L. (1983). High school: A report on secondary education in America. New York: Harper and Row.

- Brandt, R. (1988). On students' needs and team learning: A conversation with William Glasser. Educational Leadership, 45, 38-45.
- Bronfenbrenner, U. (1974). The origins of alienation. Scientific American, 231, 53-61.
- Bronfenbrenner, U. (1986). The four worlds of childhood. Phi Delta Kappan, 67, 430-436.
- Brookover, W., Beady, D., Flood, P., Schweitzer, J., & Wisenbaker, J. (1979). School social systems and student achievement: Schools can make a difference. New York: Praeger.
- Brookover, W., & Lezotte, L. W. (1979). Changes in school characteristics: Coincident with change in student achievement. East Lansing: Institute for Research on Teaching, Michigan State University. (ERIC Document Reproduction Service No. ED 181 005)
- Burns, R. (1982). <u>Self-concept development and education</u>. London: Holt, Rinehart and Winston.
- Calabrese, R. L. (1987). Adolescence: A growth period conducte to allenation. <u>Adolescence</u>, <u>22</u>, 929-938.
- Canfield, A. A., & Lafferty, J. C. (1970). <u>Learning styles</u> <u>inventory</u>. Detroit, MI: Humanics Media.
- Carbo, M., & Hodges, H. (1988). Learning styles strategies can help students at risk. <u>Teaching Exceptional</u> <u>Children</u>, <u>20</u>, 55-58.
- Carlisle, K. E. (1985). Learning how to learn. Training and Development Journal, 39, 75-80.
- Carroll, J. B. (1963). A model for school learning. Teachers College Record, 64, 723-733.
- Cohen, L., & Manion, L. (1985). The methods in education.
  Dover, NH: Crom Heim.
- Coleman, J. C. (1978). Current contradiction in adolescent theory. <u>Journal of Youth and Adolescence</u>, 7, 1-11.
- Combs, A. W. (1988). New assumptions for educational reform. Educational Leadership, 45, 38-40.
- Combs, A. W., Avila, D. L., & Purkey, W. (1971). Helping relationships. basic concepts for the helping professors. Boston: Allyn and Bacon.

- Corbett, H. D., Firestone, W. A., & Rossman, G. B. (1987).
  Resistance to planned change and the sacred in school cultures. Educational Administration Quarterly, 23, 36-59.
- Crenshaw, B. A. (1986). I'm teaching middle school because I like it. <u>Independent School</u>, 45, 31-33.
- Cureton, G. O. (1978). Using a black learning style. The Reading Teacher, 31, 751-756.
- DeAnde, N. (1975). <u>Helping children develop healthy</u>
  <u>self-concepts (Pre-school--third grade</u>). San Francisco,
  CA: Far West Laboratory for Educational Research and
  development. (ERIC Document Reproduction Service No.
  ED 129 455)
- Doyle, W., & Ponder, R. W. (1977). Environment influences on professional behavior: a case of elementary teaching. Journal of Educational Research, 70, 55.
- Duckett, W., Parke, D., Clark, D., McCarty, M., Lotto, L., Gregory, L., Herling, J., & Burlson, D. (1980). Why do some schools succeed? The Phi Delta Kappa study of exceptional elementary schools. Bloomington, IN: Phi Delta Kappa.
- Dunn, R. (1982). Teaching students through their individual learning styles: A research report. In <u>Student learning</u> styles and brain behavior (pp. 142-151). Reston, VA:
  National Association of Secondary School Principals.
- Dunn, R. (1989). Learning styles: Key to improving schools and student achievement. NASSP Curriculum Report, 18, 1-4.
- Dunn, R., & Carbo, M. (1981). Break through: How to improve early reading instruction. Phi Delta Kappan, 62, 675.
- Dunn, R., DeBello, T., Brennan, P., Krimsky, J., & Murrain, P. (1981). Learning style researchers define differences differently. Educational Leadership, 38, 372-387.
- Dunn, R., & Dunn, K. (1975). Finding the best fit:

  Learning styles, teaching styles. NASSP Bullletin, 59,

  37-49.
- Dunn, R., & Dunn, K. (1978). <u>Teaching students through</u>
  their individual learning styles: A practical approach.
  Reston, VA: Reston Publishing.

- Edmonds, R. (1978, August). An address to the teacher corps. Speech delivered in Washington, D. C.
- Elias, M. J., Gara, M., & Ubriaco, M. (1985). Sources of stress and support in children's transition to middle school: A empirical analysis. <u>Journal of Clinical Child Psychology</u>, 14, 112-118.
- Elmore, R. F. (1987). Reform and the culture of authority in schools. Educational Administration Quarterly, 23, 60-78.
- Erikson, E. H. (1963). Childhood and society (2nd ed.). New York: W. W. Norton.
- Evertson, C. M., & Emmer, E. T. (1981). Effective management at the beginning of the school year in junior high classes. <u>Journal of Educational Psychology</u>, 74, 485-498.
- Feiman-Nemser, S., & Floden, R. (1986). The cultures of teaching. In Wittrock (Ed.), <u>Handbook of research on teaching</u> (3rd ed.; pp. 505-526). New York: Macmillan.
- Fenstermacher, G. M., & Berlinger, D. C. (1983). A conceptual framework for the analysis of staff development. Santa Monica, CA: Rand Corporation.
- Fischer, B. B., & Fischer, L. (1979). Styles in teaching and learning. <u>Educational Leadership</u>, 36, 245-254.
- Fogelman, K. (1978). School attendance attainment and behavior. British Journal of Educational Psychology, 48, 148-158.
- Ford, D. (1986). Improving urban students' perception of school. Clearing House, 60, 52.
- Garten, T., & Valentine, J. (1989). Strategies for faculty involvement in effective schools. NASSP Bulletin, 73, 1-6.
- Gephart, W., Strother, D., & Duchett, W. (1980). On mixing and matching of teaching and learning styles. Practical application of research, Bloomington, IN: Phi Delta Kappa.

- Giroux, H., & Purpel, D. E. (Eds.). (1983). The hidden curriculum and moral education: Deception or dishonesty? Berkley, CA: McCutchan.
- Goodlad, J. I. (1984). A place called school: Prospects for the future. New York: McGraw Hill.
- Grant, G. (1985, Fall). The character of education and the education of character. <u>Kettering Review</u>, 51-52.
- Gregorc, A. F. (1979). Learning/teaching styles: Their nature and effects. In J. W. Keefe (Ed.), Student learning styles: diagnosing and prescribing programs (pp. 19-26). Reston, VA: National Association of Secondary School Principals.
- Gross, N., Giaquinta, J. B., & Bernstein, M. (1971).

  Implementing organizational change. New York: Basic Books.
- Guild, P. B., & Garger, S. (1985). <u>Marching to different</u> <u>drummers</u>. Alexandra, VA: Association for Supervision and Curriculum Development.
- Gullotta, T. (1983). Early adolescence, alienation and education. Theory Into Practice, 22, 151-154.
- Hale-Benson, J. E. (1988). <u>Black children: Their roots.</u> <u>culture. and learning styles</u> (3rd ed.). Baltimore: MD: Johns Hopkins University Press.
- Hale-Benson, J. E. (1989, March 2). Designing education for Black students. Newsday. p. 81.
- Hall, G., Rutherford, W. L., Hord, S. M., & Huling, L. L. (1984). Effects of three principal styles on school improvement. <u>Educational Leadership</u>, <u>41</u>, 22-29.
- Haviland, J. M., & Scarborough, H. S. (1981). Adolescent development in contemporary society. New York: D. Van Nostrand Company.
- Heikkinen, M., Pettigrew, F., & Zakrajsek, D. (1985). Learning styles vs. teaching styles--studying the relationship. NASSP Bulletin, 69, 80-85.
- Hiebert, E. H. (1980). Peers as reading teachers. Language Arts, 57, 877-881.
- Huberman, M., & Miles, M. B. (1984). <u>Innovation up close:</u>
  How school improvement works. New York: Praeger.

- Huling-Austin, L., & O'Bryan-Garland, S. (1987). Increasing faculty involvement with the public schools. Action in Teacher Education, 9, 41-48.
- Hunkins, F. P. (1987). Sharing our instructional secrets. Educational Leadership, 45, 65-67.
- Hunt, D. E. (1979). Learning style and student needs: An introduction to conceptual level. In J. W. Keefe (Ed.), Student learning styles: Diagnosing and prescribing problems (pp. 27-38). Reston, VA: National Association of Secondary School Principals.
- Hurtford, A., & Hart, D. (1979). Social interaction in a language unit. <u>Special Education</u>, <u>6</u>, 8-10.
- Ianni, F. A. J. (1989). The search for structure: A report on American youth today. New York: The Free Press.
- Ishler, R. E., & Leslie, E. C. (1987). Bridging the gap between a public school system and a university. Phi Delta Kappan, 68, 615-616.
- Jones, B. L., & Maloy, R. W. (1988). <u>Partnerships for improving schools</u>. Westport, CT: Greenwood Press.
- Joyce, B., Showers, B., & Rolhelser-Bennett, C. (1987).
  Staff development and student learning: A synthesis of research on models of teaching. Educational Leadership, 45, 11-23.
- Keefe, J. W. (Ed.). (1979). <u>Student learning styles:</u>
  <u>Diagnosing and prescribing programs</u>. Reston, VA:
  National Association of Secondary School Principals.
- Kolb, D. A. (1981). Disciplinary inquiry norms and student learning styles: Diverse pathways for growth. In
   A. Chickering (Ed.), The modern American college. San Francisco, CA: Jossey-Bass.
- Korpinen, E. (1987). Effect of evaluation on the student's self-concept. <u>Kasvatus</u>. The Finnish Journal of <u>Education</u>, 18, 50-61.
- Lambert, L. G. (1988). Building school culture: An open letter to principals. NASSP Bulletin, 72, 54-62.
- Lieberman, A., & Miller, L. (1984). <u>Teachers. their world</u> and their work: <u>Implications for school improvement</u>. Alexandria, VA: Association for Supervision and Curriculum Development.

- Lightfoot, S. L. (1983). The good high school: Portraits of Character and culture. New York: Basic Books.
- Louis, M. R. (1985). Sourcing workplace cultures: Why, when, and how. In R. H. Kilmann, M. J. Saxton, R. Serpa, & Associates (Eds.). Gaining control of the corporate culture (pp. 126-136). San Francisco: Jossey-Bass.
- Lucker, G. W., Rosenfield, D., Sikes, J., & Aronson, E. (1976). Performance in the interdependent classrsoom: A field study. American Educational Research Journal, 13, 115-123.
- Mackenzie, D., & Urich, T. (1984). Support staff workshops: A method to develop P R skills and improve productivity. NASSP Bulletin, 68, 137-139.
- Mackey, J., & Alghren, A. (1977). Dimensions of adolescent alienation. Applied Psychological Measurement, 1, 219-232.
- MacNeil, R. D. (1980). The relationship of cognitive style and instructional style to the learning performance of undergraduate students. <u>Journal of Educational Research</u>, 73, 354-359.
- Maeroff, G. (1983). School and college: Partnerships in education. Princeton, NJ: Carnegie Foundation for the Advancement of Teaching.
- Maier, H. W. (1969). <u>Three theories of child development</u> (2nd ed.). New York: Harper and Row.
- Manning, M. L. (1988). Erikson's psychosocial theories help explain early adolescence. NASSP Bulletin, 72, 95-100.
- McDaniel, T. R. (1982). What's your P.Q. (Principalship Quotient)? A quiz on improving instruction. Phi Delta Kappan, 63, 464-468.
- Monk, D. H., & Ibrahim, M. A. (1984). Patterns of absence and pupil achievement. <u>American Educational Research</u>
  <u>Journal</u>, 21, 295-310.
- Morgan, G. (1986). <u>Images of organization</u>. Beverly Hills, CA: Sage.
- Nielsen, L. (1984). Teaching adolescents the skill of self management. Educational Digest, 50, 32-35.

- Nisbet, J., & Entwistle, N. (1969). <u>Transition to secondary education</u>. Edinburgh: University of London Press.
- Oakes, J. (1985). <u>Keeping track: How schools structure inequality</u>. New Haven, CT: Yale University Press.
- Ogle, D. M. (1988/1989). Implementing strategic teaching. Educational Leadership, 46, 47-60.
- Olexa, C., & Schafer, W. (1971). <u>Tracking and opportunity</u>. Scronton, PA: Chandler Publishing Company.
- Ornstein, A. C., & Hunkins, F. P. (1988). Implementing curriculum changes: Guidelines for principals. NASSP Bulletin, 72, 67-72.
- Otterbourg, S. D. (1986). School partnerships handbook:
  How to set up and administer programs with business.
  government. and your community. Englewood Cliffs, NJ:
  Prentice-Hall.
- Palincsar, A. S., Ransom, K., & Derber, S. (1989).
  Collaborative research and development of reciprocal teaching. Educational Leadership, 46, 37-40.
- Papalewis, R. (1988). A case study in organizational culture: Administrator's shared values, perceptions, and beliefs. Planning and Changing, 19, 158-165.
- Parish, R., Eubanks, E., Aquila, F. D., & Walker, S. (1989). Knock at any school. Phi Delta Kappan, 70, 386-394.
- Pfeffer, J. (1981). <u>Power in organizations</u>. Marshfield, MA: Pitman.
- Pizzo, J. (1982). Small group techniques big gains in reading. Early Years, 12, 72-74.
- Purkey, S. C., & Smith, M. S. (1983). Effective schools: A review. The Elementary School Journal, 83, 427-452.
- Reiff, J. (1985). Investigating reading preferences, reading attitudes, and teacher observations of third graders. Reading Improvement, 22, 29-33.
- Reiff, J. (1987). Student modality preference, reading achievement and reading attitude. Reading Improvement, 24, 107-111.

- Rutter, M., Maughan, B., Mortimore, P., Ouston, J., & Smith, A. (1979). Fifteen thousand hours: Secondary schools and their effects on children. Cambridge, MA: Harvard University Press.
- Safer, D. J. (1986). The stress of secondary school for vulnerable students. <u>Journal of Youth and Adolescence</u>, <u>15</u>, 405-407.
- Sarason, S. B. (1982). The culture of the school and the problem of change (2nd ed.). Boston: Allyn and Bacon.
- Schmeck, R. R., Ribich, F., & Ramanaiah, N. (1977).

  Development of a self-respect inventory for assessing individual differences in learning processes. Applied Psychological Measurement, 1, 413-431.
- Sergiovanni, T. J. (1984). Leadership and excellence in schooling. Educational Leadership, 2, 4-13.
- Shades, B. J. (1982). Afro-American cognitive style: A variable in school success? Review of Educational Research. 52, 219-244.
- Slaughter, C. H. (1969). Cognitive style: Some implications for curriculum and instructional practices among Negro children. <u>Journal of Negre Education</u>, 38, 105-111.
- Slavin, R. E. (1977). Classroom reward structure: An analytic and practical review. Review of Educational Research, 47, 633-650.
- Slavin, R. E. (1981). Synthesis of research on cooperative learning. <u>Educational Leadership</u>, 38, 655-660.
- Slavin, R. E. (1984). Component building: A Strategy for research-based instructional improvement. <u>Elementary</u> <u>School Journal</u>, <u>84</u>, 255-269.
- Slavin, R. E. (1987). Cooperative learning and the cooperative school. <u>Educational Leadership</u>, 45, 7-13.
- Slavin, R. E., & Karweit, N. L. (1984). Mastery learning and student teams: A factorial experiment in urban general mathematics classes. American Educational Research Journal, 21, 725-736.

- Slavin, R. E., & Oickle, E. (1980). Effects of learning teams on student achievement and race relations in a desegregated middle school. Paper presented at the annual convention of the American Educational Research Association, Boston.
- Smith, F. (1975). <u>Comprehension and learning: A conceptual</u> <u>framework for teachers</u>. New York: Holt, Rhinehart and Winston.
- Smith, S. C. (1987). The collaborative school takes shape. Educational Leadership, 45, 4-6.
- Squires, D. (1980). Characteristics of effective schools:
  The importance of school processes. Philadelphia:
  Research for Better Schools, Inc. (ERIC Document No. ED 197 486).
- Squires, D. A., Huitt, W. G., & Segars, J. K. (1981). Improving classrooms and schools: What's important. Educational Leadership, 39, 174-179.
- Squires, D. A., Huitt, W. G., & Segars, J. K. (1983).

  Effective schools and classrooms: A research based perspective. Alexandria, VA: Association for Supervision and Curriculum Development.
- Stefanich, G. (1983). The relationship of effective school research to school evaluation. North Central Association Quarterly, 58, 343-349.
- Trickett, E., & Moos, R. (1973). The social environment of junior high and high school classrooms. <u>Journal of Educational Psychology</u>, 65, 93-102.
- U.S. Department of Commerce. Bureau of the Census.

  <u>Twentieth Census of the United States</u>, 1980: Population and Housing, Code no. 280208.
- Webb, N. M. (1977). <u>Learning in individual and small group</u> <u>settings</u> (Technical Report #7), Stanford, CA: Aptitude Research Project, School of Education, Stanford University, 1977.
- Wedlund, G. (Ed.). (1987). <u>Directory of learning/teaching styles practitioners</u>. Alexandria, VA: Association for Supervision and Curriculum Development.
- Wynne, E. A. (1980). Looking at schools: Good, bad and Indifferent. Lexington, MA: Lexington Books.

